**TESIS DE LA UNIVERSIDAD** 

## DE ZARAGOZA

# David Díaz-Guerra Aparicio

2023 1 1

# A Geometric Deep Learning Approach to Sound Source Localization and Tracking

Director/es Beltrán Blázquez, José Ramón Miguel Artiaga, Antonio



Prensas de la Universidad Universidad Zaragoza

ISSN 2254-7606

© Universidad de Zaragoza Servicio de Publicaciones

ISSN 2254-7606



**Tesis Doctoral** 

# A GEOMETRIC DEEP LEARNING APPROACH TO SOUND SOURCE LOCALIZATION AND TRACKING

Autor

David Díaz-Guerra Aparicio

Director/es

Beltrán Blázquez, José Ramón Miguel Artiaga, Antonio

#### UNIVERSIDAD DE ZARAGOZA Escuela de Doctorado

Programa de Doctorado en Ingeniería Electrónica



# A Geometric Deep Learning Approach to Sound Source Localization and Tracking

Author: David Díaz-Guerra Aparicio

Thesis supervisors: Dr. José Ramón Beltrán Blázquez Dr. Antonio Miguel Artiaga

Department of Electronic Engineering and Communications

University of Zaragoza

A thesis submitted for the degree of Doctor of Philosophy (PhD)

January 2023

#### Acknowledgements

First and foremost, I cannot begin to express my thanks to my thesis supervisors, Dr. José Ramón Beltrán and Dr. Antonio Miguel Artiaga, for always letting me conduct my research through the paths I chose but, at the same time, always being there when I needed any kind support or advice. This endeavor would not have been possible without them. I would also like to recognize all the assistance received over these years from the staff of the Department of Electronic Engineering and Communications.

I am also really grateful to all the people from the Audio Research Group at Tampere University, who made me feel at home during my stay there, and especially to Dr. Tuomas Virtanen and Dr. Archontis Politis, who supervised my work during it.

I should express my gratitude to the Regional Government of Aragon who has funded this work with a grant for postgraduate research contracts, but not to the government of the same institution that did not open a call for this kind of grant in 2015 originating a delay that caused that the funding of this work started one year later than it should have done. I should also express my gratitude to my family for funding the first year of this thesis. Finally, I want to thank all the people who have fought, fight, and will fight for quality public education and for the labor rights and dignity of those who work in the research sector, especially to my mates from *Estudiantes en Movimiento* (previously known as *CEPA*) and *PIF Unizar*.

#### Abstract

The localization and tracking of sound sources using microphone arrays is a problem that, even if it has attracted attention from the signal processing research community for decades, remains open. In recent years, deep learning models have surpassed the state-of-the-art that had been established by classic signal processing techniques, but these models still struggle with handling rooms with strong reverberations or tracking multiple sources that dynamically appear and disappear, especially when we cannot apply any criteria to classify or order them. In this thesis, we follow the ideas of the Geometric Deep Learning framework to propose new models and techniques that mean an advance of the state-of-the-art in the aforementioned scenarios.

As the input of our models, we use acoustic power maps computed using the SRP-PHAT algorithm, a classic signal processing technique that allows us to estimate the acoustic energy received from any direction of the space and, therefore, compute arbitrary-shaped power maps. In addition, we also propose a new technique to analytically cancel a source from the generalized cross-correlations used to compute the SRP-PHAT maps. Based on previous narrowband cancellation techniques, we prove that we can project the cross-correlation functions of the signals captured by a microphone array into a space orthogonal to a given direction by just computing a linear combination of time-shifted versions of the original cross-correlations. The proposed cancellation technique can be used to design iterative multi-source localization systems where, after having found the strongest source in the generalized cross-correlation functions or in the SRP-PHAT maps, we can cancel it and find new sources that were previously masked by the first source.

Before being able to train deep learning models we need data, which, in the case of following a supervised learning approach, means a dataset of multichannel recordings with the position of the sources accurately labeled. Although there exist some datasets like this, they are not large enough to train a neural network and the acoustic environments they include are not diverse enough. To overcome this lack of real data, we present a technique to simulate acoustic scenes with one or several moving sound sources and, to be able to perform these simulations as they are needed during the training, we present what is, to the best of our knowledge, the first free and open source room acoustics simulation library with GPU acceleration. As we prove in this thesis, the presented library is more than two orders of magnitude faster than other state-of-the-art CPU libraries.

The main idea of the Geometric Deep Learning philosophy is that the models should fit the symmetries (i.e. the invariances and equivariances) of the data and the problem we want to solve. For single-source direction of arrival estimation, the use of SRP-PHAT maps as inputs of our models makes the rotational equivariance of the problem undeniably clear and, after a first approach using 3D convolutional neural networks, we present a model using icosahedral convolutions that approximate the equivariance to the continuous group of spherical rotations by the discrete group of the 60 icosahedral symmetries. We prove that the SRP-PHAT maps are a much more robust input feature than the spectrograms typically used in many state-of-the-art models and that the use of the icosahedral convolutions. combined with a new soft-argmax function that obtains a regression output from the output of the convolutional neural network by interpreting it as a probability distribution and computing its expected value, allows us to dramatically reduce the number of trainable parameters of the models without losing accuracy in their estimations.

When we want to track multiple moving sources and we cannot use any criteria to order or classify them, the problem becomes invariant to the permutations of the estimates, so we cannot directly compare them with the ground truth labels since we cannot expect them to be in the same order. This kind of models has typically been trained using permutation invariant training strategies, but these strategies usually do not penalize the identity switches and the models trained with them do not keep the identity of every source consistent during the tracking. To solve this issue, we propose a new training strategy, which we call sliding permutation invariant training, that is able to optimize all the features that we could expect from a multi-source tracking system: the precision of the direction of arrival estimates, the accuracy of the source detections, and the consistency of the assigned identities.

Finally, we propose a new kind of recursive neural network that, instead of using vectors as their input and their state, uses sets of vectors and is invariant to the permutation of the elements of the input set and equivariant to the permutations of the elements of the state set. We show how this is the behavior that we should expect from a tracking model which takes as inputs the estimates of a multi-source localization model and compare these permutation-invariant recursive neural networks with the conventional gated recurrent units for sound source tracking applications.

#### Resumen

La localización y el *tracking* de fuentes sonoras mediante agrupaciones de micrófonos es un problema que, pese a llevar décadas siendo estudiado, permanece abierto. En los últimos años, modelos basados en *deep learning* han superado el estado del arte que había sido establecido por las técnicas clásicas de procesado de señal, pero estos modelos todavía presentan problemas para trabajar en espacios con alta reverberación o para realizar el *tracking* de varias fuentes sonoras, especialmente cuando no es posible aplicar ningún criterio para clasificarlas u ordenarlas. En esta tesis, se proponen nuevos modelos que, basados en las ideas del *Geometric Deep Learning*, suponen un avance en el estado del arte para las situaciones mencionadas previamente.

Los modelos propuestos utilizan como entrada mapas de potencia acústica calculados con el algoritmo SRP-PHAT, una técnica clásica de procesado de señal que permite estimar la energía acústica recibida desde cualquier dirección del espacio. Además, también proponemos una nueva técnica para suprimir analíticamente el efecto de una fuente en las funciones de correlación cruzada usadas para calcular los mapas SRP-PHAT. Basándonos en técnicas de banda estrecha, se demuestra que es posible proyectar las funciones de correlación cruzada de las señales capturadas por una agrupación de micrófonos a un espacio ortogonal a una dirección dada simplemente usando una combinación lineal de las funciones originales con retardos temporales. La técnica propuesta puede usarse para diseñar sistemas iterativos de localización de múltiples fuentes que, tras localizar la fuente con mayor energía en las funciones de correlación cruzada o en los mapas SRP-PHAT, la cancelen para poder encontrar otras fuentes que estuvieran enmascaradas por ella.

Antes de poder entrenar modelos de *deep learning* necesitamos datos. Esto, en el caso de seguir un esquema de aprendizaje supervisado, supone un *dataset* de grabaciones de audio multicanal con la posición de las fuentes etiquetada con precisión. Pese a que existen algunos *datasets* con estas características, estos no son lo suficientemente extensos para entrenar una red neuronal y los entornos acústicos que incluyen no son suficientemente variados. Para solventar el problema de la falta de datos, presentamos una técnica para simular escenas acústicas con una o varias fuentes en movimiento y, para realizar estas simulaciones conforme son necesarias durante el entrenamiento de la red, presentamos la que es, que sepamos, la primera librería de *software* libre para la simulación de acústica de salas con aceleración por GPU. Tal y como queda demostrado en esta tesis, esta librería es más de dos ordenes de magnitud más rápida que otras librerías del estado del arte.

La idea principal del *Geometric Deep Learning* es que los modelos deberían compartir las simetrías (i.e. las invarianzas y equivarianzas) de los datos y el problema que se quiere resolver. Para la estimación de la dirección de llegada de una única fuente, el uso de mapas SRP-PHAT como entrada de nuestros modelos hace que la equivarianza a las rotaciones sea obvia y, tras presentar una primera aproximación usando redes convolucionales tridimensionales, presentamos un modelo basado en convoluciones icosaédricas que son capaces de aproximar la equivarianza al grupo continuo de rotaciones esféricas por la equivarianza al grupo discreto de las 60 simetrías del icosaedro. En la tesis se demuestra que los mapas SRP-PHAT son una característica de entrada mucho más robusta que los espectrogramas que se usan típicamente en muchos modelos del estado del arte y que el uso de las convoluciones icosaedricas, combinado con una nueva función *softargmax* que obtiene una salida de regresión a partir del resultado de una red convolucional interpretándolo como una distribución de probabilidad y calculando su valor esperado, permite reducir enormemente el número de parámetros entrenables de los modelos sin reducir la precisión de sus estimaciones.

Cuando queremos realizar el tracking de varias fuentes en movimiento y no podemos aplicar ningún criterio para ordenarlas o clasificarlas, el problema se vuelve invariante a las permutaciones de las estimaciones, por lo que no podemos compararlas directamente con las etiquetas de referencia dado que no podemos esperar que sigan el mismo orden. Este tipo de modelos se han entrenado típicamente usando estrategias de entrenamiento invariantes a las permutaciones, pero estas normalmente no penalizan los cambios de identidad por lo que los modelos entrenados con ellas no mantienen la identidad de cada fuente de forma consistente. Para resolver este problema, en esta tesis proponemos una nueva estrategia de entrenamiento, a la que llamamos *sliding permutation invariant training* (sPIT), que es capaz de optimizar todas las características que podemos esperar de un sistema de *tracking* de múltiples fuentes: la precisión de sus estimaciones de dirección de llegada, la exactitud de sus detecciones y la consistencia de las identidades asignadas a cada fuente.

Finalmente, proponemos un nuevo tipo de red recursiva que usa conjuntos de vectores en lugar de vectores para representar su entrada y su estado y que es invariante a las permutaciones de los elementos del conjunto de entrada y equivariante a las del conjunto de estado. En esta tesis se muestra como este es el comportamiento que deberíamos esperar de un sistema de *tracking* que toma como entradas las estimaciones de un modelo de localización multifuente y se compara el rendimiento de estas redes recursivas invariantes a las permutaciones con redes recursivas GRU convencionales para aplicaciones de *tracking* de fuentes sonoras.

## Contents

Li	st of	Publications xv	/ii	
Li	st of	Figures x	ix	
Li	st of	Tables xx	٢v	
1	Intr	troduction		
	1.1	Context and motivation	1	
	1.2	Sound source localization and tracking using neural networks $\ . \ . \ .$	3	
		1.2.1 Input representation	3	
		1.2.2 Network architecture	4	
		1.2.3 Output representation	5	
		1.2.4 Training and datasets	6	
	1.3	Geometric deep learning	7	
		1.3.1 General concepts	7	
		1.3.2 Geometric deep learning for sound source localization and tracking	8	
		1.3.2.1 Rotational symmetry	10	
		1.3.2.2 Source permutation symmetry	10	
	1.4	Main contributions and results of the thesis	11	
	1.5	Outline and organization of the thesis	13	
<b>2</b>	The	SRP-PHAT power maps	15	
	2.1	Classic signal processing techniques for broadband DOA estimation $\  \   $	15	
	2.2	The SRP-PHAT algorithm	18	
	2.3	Source cancellation in Cross-Correlation functions	21	

		2.3.1	Narrowh	band source elimination	22
		2.3.2	Broadba	and source elimination	23
		2.3.3	Coarse a	$\alpha$ pproximation $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	25
		2.3.4	Phase T	ransform (PHAT)	26
		2.3.5	Example	es	27
			2.3.5.1	Two uncorrelated Gaussian sources	27
			2.3.5.2	Two impulsive sources	27
			2.3.5.3	Two speech sources with reverberation $\ldots \ldots \ldots$	29
	2.4	An ite	rative mu	lti-source localization algorithm based on source cancel-	
		lation	and SRP	-PHAT	31
		2.4.1	Algorith	m	31
		2.4.2	Simulati	ons	33
			2.4.2.1	Uncorrelated white sources	33
			2.4.2.2	Uncorrelated speech sources	35
			2.4.2.3	Correlated speech sources	37
		2.4.3	Recordin	ngs	40
			2.4.3.1	White sources	41
			2.4.3.2	Speech sources	43
	2.5	Conclu	usions		45
3	An	infinit	e-size sy	nthetic dataset for sound source localization and	1
	trac	king			49
	3.1	Movin	g sources	simulation	50
		3.1.1	Trajecto	ory generation	50
		3.1.2	Trajecto	ory simulation	51
		3.1.3	Multi-sc	vurce scenes	53
	3.2	Room	impulse i	response simulation with GPU acceleration $\ldots$	55
		3.2.1	Introduc	ction	55
		3.2.2	The Ima	age Source Method (ISM)	56
			3.2.2.1	Original Allen and Berkley algorithm	57
			3.2.2.2	Improvements to the original algorithm $\ldots \ldots \ldots$	59
		3.2.3	Parallel	implementation	61

			3.2.3.1	Amplitudes and delays computation	63
			3.2.3.2	Computation and sum of the contribution of each im-	
				age source	63
			3.2.3.3	Diffuse reverberation computation $\ldots \ldots \ldots$	65
			3.2.3.4	Simulating moving sources	65
			3.2.3.5	Lookup tables (LUTs)	65
			3.2.3.6	Mixed precision	66
		3.2.4	Python	library	68
		3.2.5	Results		69
			3.2.5.1	Base implementation	69
			3.2.5.2	Lookup tables	73
			3.2.5.3	Mixed precision	74
	3.3	Conclu	usions		76
4	Rob	oust sii	ngle sou	ce localization	79
	4.1	Robus	t sound s	source localization using SRP-PHAT and 3D convolu-	
		tional	neural ne	tworks	80
		4.1.1	Preproce	essing	81
		4.1.2	Model a	rchitecture	82
		4.1.3	Training		84
		4.1.4	Evaluati	on	85
			4.1.4.1	Baseline methods	85
			4.1.4.2	Simulated dataset	87
			4.1.4.3	LOCATA dataset	90
	4.2	A sour	nd source	localization model equivariant to the rotations of the	
		source	and the	array	93
		4.2.1	Icosahed	ral CNNs	94
		4.2.2	Soft-arg	max regression	97
		4.2.3	Propose	d technique	98
			4.2.3.1	Model architecture	98
			4.2.3.2	Training	101

		4.2.4	Evaluati	on	102
			4.2.4.1	Baseline techniques	102
			4.2.4.2	Simulated dataset	102
			4.2.4.3	LOCATA dataset	106
	4.3	Conclu	usions		109
5	Per	mutati	on invar	iant multi-source tracking	111
	5.1	Sound	source tr	acking (SST) and the permutation invariance $\ldots$ .	111
	5.2	Iterati	ve multi-	source localization through source cancellation using	
		deep le	earning .		113
	5.3	Permu	tation inv	variant gated recurrent units (PI-GRUs)	117
	5.4	Permu	tation inv	variant training (PIT)	121
		5.4.1	Frame-le	evel Permutation Invariant Training (fPIT)	121
		5.4.2	Utteranc	ce-level Permutation Invariant Training (uPIT) $\ldots$	123
		5.4.3	Sliding F	Permutation Invariant Training (sPIT)	123
	5.5	Evalua	$tion \ldots$		124
		5.5.1	Experim	ent design	124
		5.5.2	Results		126
	5.6	Conclu	usions		129
6	Con	clusio	ns and fu	uture work	135
	6.1	Conclu	usions		135
	6.2	Future	work		137
		6.2.1	Training	data	137
		6.2.2	Rotation	n-equivariant models for sound source localization (SSL)	138
		6.2.3	Permuta	tion invariant recurrent networks	139
		6.2.4	Eliminat	ing the information bottlenecks of the proposed model	141

$\mathbf{A}$	Network architecture of the proposed and baseline models					
	A.1 Cross3D models	162				
	A.2 icoCNN models	167				
	A.3 Baseline models	170				
в	Results in every recording of the LOCATA dataset	175				
$\mathbf{C}$	Conclusiones	179				

### List of Publications

The following publications have been produced as result of the works in this thesis:

- [I] D. Diaz-Guerra and J. R. Beltran, "Direction of Arrival Estimation with Microphone Arrays Using SRP-PHAT and Neural Networks," in 2018 IEEE 10th Sensor Array and Multichannel Signal Processing Workshop (SAM), Jul. 2018, pp. 617–621. doi: 10.1109/SAM.2018.8448492.
- [II] D. Diaz-Guerra and J. R. Beltran, "Source cancellation in cross-correlation functions for broadband multisource DOA estimation," in Signal Processing, vol. 170, p. 107442, May 2020, doi: 10.1016/j.sigpro.2019.107442.
  IF (JCR 2020): 4.662. Engineering, Electrical & Electronic rank: 44/273 (Q1)
- [III] D. Diaz-Guerra, A. Miguel, and B. Jose R., "Acoustic Room Impulse Response Simulation with GPUs," presented at the Second International Conference on Advances in Signal Processing and Artificial Intelligence (ASPAI' 2020), Berlin, Germany, Jul. 2020.
- [IV] D. Diaz-Guerra, A. Miguel, and J. R. Beltran, "gpuRIR: A python library for room impulse response simulation with GPU acceleration," in Multimedia Tools and Applications, Oct. 2020, doi: 10.1007/s11042-020-09905-3.
   IF (JCR 2020): 2.757. Engineering, Electrical & Electronic rank: 120/273 (Q2)
- [V] D. Diaz-Guerra, A. Miguel, and J. R. Beltran, "Robust Sound Source Tracking Using SRP-PHAT and 3D Convolutional Neural Networks," in IEEE/ACM Transactions on Audio, Speech, and Language Processing, vol. 29, pp. 300–311,

2021, doi: 10.1109/TASLP.2020.3040031.

IF (JCR 2021): 4.364. Acoustics rank: 5/32 (Q1)

- [VI] D. Diaz-Guerra, A. Miguel and J. R. Beltran, "Direction of Arrival Estimation of Sound Sources Using Icosahedral CNNs," in IEEE/ACM Transactions on Audio, Speech, and Language Processing, vol 31, pp. 313-321, 2023, doi: 10.1109/TASLP.2022.3224282.
  IF (JCR 2021): 4.364. Acoustics rank: 5/32 (Q1)
- [VII] D. Diaz-Guerra, A. Politis, T. Virtanen, "Position tracking of a varying number of sound sources with sliding permutation invariant training", arXiv preprint arXiv:2210.14536.
   Submitted to the 2023 IEEE International Conference on Acoustics, Speech, and Signal Processing (ICASSP 2023)

David Diaz-Guerra is the first author of all the publications in this thesis and he proposed the initial ideas for them and conducted the experimental works. Jose Ramon Beltran and Antonio Miguel supervised the works, discussed the ideas, and provided constant feedback during the experimentation and the writing of the manuscripts. The work presented in [VII] was conducted during a stay at the Audio Research Group at the Tampere University (Finland) under the supervision of Archontis Politis and Tuomas Virtanen.

The manuscript [VII] has been submitted to the 2023 IEEE International Conference on Acoustics, Speech, and Signal Processing (ICASSP 2023) and is currently under review.

This thesis includes the reproduction of some figures and text fragments from these papers with permission from the respective copyright holders.

# List of Figures

2.1	Example of SRP-PHAT power maps with different resolutions in a fa-	
	vorable scenario: SNR=30 dB and $T_{60}=0.3$ s	20
2.2	Example of SRP-PHAT power maps with different resolutions in an	
	adverse scenario: SNR=5 dB and $T_{60}$ =0.9 s	21
2.3	A representation of the auto-correlation functions of 2 sources and 3 $$	
	sensors	25
2.4	A representation of the terms $P_{12}(t)$ and $Q_{12}(t)$ in equations (2.17),	
	(2.18) and $(2.19)$	26
2.5	Original CCs between sensors in the scenario described in the exam-	
	ple 2.3.5.1, after applying the coarse approximation to cancel the first	
	source, and after applying the complete equation of the proposed can-	
	cellation technique	28
2.6	Original CCs between sensors in the scenario described in the exam-	
	ple 2.3.5.2, after applying the coarse approximation to cancel the first	
	source, and after applying the complete equation of the proposed can-	
	cellation technique	30
2.7	Original CC between sensors 1 and 5 in the scenario described in the	
	example 2.3.5.3 and the result of applying the coarse approximation	
	and the complete equation over it. $\ldots$ . $\ldots$ . $\ldots$ . $\ldots$ .	31
2.8	Iterative algorithm to find several sources using source cancellation and	
	SRP techniques.	32

#### LIST OF FIGURES

2.9	Probability of finding the second source in a 2 random sources scenario	
	with $P_{s1} = 0 \mathrm{dB}$ , $\mathrm{SNR}_{s1} = -10 \mathrm{dB}$ and different values of $P_{s2}$	34
2.10	Probability of finding the second source in a 2 speech sources scenario	
	with $P_{s1} = 0 \mathrm{dB}$ , $\mathrm{SNR}_{s1} = -10 \mathrm{dB}$ and different values of $P_{s2}$	36
2.11	SRP power map resulting from the simulation of two speech sources	
	with equal power and the SRP power map after applying our source	
	cancellation technique at its maximum.	37
2.12	SRP and SRP-PHAT power maps resulting from the simulation of two	
	speech sources with $P_{s2} = P_{s1} - 13 \mathrm{dB}$ and after applying our source	
	cancellation technique	38
2.13	Probability of finding the second source in a 2 correlated speech sources	
	scenario with $P_{s1} = 0 \mathrm{dB}$ , $\mathrm{SNR}_{s1} = -10 \mathrm{dB}$ and different values of $P_{s2}$ .	39
2.14	Probability of finding the second source in a 2 correlated speech sources	
	scenario with $P_{s1} = 0 \mathrm{dB}$ , $\mathrm{SNR}_{s1} = -10 \mathrm{dB}$ and different values of $P_{s2}$	
	using a longer window size $(K = 4096)$	40
2.15	Set-up of the recordings	41
2.16	Power maps obtained with SRP-PHAT and applying our algorithm at	
	the position of the maximum of the previous map for 2 loudspeakers	
	with $70\mathrm{dB}_{\mathrm{SPL}}$ and $65\mathrm{dB}_{\mathrm{SPL}}$ of white noise at the array position	42
2.17	Power maps obtained with SRP-PHAT and applying our algorithm at	
	the position of the maximum of the previous map for 3 loudspeakers	
	with $75dB_{\rm SPL},70dB_{\rm SPL}$ and $65dB_{\rm SPL}$ of white noise at the array position.	42
2.18	Power maps obtained with SRP-PHAT and applying our algorithm at	
	the position of the maximum of the previous map for 4 loudspeakers	
	with $65  \mathrm{dB}_{\mathrm{SPL}}$ of white noise at the array position	43
2.19	Power maps obtained with SRP-PHAT and applying our algorithm at	
	the position of the maximum of the previous map for 2 loudspeakers	
	playing speech signals	44
2.20	Power maps obtained with SRP-PHAT and applying our algorithm at	
	the position of the maximum of the previous map for 3 loudspeakers	
	playing speech signals	44

3.1	Examples of source trajectories used to train the models	51
3.2	Distribution of the source speed in the synthetic dataset	52
3.3	Dataset generation process.	52
3.4	Distribution of the number of sources in every time frame	54
3.5	Image sources for a two-dimensional room	57
3.6	image source method parallel implementation	62
3.7	Parallel reduction sum of the sincs	64
3.8	Runtime of each library for computing different numbers of RIRs ( $M_{\rm src}$ )	
	in a room with size $3 \mathrm{m} \times 4 \mathrm{m} \times 2.5 \mathrm{m}$ and $T_{60} = 0.7 \mathrm{s} \ldots \ldots \ldots \ldots$	70
3.9	Runtime of each library for computing 128 RIRs in a room with size	
	$3\mathrm{m}\times4\mathrm{m}\times2.5\mathrm{m}$ and different reverberation times. 	72
3.10	RIR computed with single precision trigonometric functions and the	
	errors introduced using a LUT and half-precision functions	75
11	Cross <sup>2</sup> D model prohitesture	82
4.1	Localization next mean ground angular error (DMSAE) for ground	00
4.2	power map resolutions. SNP and reverberation times	00
13	Examples of the DOA estimated in a favorable scenario using high	00
4.0	resolution maps and in an adverse scenario using low resolution maps	80
4.4	Tracking root mean general angular owner (PMSAE) of Cross2D with	89
4.4	reacting root mean squared angular error (KMSAE) or Crossad with	
	verberation times and SNP ratios	00
45	POA estimated for the second recording of the second task of the IO	90
4.0	DOA estimated for the second recording of the second task of the EO- CATA shallongs using maps with $Cross^{2}D$ over $22 \times 64$ maps and the	
	CATA channenge using maps with Crossab over $32 \times 04$ maps and the	02
16	Sampling points of a face of the isosphedron for different resolutions	92
4.0	Sampling points of a face of the icosahedrol SDD DLAT norman man with	90
4.1	several representations of an icosanedral SKF-FHAT power map with	06
4.0	resolution $r = 3$ in a high reverberation low noise scenario	90
4.8	Output of the last convolutional layer after passing through the soft-	
	max function and the result of the soft-argmax function converted into	00
	spnerical coordinates.	-99

#### LIST OF FIGURES

4.9	Architecture of the proposed model. $\ldots$	100
4.10	Localization root mean squared angular error (RMSAE) under several	
	simulated conditions for the proposed and the baseline techniques. $\ .$ .	103
4.11	Localization root mean squared angular error (RMSAE) vs the number	
	of computations of equation (2.3) needed to compute each input map.	104
4.12	DOA estimation for a random trajectory simulated with $T_{60}{=}1.0\mathrm{s}$ and	
	SNR=30 dB	105
4.13	Average localization root mean squared angular error (RMSAE) in the	
	LOCATA dataset vs the number of computations of equation $(2.3)$	
	needed to compute each input map. $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	107
4.14	DOA estimation for the third recording of task 5 of the LOCATA dataset	t.108
4.15	Example of an input map from the 1st recording of the 5th task of the	
	LOCATA evaluation dataset and the output of the last convolutional	
	layer	109
<b>F</b> 1		110
5.1	Diagram of a tracking-by-detection system seen as a recurrent process.	113
5.2	Structure of iterative multi-source localization systems using source can-	
	cellation over the GCC and over the SRP-PHAT maps	114
5.3	Experiments training an MLP to replicate the analytical coefficients and	
	time shifts of the source cancellation technique presented in chapter 2.	116
5.4	Architecture of the proposed permutation invariant gated recurrent unit	
	(PI-GRU)	118
5.5	Examples of 1D trajectories and the result of applying the different PIT	
	strategies.	122
5.6	Architecture of the permutation invariant RNN used to evaluate the	
	tracking capabilities of the PI-GRUs	125
5.7	Architecture of the RNN used after the icoCNN as baseline in the eval-	
	uation	126
5.8	Evaluation metrics obtained using sPIT and fPIT to train the three	
	analyzed models.	127

5.9	Example of the trajectories estimated by the evaluated models in an	
	evaluation acoustic with up to 2 concurrent sound sources	131
5.10	Example of the trajectories estimated by the evaluated models in an	
	evaluation acoustic with up to 3 concurrent sound sources	132
5.11	Another example of the trajectories estimated by the evaluated models	
	in an evaluation acoustic with up to 3 concurrent sound sources. $\ . \ .$	133
6.1	Different permutation invariant recurrent architectures	140

# List of Tables

3.1	Comparison of some state-of-the-art ISM implementation	56
3.2	Kernels and functions of the CUDA implementation	61
3.3	GPUs employed for the performance analysis	70
3.4	Lookup table (LUT) and mixed precision (MP) simulation times and	
	speedups for computing different numbers of room impulse response	
	(RIR) with $T_{60} = 0.7 \text{s.}$	73
3.5	Lookup table (LUT) and mixed precision (MP) simulation times and	
	speedups for computing 128 RIRs with different reverberation times	74
4.1	Models employed for the evaluation of Cross3D	86
4.2	RMSAE of the DOA estimated for the LOCATA dataset with Cross3D	
	using several map resolutions and the baseline tracking methods	91
4.3	Models employed for the evaluation	102
4.4	Mean RMSE in the simulated dataset with the different strategies to	
	handle the vertices in the icosahedral convolutions	105
4.5	Mean RMSAE in the simulated dataset with different numbers of con-	
	volutional kernels.	106
4.6	Mean RMSAE of the DOA estimated for every task of the evaluation	
	partition of the LOCATA dataset with the icosahedral CNNs using	
	several map resolutions and the baseline tracking methods. $\ldots$ .	107
5.1	Mean RMSAE simulating just one source and simulating two sources	
	and analytically canceling one of them	113

A.1	Architecture of Cross3D for 4x8 power maps	162
A.2	Architecture of Cross3D for 8x16 power maps	163
A.3	Architecture of Cross3D for 16x32 power maps	164
A.4	Architecture of Cross3D for 32x64 power maps	165
A.5	Architecture of Cross3D for 64x128 power maps	166
A.6	Architecture of the icoCNN model for r=1 power maps	167
A.7	Architecture of the icoCNN model for r=2 power maps. $\ldots$ .	168
A.8	Architecture of the icoCNN model for r=3 power maps	169
A.9	Architecture of the 1D CNN for DOA estimation from 858 sequences of	
	GCCs	170
A.10	Architecture of the 1D CNN for DOA estimation from the coordinates	
	of the maximums of the SRP-PHAT maps	171
A.11	Architecture of the 2D CNN for DOA estimation from the spectrograms	
	of the 12 microphone signals	172
A.12	Architecture of SELDnet for DOA estimation from spectrograms of the	
	12 microphone signals $\ldots \ldots \ldots$	173
D 1	DMCAE of the DOA estimated for the development partition of the	
В.1	RMSAE of the DOA estimated for the development partition of the	170
<b>.</b>	LOCATA dataset	176
B.2	RMSAE of the DOA estimated for the evaluation partition of the LO-	
	CATA dataset	177

#### 1

### Introduction

In this introduction chapter, we first make a general description of the context of the research developed along this thesis and the main ideas that guided it (section 1.1). Then, we make a brief review of the start-of-the-art of sound source localization and tracking with neural networks (section 1.2) and introduce the concept of Geometric Deep Learning (section 1.3). Finally, we summarize the main contributions and results of this thesis (section 1.4) and outline the structure of the following chapters of the thesis (section 1.5).

#### **1.1** Context and motivation

Microphone array signal processing is a research field with a long trajectory within the signal processing community. Even if we can find earlier references ([1, 2] for example), it is typically considered that their study and implementation started in the late 70s and early 80s [3] and, since then, there has been a constant stream of both research studies and applications. However, in the last few years, after the revolution originated by the rebirth of neural networks in computer vision applications, there has been an explosion of new research studies that, using deep learning techniques, have clearly overcome the classic techniques that had constituted the state-of-the-art for years. This improvement in the capabilities of, for example, speech recognition systems, has led to the emergence of new applications of microphone arrays, such as

#### 1. INTRODUCTION

*smart speakers* or teleconference devices, that have increased even more the attention to multichannel signal processing research.

Focusing on the main topic of this dissertation, sound source localization and tracking have always been active fields of research since, apart from their own applications, they are a preliminary step for many other applications such as speech enhancement using beamforming techniques. Knowing this, it is not surprising that, in recent years, many deep learning-based techniques had been proposed. However, although these new techniques improve the state-of-the-art previously established by the classical methods, we are still far from being able to perfectly estimate and track the position of sound sources in highly reverberant spaces using compact arrays.

In this context of a growing number of proposals for new sound source localization and tracking systems based on deep learning, the main ideas that guided this thesis were:

- 1. Exploiting the classic signal processing knowledge of the problem: even if the performance of the classical methods quickly decreased in presence of reverberation, they were based on interesting analytical models that allow us to extract spatial information from the audio signals recorded with microphone arrays. Even if end-to-end approaches can also offer some advantages, carefully designed preprocessing stages can provide more refined input features to the neural networks and allow them to obtain better results.
- 2. Using models that fit the nature of the problem: many of the first deep learningbased proposals for sound source localization and tracking were straightforward translations of the techniques that had obtained good results in computer vision problems using the spectrograms of the microphone array signals as input images of convolutional neural networks (CNNs). However, audio spectrograms have a very different nature than images since, while both dimensions of an image represent the same magnitude (space), the two dimensions of an audio spectrogram represent two different magnitudes (time and frequency) with different properties and characteristics. What started more like an intuition at the beginning of this thesis has recently been properly formalized under the name of Geometric

Deep Learning (GDL), an approach with growing popularity in many fields and applications of deep learning.

3. Avoiding the use of non-causal elements: while most of the classic methods were strictly causal, and therefore they were feasible for real-time applications (although the computational complexity of some of them made impossible their real-time implementation at that time), many of the techniques based on deep learning that have been proposed in the recent literature include non-causal elements (such as bidirectional recurrent units) that make impossible any kind of real-time implementation independently of the available computational resources. However, although sound source localization can be performed offline in some applications (like audio forensics), most require real-time estimations.

#### 1.2 Sound source localization and tracking using neural networks

In this section, we present a brief review of the state-of-the-art of sound source localization and tracking using deep learning techniques, especially focusing on those techniques that have influenced this work. For a more in-depth and extensive literature review, we recommend the remarkable survey by Grumiaux et al. [4].

In general, we can consider that, in order to design a deep learning system, we must choose: i) with which representation of the input of our problem we will feed our neural network, ii) with which neural architecture we are going to process that input, iii) how the neural network should represent the result of our problem and iv) how we are going to train it (including with which dataset).

#### 1.2.1 Input representation

Although feeding a neural network directly with the raw audio samples obtained with a microphone array is also possible [5, 6, 7], most of the state of the state-of-the-art techniques perform some preprocessing over the audio samples to provide their neural networks with a higher level representation of the acoustic scene.

#### 1. INTRODUCTION

The simplest (and most general) input features typically used as input of neural networks for sound source localization are multichannel spectral representations. Some proposals use only magnitude spectrograms [8, 9, 10] but, since most of the localization information is usually on the inter-channel phase differences, it is more common to use phase spectrograms [11, 12, 13] or combinations of both, stacking two separate channels with the magnitude and phase spectrum [14, 15, 16, 17] or the real and imaginary part of the spectrum [18, 19, 20], or using a single channel with complex values [21].

Other proposals use higher-level representations more suited for the sound source localization task, such as generalized cross-correlation coefficients [22, 23, 24, 25] or Ambisonics intensity vectors [26, 27, 28]. We could consider as the highest level representations those based on classical signal processing techniques that allow computing spatial representation of an acoustic scene from the multichannel signals of a microphone array, which can be based on the eigendecomposition of the cross-correlation matrices of the signals, such as MUSIC [29] or ESPRIT [30], or in acoustic power maps, such as the SRP-PHAT method [31, 32]. For example, we can find the use of the eigenvectors of the cross-correlation matrices as inputs of a neural network in [33], the use of MUSIC-based spatial pseudo-spectrums in [34], or SRP-PHAT power maps in [35] and in the works presented in this thesis.

Finally, it is worth mentioning that combining several of the input features discussed above is also a popular approach [36, 37, 38]. A comparison between several of these combinations can be found in [39].

#### **1.2.2** Network architecture

Although multi-layer perceptrons were the most popular approach among the first proposals for sound source localization using neural networks [22, 23], the attention soon moved towards the architecture that was revolutionizing several computer vision problems: convolutional neural networks (CNNs) [9, 12, 37, 40]. In order to improve the tracking capabilities of CNNs, many authors propose the use of one or several recurrent layers after them, creating recurrent convolutional neural networks (RCNNs) [14, 41, 42]; however, most of these RCNNs use bidirectional recurrent layers that are
non-causal and, therefore, are not suitable for real-time applications. More recently, other architectures have also been proposed for sound source localization and tracking, like the use of attention mechanisms [43, 44, 45] or autoencoder and variational autoencoder architectures [46, 47, 48, 49, 50].

In section 1.3, we provide a further analysis of the different approaches of using CNNs for sound source localization and their advantages and disadvantages.

# 1.2.3 Output representation

Among the first proposals for sound source localization based on deep learning, the most popular approach to encode the direction of arrival (DOA) at the network output was to define a set of possible DOAs and use a network output to represent the probability of each one of them containing a sound source; i.e., they solved the localization problem as a classification problem [22, 51, 52]. In the case of single-source localization, the estimated DOA simply corresponds to the maximum of the outputs, but in the case of multiple sound source localization, we need to use some peak-picking algorithm to determine the number of sources and which maxima actually correspond to real sources.

One of the main drawbacks of this classification approach is that the maximum accuracy that the system can reach is limited by the number of outputs of the network. This problem is not too serious if we only want to estimate the azimuth component of the DOAs, but the number of outputs needed to keep the desired resolution grows quadratically if we want to estimate both the azimuth and the elevation and even cubically if we also want to estimate the distance or if we want to estimate XYZ Cartesian coordinates in the case of distributed arrays. Increasing the number of outputs of the network does not only mean increasing its size (and therefore the computational complexity of the system) but also the size of the training dataset, which should contain enough examples of sources in all the possible positions (especially when using fully-connected layers at the end of the network, which does not have any geometrical structure and whose outputs are completely independent regardless of the position they represent).

#### 1. INTRODUCTION

On the other hand, most of the recent proposals seem to prefer regression approaches, where the DOAs are directly provided by the continuous value of the outputs of the network. Since DOAs are generally expressed in terms of spherical coordinates (azimuth and elevation or polar angle) we could expect them to be the most common representation at the output of the regression networks, but it has been proved that estimating unitary 3D Cartesian vectors pointing towards the DOA provides better results [53, 54] and this is the most popular approach nowadays [14, 38, 55].

However, the regression approach makes working with several sources more complex. First, we need to set a maximum number of sources the system can work with (which will determine the number of outputs of the network) and, after that, we need to establish a strategy to determine which outputs are representing active sources at every time frame. To do this, many proposals use an additional binary-classification output for every localization output to act as activity detector (which can also be used for sound event detection when every one of these outputs represents a different event class) [14, 38, 56, 57] but in the case of using 3D Cartesian vectors to represent the DOA, we can also encode the activity estimation into the norm of the DOA vector, leading to what it is called activity-coupled Cartesian direction of arrival (ACCDOA) in [58].

In sound event localization and detection (SELD) techniques, it is frequently supposed that only one source of each type of event can be simultaneously present and a localization output is employed for each one of them. However, systems designed to locate several sources of the same class (e.g., several unknown speakers) have to face the source permutation problem [13], which is further explained in section 1.3.

Finally, it is worth saying that there are also some proposals where the neural networks are not used to directly estimate the DOAs, but to help other classical methods through time-frequency mask estimation [8, 59], signal dereverberation [60], or time difference of arrival estimation [48] for example.

# 1.2.4 Training and datasets

Although some unsupervised, semi-supervised, or weakly supervised approaches have been recently proposed [16, 46, 50], most of the deep learning-based sound source localization systems use supervised strategies where the train dataset includes the ground-truth DOA of the sound sources present in the recordings This leads us to another problem: the lack of large datasets of multichannel recordings with accurate spatial annotations.

To solve this issue, the image source method (ISM) [61] is typically used to simulate RIRs for different positions and in different acoustic environments and then signals taken from single-channel datasets are convolved with them to be used for training; speech corpora are typically used to obtain the source signals, but more diverse datasets can also be used [62]. Since we can simulate as many positions and scenarios as we want, this should allow us to have infinite-size training datasets, but, due to the time needed to execute the ISM, most proposals first generate a fixed-size dataset and then use it to train their models instead of generating the training signals on the fly, which has been proven to provide better results in other acoustic signal processing problems [63]. The use of synthetic signals for training usually generates a drop in the performance of the models when tested on actual real-world recordings and several techniques have been proposed in recent years to fix this issue [64, 65, 66].

# **1.3** Geometric deep learning

# **1.3.1** General concepts

Geometric Deep Learning (GDL) is an emerging concept in the deep learning community that attempts to propose a unifying framework for many well-known deep learning architectures and that also provides a new approach to design new architectures for new kinds of data [67]. Their main idea is that, although in multidimensional spaces the amount of data needed to learn a distribution grows exponentially with the number of dimensions (phenomena known as *the curse of dimensionality*), we can still work on them by exploiting their symmetries (i.e., those transformations that leave invariant the properties of interest).

A function f is said to be invariant to a group of transformations  $\mathfrak{G}$  if applying them to the function input does not affect the function output (i.e.,  $f(\mathfrak{g}(x)) = f(x) \forall \mathfrak{g} \in \mathfrak{G}$ ) and it is said to be equivariant if applying the transformation to the input leads to the same transformation in the output (i.e.,  $f(\mathfrak{g}(x)) = \mathfrak{g}(f(x)) \quad \forall \mathfrak{g} \in \mathfrak{G}$ ). Probably, one of the clearest and best-known examples of invariant and equivariant models that have succeeded in deep learning are the CNNs [68]. Bidimiensional convolutions are equivariant to translations and, according to the GDL ideas, this would explain the success of convolutional architectures in many computer vision problems that are invariant or equivariant to translations, such as image classification [69, 70] and image segmentation [71, 72] respectively.

Conventional CNNs can be generalized to group-equivariant convolutional neural networks (G-CNNs) [73] to exploit the equivariance to different types of symmetry groups. These symmetry groups can be discrete, such as discrete sets of rotations in the plane [74], in the space [75] or in 3D surfaces [76], or continuous, such as the group SO(3) of spherical rotations [77].

Another family of models that can be analyzed under the perspective of GDL are the graph neural networks (GNNs). The convolutional [78, 79], attentional [80, 81] and message-passing [82] approaches are all invariant to the permutations of their nodes and many other permutation-invariant models, such as Deep Sets [83] or the Transformers [84], can be seen as special cases of GNNs for fully connected graphs. Continuing with this GDL approach, we can find even modifications of the GNNs to include spatial information in their nodes and make them invariant both to their permutation and to spatial transformations [85].

# 1.3.2 Geometric deep learning for sound source localization and tracking

Since the success of 2D CNNs on computer vision tasks led to their use over audio spectrograms for many audio analysis and processing tasks, it makes sense to analyze if the translational equivariance of 2D convolutions is also an interesting property when analyzing audio spectrograms. For the time dimension, translational equivariance means equivariance to time shifts, which might be an interesting property in most of the audio processing tasks, however, in the frequency domain, the translational equivariance means equivariance to frequency shifts, whose interest is much more difficult to determine. In the case of sound source localization, the main source of information is usually considered to be in the phase difference between the signals, but the phase difference for a determined position depends on the frequency, so being equivariant to frequency shifts does not seem to be an interesting property for this task. In the case of multichannel signals, the spectrum of every channel is typically used as a different input channel for the CNNs, but we can find a different approach in [12]. In this work, the time is not taken into account and the convolutions are performed along the channel axis. Since they use uniform linear arrays, we can expect the phase differences for a given DOA to be the same between adjacent channels, and therefore being equivariant to shifts in this dimension can be positive according to the ideas of GDL, but this would not be true for other array geometries.

As commented in the previous section, several input features are usually combined in sound source localization systems by stacking them as additional input channels for the convolutional layers. This might make sense from the point of view of GDL when all of them represent spectral features with the same frequency scale, but this is not the case when combining features that do not share the same axis, such as spectrograms and generalized cross-correlation (GCC) coefficients, where the frequencies of the spectrograms that are analyzed with each GCC coefficients as the convolution kernels move along them are arbitrary. In any case, it is worth saying that architectures that do not follow the GDL ideas can still obtain good results; for example, good results were obtained in [39] by stacking spectrograms and GCC coefficients and applying 2D convolutions over them. This means that spectrograms and GCCs indeed contain important information for sound source localization, but GDL suggests that there are probably better architectures to analyze them.

As we have seen, most of the deep-learning models for sound source localization and tracking that we can find in the literature present equivariances that are not shared by their inputs and by the problem that they are intended to solve. However, there are at least two symmetries closely related to this problem that we can exploit to design sound source localization and tracking models: the rotational symmetry and the source permutation symmetry.

#### 1. INTRODUCTION

#### 1.3.2.1 Rotational symmetry

Since, in compact arrays, DOAs are typically represented in spherical coordinates, we could expect the sound source localization problem to exhibit a strong equivariance to spherical rotations. However, this symmetry is not easily exploited in many input representations, especially when working with arbitrary array geometries. In chapter 4, we propose the use of the steered response power with phase transform (SRP-PHAT) algorithm to obtain spatial representations of the acoustic scene where this symmetry becomes more obvious and it is easier to design network architectures that are able to exploit it. It is worth saying that, even if we used the SRP-PHAT algorithm, other kinds of spatial pseudo-spectrums could also be used, such as those computed with the multiple signal classification (MUSIC) algorithm, or even a combination of several techniques stacking different spatial representations as different input channels of the networks. Finally, spherical harmonics or Ambisonics representations could also be interesting for designing neural networks equivariant to the spherical rotations of the sources; some works have been recently proposed in this line [86, 87] but we do not explore this possibility in this work.

#### **1.3.2.2** Source permutation symmetry

While the rotational symmetry could be ignored and still obtain good localization results, when we want to locate multiple sound sources and we do not have any valid criteria to organize them (as it is done in SELD) we have to face the problem of the source permutation symmetry when training our models; i.e., we cannot expect our model to output the DOAs in the same arbitrary order that they have in the groundtruth data of our dataset [13]. The typical way to solve this is to compute the loss function using the permutation of the sources that generates a better match between the network output and the ground truth, but choosing a common permutation for the whole acoustic scene generates local minima in the loss function where the training usually get stuck and choosing a different permutation for each time instant does not encourage the model to keep the identities of every output stable (this issue is further explained in chapter 5). The source permutation symmetry has typically been seen as a problem when trying to train neural networks, but it also opens the door to the design of new network architectures to exploit it. In chapter 5, we propose both a new loss function to train multi-source trackers and a new permutation invariant recurrent layer where the information of every source is encoded in independent embedding vectors whose order does not affect the network output.

# **1.4** Main contributions and results of the thesis

The main contributions and results of this thesis are:

- 1. We present a new analytical method to cancel the effect of a source in the interchannel cross-correlation functions: We generalize previous narrowband cancellation techniques to the broadband case and present an efficient time-domain implementation that can eliminate the effect of a sound source from the interchannel cross-correlation functions. This new technique can be used to design iterative multi-source localization systems using these modified cross-correlation functions or other representations derived from them, such as the SRP-PHAT maps. We published this technique in [II] and we explain it in chapter 2.
- 2. We present the first free and open-source implementation of the image source with GPU acceleration: our python library gpuRIR reduces in two orders of magnitude the time needed to simulate room impulse responses using the image source method, which allows us to perform the simulations on the fly as we train our models, obtaining infinite-size datasets. This GPU implementation is explained in chapter 3 and we published it in [IV] and also presented it in [III].
- 3. We prove that SRP-PHAT power maps are robust input features that allow the models to obtain higher accuracy in highly reverberant scenarios: Compared with models using spectrogram as inputs, we prove that models using higher-level features such as SRP-PHAT power maps (or even GCCs) are more robust against reverberation. This is explained in chapter 4, whose results were published in [V].

#### 1. INTRODUCTION

- 4. We present the first free and open-source implementation of the icosahedral convolutional neural networks (icoCNNs): The icosahedral CNNs were originally proposed by Cohen et al. in [76] but they did not publish their implementation. We have published a PyTorch implementation which we believe will be useful for many researchers working with spherical signals.
- 5. We present a new output layer that combines classification and regression: We explain in chapter 5 how, interpreting them as probability distributions, we can use the spatial output of a CNN to obtain a regression output without needing to add any extra learnable parameters [VI].
- 6. We prove that the use of models with rotational symmetry can improve the localization accuracy even with coarser inputs: the use of power maps as inputs allows us to use models that are invariant to rotations and we show in chapter 4 that, as published in [VI], these models have higher localization accuracy even using lower resolution power maps.
- 7. We present a new permutation invariant training strategy which allows the training of tracking models while encouraging them to keep the identities of the inputs stable: while many of the loss functions and training strategies used to train tracking networks do not prevent identity switches in the models' outputs in order avoid local minima that make the training impossible, our new training strategy explained in chapter 5 (and published in [VII]) makes the model converge to solutions with lower identity switches without losing localization accuracy.
- 8. We present a new permutation invariant recurrent layer for sound source tracking: opposed to conventional recurrent units where the information of each source is represented in a unique state vector, our permutation invariant gated recurrent unit (PI-GRU) explained in chapter 5 split the information of every source in independent state vectors whose order does not affect the result of the model.

# 1.5 Outline and organization of the thesis

The reminder of this thesis is organized as follows:

- *Chapter 2* introduces the SRP-PHAT algorithm since we use them to compute the acoustic power maps that we use as input for our models in the rest of the thesis. In addition, we present the analytical method that we published in [II] to cancel the effect of a sound source in the inter-channel correlation functions of a microphone array, and that can be used to remove the effect of a source from the SRP-PHAT maps and therefore design iterative multi-source localization algorithms.
- *Chapter 3* explains the technique that we utilized to generate our training datasets, including our GPU implementation of the image source method [IV] and the algorithm that we used to generate synthetic moving sound sources.
- *Chapter 4* introduces our works on single source localization with neural networks using SRP-PHAT power maps as inputs. In section 4.1, we first present the 3D CNN published in [V] and then, in section 4.2, we present our model based on an icosahedral CNN [VI].
- *Chapter 5* explains our works on multi-source tracking. This includes both a new permutation invariant training strategy [VII] and our permutation invariant recurrent units.
- *Chapter 6* presents the conclusion of the thesis and discusses some future directions for sound source localization and tracking research.

 $\mathbf{2}$ 

# The SRP-PHAT power maps

In this chapter, we present an introduction to the classic signal processing techniques for direction of arrival (DOA) estimation, focusing on the SRP-PHAT power maps since we will use them as input representation for the deep learning models presented in the following chapters. Finally, we also present a new technique to cancel the effect of a sound source on the inter-microphone generalized cross-correlation (GCC) functions that allows us to design iterative algorithms for multi-source DOA estimation systems.

This chapter includes the reproduction of some figures and text fragments from [II] and [V] with the permission of the copyright holders.

# 2.1 Classic signal processing techniques for broadband DOA estimation

As proposed in [32] for the sound source localization (SSL) problem, the broadband DOA estimation strategies may be divided into three categories: techniques employing time difference of arrival (TDOA) information, strategies based on maximizing the steered response power (SRP) of a beamformer, and approaches adopting highresolution spectral estimation concepts. Some techniques have been proposed for specific geometric configurations, such as those based on the Spherical Harmonic Domain [88, 89] that can only be used in spherical arrays, but in this chapter we will focus on geometry-independent techniques.

TDOA-based locators follow two-step strategies where, in the first step, the TDOA between pairs of microphones is estimated and, in the second step, the most reliable DOA for those TDOAs is computed. The most common approach to the estimation of the TDOA between two microphones is finding the maxima of the generalized cross-correlation (GCC) function [90]. In audio applications, especially for speaker localization, the phase transform (PHAT) is typically used [91, 92, 93] due to its good performance in reverberant environments [94]. The two-step strategy greatly reduces the complexity of these algorithms, but the loss of information between the two steps (only the maximums of the GCCs are used for the DOA estimation) dramatically degrades their performance when noise or reverberation levels increase [95].

The SRP methods are based on the calculation of the output power of a beamformer steered towards different directions and the search for its maximum. Computing the filter-and-sum beamformer output for each direction is computationally exhaustive, but fortunately, as proved in [31, 32], the SRP can be computed in terms of GCCs. In this case, most of the computational cost of each beamformer is shared and the computational complexity of the algorithms increases slower with the search space. Another advantage of computing the SRP in terms of GCCs is that including the phase transform becomes trivial, which results in the steered response power with phase transform (SRP-PHAT) algorithm [31, 32]. Despite this, its complexity can also become excessive if the search space increases considerably (as in 3D SSL problems) and several efficient search algorithms [96, 97, 98], functional modifications to allow coarser grids [98, 99, 100] and GPU [101, 102], FPGA or ASIC [103] implementations have been proposed.

Finally, the high-resolution spectral-estimation-based locators use beamforming methods based on techniques like autoregressive (AR) modeling, minimum variance (MV), spectral estimation, etc. Some of the most commonly used narrowband algorithms, such as multiple signal classification (MUSIC) [29] or estimation of signal parameters via rotational invariant techniques (ESPRIT) [30, 104], are based on the signal and noise subspaces. After applying these narrowband techniques to each frequency bin of a broadband signal it is necessary to coherently combine the results of each bin [105] and many of these techniques are more sensitive to noise and reverberation [106] than the SRP-PHAT. In [107], an extension of the MUSIC algorithm for broadband signals based on the parameterized spatial correlation matrix (PSCM) or steered covariance matrix [108, 109] claims to outperform the SRP-PHAT algorithm even in high reverberant scenarios. This approach has a higher computational cost since, after computing the GCCs, the eigenvalue decomposition of the PSCM for each search direction must be computed.

Multi-source localization is another challenge in addition to robustness and computational efficiency, especially in wide dynamic range scenarios, where a more powerful source can mask a weaker one. Various techniques have been recently proposed to find multiple sources [110, 111, 112, 113, 114, 115], most assuming different degrees of time-frequency sparsity on the source signals. These sparsity constraints can be applied, for example, when speech sources need to be found [116] but cannot be applied with other sources like music or noise. Furthermore, these techniques are usually based on the incoherent application of subspace DOA estimation techniques in the time-frequency bins where just one source (or only a few sources) is present. This has two disadvantages: we lost the spatial information present in the bins where multiple sources are active and, on the other hand, subspace methods usually have high computational complexity and some of them may not work properly if correlation between sources exists. Finally, there are no studies about the performance of these techniques when the power level difference between sources increases.

Due to the non-stationary nature of most of the signals of interest, such as the speech or the music, a tracking stage is needed after the DOA estimation to exploit the temporal correlation between the source positions and to avoid inaccurate estimations in frames where the power of the signal is low or its auto-correlation makes the maximum of the power maps become too wide. The algorithms for one source tracking are typically based on the Kalman filter [117, 118] although more advanced techniques have been proposed to deal with multiple sources, such as those based on particle filtering [119, 120]. However, in these approaches, they use two-step strategies which make

them sensitive to potential information loss when only the DOA estimations are used for the tracking; e.g. the absolute maximum of the SRP-PHAT maps is always selected even if another local-maximum was much closer to the previous estimations and we assign the same likelihood to all the DOA estimations while some of them correspond to wider maximums from frames where the source was weaker. Including some of this information in the tracking algorithms may be possible, but it would increase both its complexity and the number of parameters that need to be fine-tuned. For example, in [121] a technique to share information between an iterative DOA estimator based on Expectation-Maximization [122] and a tracking system based on particle filtering is proposed.

# 2.2 The SRP-PHAT algorithm

The signal received at the  $n^{th}$  sensor of a microphone array in a room with  $N_s$  sources can be modeled as

$$x_n(t) = \sum_{i=1}^{N_s} a_i(t) * h_n(\boldsymbol{\theta}_i, t) + v_n(t), \qquad (2.1)$$

where  $a_i(t)$  is the signal generated by the source in the position  $\boldsymbol{\theta}_i$ ,  $h_n(\boldsymbol{\theta}_i, t)$  is the impulse response from  $\boldsymbol{\theta}_i$  to the  $n^{th}$  sensor, and  $v_n(t)$  is the noise of the sensor, which is typically supposed to be white, Gaussian, and uncorrelated with the source signal and with the noises of other sensors. It is worth mentioning that  $\boldsymbol{\theta}_s$  is written in bold because it can represent an angle, two spherical coordinates, or even a point in 3D Cartesian coordinates depending on the geometry of the array.

One of the most classic and popular approaches to DOA estimation is finding the peaks of the steered response power (SRP) maps that we would obtain using a filter-and-sum beamformer:

$$P(\boldsymbol{\theta}) = \int_{-\infty}^{+\infty} \left| \sum_{n=0}^{N-1} G_n(\omega) X_n(\omega) e^{-j\omega\tau_n(\boldsymbol{\theta})} \right|^2 d\omega, \qquad (2.2)$$

where N is the number of sensors of the array,  $X_n(\omega)$  is the Fourier transform of  $x_n(t)$ ,  $G_n(\omega)$  is the frequency response of the filter for the channel n, and  $\tau_n(\boldsymbol{\theta})$  is the time delay occurring from the position or direction  $\boldsymbol{\theta}$  to the  $n^{th}$  sensor. Although directly implementing (2.2) would be computationally expensive, it can be computed in terms of the GCC functions as

$$P(\boldsymbol{\theta}) = \int_{-\infty}^{+\infty} \left| \sum_{n=1}^{N} G_n(\omega) X_n(\omega) e^{-j\omega\tau_n(\boldsymbol{\theta})} \right|^2 d\omega = \sum_{n=1}^{N} \sum_{m=1}^{N} \int_{-\infty}^{+\infty} (G_n(\omega) X_n(\omega)) (G_m(\omega) X_m(\omega))^* e^{-j\omega\tau_n(\boldsymbol{\theta})} d\omega = 2\pi \sum_{n=1}^{N} \sum_{m=1}^{N} R_{nm}(\Delta\tau_{nm}(\boldsymbol{\theta})), \quad (2.3)$$

where  $\Delta \tau_{nm}(\boldsymbol{\theta}) = \tau_n(\boldsymbol{\theta}) - \tau_m(\boldsymbol{\theta})$  and  $R_{nm}$  is the GCC between the signals of the  $n^{th}$  and the  $m^{th}$  sensor:

$$R_{nm}(\tau) = \frac{1}{2\pi} \int_{-\infty}^{+\infty} \Psi_{nm}(\omega) X_n(\omega) X_m^*(\omega) e^{j\omega t} d\omega, \qquad (2.4)$$

where \* denotes the complex conjugate and  $\Psi_{nm}(\omega) = G_n(\omega)G_m^*(\omega)$  is a weighting function.

Equation (2.3), combined with the use of the phase transform (PHAT)  $G_n(\omega) = 1/|X_n(\omega)|$ , is commonly known as the SRP-PHAT algorithm [31, 32], and allows us to obtain an acoustic power map of the environment whose peaks should correspond with the position of the sources.

Although the SRP-PHAT algorithm is a good trade-off between robustness and computational efficiency, obtaining more accurate results than two-step TDOA based techniques with a lower computational cost than most of the broadband subspace techniques, it still presents several issues. The main advantage of (2.3) is that most of its computational cost is in computing the GCCs and does not increase with the search space. However, the computation of its sums for each direction, especially if it is needed to interpolate  $R_{nm}(\Delta \tau_{nm}(\boldsymbol{\theta}))$  from its adjacent samples, may not be negligible; this problem becomes more challenging when the search space is two-dimensional, e.g. the two angles of the spherical coordinates, or even three-dimensional, e.g. XYZ coordinates. Some search strategies have been proposed to reduce the number of evaluations of (2.3) that need to be computed to accurately find the maximum of  $P(\boldsymbol{\theta})$  [97, 98, 123] but, due to the non-convexity of the SRP-PHAT power maps, the



Figure 2.1: Example of SRP-PHAT power maps with different resolutions in a favorable scenario: SNR=30 dB and  $T_{60}$ =0.3 s. The red dot indicates the actual DOA of the sound source and the black dot is at the maximum of the map.

number of SRP-PHAT evaluations needed might still be an issue in some scenarios. In [98, 124], it is proposed to modify (2.3) to compute the power received from a space region instead of from a point, so they can use hierarchical search strategies over maps with lower resolution.

As we can see in Fig. 2.1, in favorable scenarios with high signal-to-noise ratio (SNR) and low reverberation, the SRP-PHAT power maps have a clear maximum in the DOA of the sound source that can be used to obtain a good estimation even with low-resolution maps but, when SNR decreases and the reverberation increases, such as in the scenario of Fig. 2.2, the maps present several local maxima that may be incorrectly interpreted as the DOA of the sound, especially when using low-resolution maps. However, in those maps, in addition to the maxima, we can also observe several patterns that are also related to the DOA of the sound and the geometry of the array and that may be exploited to obtain a more accurate DOA estimation.

Another limitation of the SRP-PHAT power maps is that, even if the phase transform is usually able to reduce the width of the maxima generated by the sources, weak



Figure 2.2: Example of SRP-PHAT power maps with different resolutions in an adverse scenario: SNR=5 dB and  $T_{60}$ =0.9 s. The red dot indicates the actual DOA of the sound source and the black dot is at the maximum of the map.

sources are still easily masked by stronger sources. Due to this reason, SRP-PHAT power maps are rarely used for multi-source SSL.

# 2.3 Source cancellation in Cross-Correlation functions

In this section, we present a new technique to eliminate the effect of a source from the inter-microphone cross-correlation (CC) functions or the GCCs in a broadband sensor array [II]. It can be applied to any array geometry and it does not need any time-frequency sparsity constraint, so the technique can work with white noise sources. In addition, this technique does not only eliminate the source in the desired direction but also the spurious maxima that appear when the sources are correlated. As it is a modification of the GCCs functions, most of the techniques and methods developed using the cross-correlation functions can be combined with it.

Our technique is based on a spatial correlation matrix (SCM) projection technique

widely employed in narrowband arrays [125, 126, 127], but we extend it to broadband sources. An efficient time-domain implementation is also presented and, if a lower computational cost is needed, we also propose a coarser approximation that can offer adequate results. Our technique is especially interesting in scenarios with correlation between sources and strong power differences between them.

In order to deal with multiple sources, the proposed technique can be used to iteratively find the strongest source, eliminate it, and find the following source. A similar idea has been recently published in [128] for DOA estimation of acoustic echoes, but it only deals with narrowband sources. A broadband technique is proposed in [129] to remove the effect of a source from the GCCs functions to locate multiple audio sources. However, it only suggests multiplying the GCCs by a de-emphasis function; they propose using a notch function with two parameters to control its sharpness. This approach lacks theoretical foundations and strongly depends on the choice of the proposed parameters, but [129] does not provide any technique to choose them.

#### 2.3.1 Narrowband source elimination

The complex snapshot received by an array of N narrowband sensors in a scenario with  $N_s$  sources can be written as

$$\mathbf{x} = \mathbf{D}\mathbf{a} + \mathbf{v},\tag{2.5}$$

where **a** is a vector of length  $N_s$  with the amplitudes of each signal, **v** is a zero-mean white noise vector of length N uncorrelated with **a**, and **D** is the steering matrix composed of the steering vectors of each source:

$$\mathbf{D} = [\mathbf{d}(\boldsymbol{\theta}_1), \mathbf{d}(\boldsymbol{\theta}_2), ..., \mathbf{d}(\boldsymbol{\theta}_{N_s})]$$
(2.6)

$$\mathbf{d}(\boldsymbol{\theta}_i) = [e^{-j\omega\tau_1(\boldsymbol{\theta}_i)}, e^{-j\omega\tau_2(\boldsymbol{\theta}_i)}, ..., e^{-j\omega\tau_N(\boldsymbol{\theta}_i)}]^T,$$
(2.7)

With this model, the complex SCM is

$$\mathbf{R} = E\{\mathbf{x}\mathbf{x}^H\} = \mathbf{D}E\{\mathbf{a}\mathbf{a}^H\}\mathbf{D}^H + \sigma_v^2\mathbf{I},\tag{2.8}$$

where  $E\{\cdot\}$  is the expectation operator, and the superscript H denotes Hermitian transpose,  $\sigma_v^2$  is the noise power and, if there is no correlation between the sources, the correlation matrix of signals  $E\{\mathbf{aa}^H\}$  is just a diagonal matrix with the power of each signal.

Several techniques have been proposed to prevent a strong source from hiding a weaker one in narrowband DOA estimation [125, 126, 127]. They are based on the projection of the correlation matrix onto a space orthogonal to the position  $\theta_0$  of the stronger source:

$$\mathbf{R}'(\boldsymbol{\theta}_0) = \mathbf{U}(\boldsymbol{\theta}_0)\mathbf{R}\mathbf{U}(\boldsymbol{\theta}_0), \qquad (2.9)$$

where  $\mathbf{U}(\boldsymbol{\theta}_0)$  is the projection matrix

$$\mathbf{U}(\boldsymbol{\theta}_0) = \mathbf{I} - \frac{\mathbf{d}(\boldsymbol{\theta}_0)\mathbf{d}^H(\boldsymbol{\theta}_0)}{\mathbf{d}^H(\boldsymbol{\theta}_0)\mathbf{d}(\boldsymbol{\theta}_0)}.$$
 (2.10)

After obtaining  $\mathbf{R}'(\boldsymbol{\theta}_0)$ , several narrowband localization techniques can be used to find the second source and, if there are still more sources to be found, apply (2.9) with  $\boldsymbol{\theta}_0$  equal to the position of the second source and then repeat the localization process.

## 2.3.2 Broadband source elimination

In broadband arrays, it is necessary to take into account the temporal characteristics of the signals and the effect of the reverberation as in the model followed in (2.1). The CC between the signals received in the  $n^{th}$  and the  $m^{th}$  sensors is defined in the frequency domain as

$$R_{nm}(t) = \frac{1}{2\pi} \int_{-\infty}^{+\infty} X_n(\omega) X_m^*(\omega) e^{j\omega t} d\omega$$
(2.11)

and the Fourier transform of the CC,  $\mathcal{F}\{R_{nm}(t)\} = \mathcal{S}_{nm}(\omega) = X_n(\omega)X_m^*(\omega)$ , is known as the cross-power spectral density (CPSD).

In order to get the CC functions with the source in the position  $\boldsymbol{\theta}_0$  canceled, we propose to apply (2.9) to each frequency bin of the CPSDs. The matrix multiplications

in (2.9) can be rewritten as:

$$S_{nm}^{SC}(\boldsymbol{\theta}_0, \omega) = \sum_{k=1}^{N} \left[ \sum_{l=1}^{N} \left[ U_{nl}(\boldsymbol{\theta}_0, \omega) S_{lk}(\omega) \right] U_{km}(\boldsymbol{\theta}_0, \omega) \right], \qquad (2.12)$$

where  $U_{nm}(\boldsymbol{\theta}_0, \omega)$  is the (n, m) entry of the matrix  $\mathbf{U}(\boldsymbol{\theta}_0)$  for the frequency  $\omega$ , obtained expanding (2.10):

$$U_{nm}(\boldsymbol{\theta}_0, \omega) = \begin{cases} e^{-j\omega\Delta\tau_{nm}(\boldsymbol{\theta}_0)}/N, & \text{if } n \neq m\\ (N-1)/N, & \text{if } n = m \end{cases}$$
(2.13)

From (2.12) we can expand the summations into three main terms:

$$S_{nm}^{SC}(\boldsymbol{\theta}_0,\omega) = \frac{(N-1)^2}{N^2} S_{nm}(\omega) - \frac{N-1}{N^2} \mathcal{P}_{nm}(\boldsymbol{\theta}_0,\omega) + \frac{1}{N^2} \mathcal{Q}_{nm}(\boldsymbol{\theta}_0,\omega), \qquad (2.14)$$

where

$$\mathcal{P}_{nm}(\boldsymbol{\theta}_0,\omega) = \sum_{k=1,\dots,N}^{k\neq n} \mathcal{S}_{km}(\omega) e^{-j\omega\Delta\tau_{nk}(\boldsymbol{\theta}_0)} + \sum_{l=1,\dots,N}^{l\neq m} \mathcal{S}_{nl}(\omega) e^{-j\omega\Delta\tau_{lm}(\boldsymbol{\theta}_0)}$$
(2.15)

and

$$\Omega_{nm}(\boldsymbol{\theta}_0,\omega) = \sum_{k=1,\dots,N}^{k\neq n} \sum_{l=1,\dots,N}^{l\neq m} \mathcal{S}_{kl}(\omega) e^{-j\omega(\Delta\tau_{nk}(\boldsymbol{\theta}_0) + \Delta\tau_{lm}(\boldsymbol{\theta}_0))}.$$
 (2.16)

Finally, through the inverse Fourier transform, we conclude that the CC between the  $n^{th}$  and the  $m^{th}$  sensor with the source in the position  $\theta_0$  removed is just the original CC and a linear combination of the CCs between each pair of sensors with different delays:

$$R_{nm}^{SC}(\boldsymbol{\theta}_0, t) = \frac{(N-1)^2}{N^2} R_{nm}(t) - \frac{N-1}{N^2} P_{nm}(\boldsymbol{\theta}_0, t) + \frac{1}{N^2} Q_{nm}(\boldsymbol{\theta}_0, t)$$
(2.17)

$$P_{nm}(\boldsymbol{\theta}_{0},t) = \sum_{k=1,...,N}^{k \neq n} R_{km} \left( t - \Delta \tau_{nk}(\boldsymbol{\theta}_{0}) \right) + \sum_{l=1,...,N}^{l \neq m} R_{nl} \left( t - \Delta \tau_{lm}(\boldsymbol{\theta}_{0}) \right)$$
(2.18)

$$Q_{nm}(\boldsymbol{\theta}_0, t) = \sum_{k=1,\dots,N}^{k \neq n} \sum_{l=1,\dots,N}^{l \neq m} R_{kl} \left( t - \Delta \tau_{nk}(\boldsymbol{\theta}_0) - \Delta \tau_{lm}(\boldsymbol{\theta}_0) \right).$$
(2.19)

With the term involving  $P_{nm}(\boldsymbol{\theta}_0, t)$  in (2.17), we are subtracting to each original CCs, i.e.  $R_{nm}$  (black star in Fig. 2.3), a delayed version of the CCs that have a sensor



Figure 2.3: A representation of the auto-correlation functions of 2 sources and 3 sensors. In order to remove the effect of the first source (whose autocorrelation is represented with a red triangle) from  $R_{12}(t)$  (indicated with a black star) we must use the CCs that have a sensor in common (indicated in blue circles) to compute  $P_{12}(t)$  and the CCs without any sensor in common (indicated with a green triangle) to compute  $Q_{12}(t)$ . See equations (2.17), (2.18) and (2.19) in the text.

in common, i.e.  $R_{n'm'}$  with n' = n or m' = m (blue circles in Fig. 2.3). The applied delay places the peak corresponding to the source we want to remove in the position  $\Delta \tau_{nm}(\boldsymbol{\theta}_0)$ . With the term involving  $Q_{nm}(\boldsymbol{\theta}_0, t)$ , we are adding the CCs without any sensor in common, i.e.  $R_{n'm'}$  with  $n' \neq n$  and  $m' \neq m$  (green triangles in Fig. 2.3), also placing in  $\Delta \tau_{nm}(\boldsymbol{\theta}_0)$  the peaks generated by the source in  $\boldsymbol{\theta}_0$ . This process is represented in Fig. 2.4.

In the presence of reverberation, the array does not only receive the source signals from its direct paths, but also from the DOA of the early reflections. Applying (2.17) with  $\theta_0$  equal to the position (or the direction) of a source, our technique does not remove the signals received due to its reflections; equivalently, with  $\theta_0$  equal to the DOA of a reflection, only that reflection is canceled. Not suppressing the reflections of the canceled source can be a disadvantage for source localization applications (although most tracking algorithms are designed to deal with the erroneous DOAs generated by the reverberation), but it is indispensable for the acoustic characterization of a room, where the goal is to locate the DOAs of its early reflections [130, 131].

# 2.3.3 Coarse approximation

Most of the computational complexity of (2.17) correspond to compute  $Q_{nm}(\boldsymbol{\theta}_0, t)$ , however, the elimination of the effect of the desired source happens in the subtraction



**Figure 2.4:** A representation of the terms  $P_{12}(t)$  and  $Q_{12}(t)$  in equations (2.17), (2.18) and (2.19). Following the picture in Fig. 2.3, in order to compute both  $P_{12}(t)$  and  $Q_{12}(t)$ , the delays applied to each CC function place the peak generated by the source we want to remove (the red triangle) in the position it was in  $R_{12}(t)$ .

of  $P_{nm}(\boldsymbol{\theta}_0, t)$  (although it is true that the CCs can have both positive and negative values, there will always be positive peaks at the positions of the sources). Therefore, we could get a strong reduction of the computational cost, i.e. the number of operations reduces from  $\mathcal{O}((N-1)^2)$  to  $\mathcal{O}(2(N-1))$ , by approximating (2.17) by:

$$R_{nm}^{SC}(\boldsymbol{\theta}_0, t) \approx R_{nm}^{SC-T}(\boldsymbol{\theta}_0, t) = \frac{(N-1)^2}{N^2} R_{nm}(t) - \frac{N-1}{N^2} P_{nm}(\boldsymbol{\theta}_0, t).$$
(2.20)

The error introduced by this approximation is further analyzed in section 2.4.2.

## 2.3.4 Phase Transform (PHAT)

As previously explained, the CC function (2.11) can be generalized to the GCC (2.4) by introducing a weighting function  $\Psi_{nm}(\omega)$  and, for SSL, the phase transform is typically used:

$$\Psi_{nm}^{PHAT}(\omega) = \frac{1}{|X_n(\omega)X_m^*(\omega)|} , \qquad (2.21)$$

which equally emphasizes all the frequencies for the DOA estimation and has been proved to approximate the Maximum Likelihood solution in low-noise reverberant environments [132] and is especially useful to deal with speech sources.

In order to correctly obtain the GCC with the source in the position  $\boldsymbol{\theta}_0$  removed, we should compute (2.14), apply the desired weighting function  $\Psi_{nm}(\omega)$ , and compute the inverse Fourier transform. To compute  $\Psi_{nm}(\omega)$  it would be necessary to get the signal  $X_n(\omega)$  without the source in the position  $\boldsymbol{\theta}_0$  but, in a real situation, we only have the signal with all the sources present. Therefore, it is impossible to obtain  $\Psi_{nm}(\omega)$ . To solve this issue and to exploit the benefits of the PHAT in reverberant scenarios, we propose to apply directly (2.17) by replacing the original CCs with the GCCs.

### 2.3.5 Examples

In order to better understand the effect of equations (2.17) and (2.20), in this section we present several examples with a linear array of 3 sensors.

#### 2.3.5.1 Two uncorrelated Gaussian sources

If the signals generated by 2 uncorrelated white Gaussian sources arrive at the sensors with delays  $\tau(\boldsymbol{\theta}_1) = [0, 10, 20]$  and  $\tau(\boldsymbol{\theta}_2) = [-5, 6, 17]$  samples in an anechoic environment, the CCs between sensors have 2 peaks, some of them in adjacent samples, as shown in Fig. 2.5 (a). If we apply the coarse approximation proposed in (2.20) to eliminate the source  $\boldsymbol{\theta}_0 = \boldsymbol{\theta}_1$ , the only remaining positive peaks are in  $\Delta \tau_{nm}(\boldsymbol{\theta}_2)$ and belong only to  $\boldsymbol{\theta}_2$  (Fig. 2.5 (b)), but they have strong negative peaks and break some properties of the CC functions, e.g. the maximum of  $R_{nn}(t)$  is not at  $\tau = 0$ . This could generate some artifacts in the power estimation, as negative values in the SRP maps, but if we only need to find the position of the main positive peak, the approximation may be good enough.

Finally, if we apply the complete expression, i.e. equation (2.17), the CCs still have only the desired positive peak, but they follow the properties of the Correlation functions and the negative peaks are reduced, as shown in Fig. 2.5 (c).

#### 2.3.5.2 Two impulsive sources

If we replace the Gaussian sources of the previous example with impulsive sources, we get 2 strongly correlated sources, that may be seen as an original source and a non-attenuated echo. In Fig. 2.6 (a), we can see how the correlation between sources generates additional peaks in the CCs that could lead to the estimation of spurious sources. However, our elimination technique, as shown in Fig. 2.6 (b) and Fig. 2.6



a) Original Cross-Correlation Functions

Figure 2.5: CCs between sensors in the scenario described in the example 2.3.5.1 (a), CCs after applying the coarse approximation to cancel the first source (b) and CCs after applying the complete equation of the proposed cancellation technique (c). The red circles are at  $\Delta \tau_{nm}(\boldsymbol{\theta}_1)$  and the red stars at  $\Delta \tau_{nm}(\boldsymbol{\theta}_2)$ .

(c) for the coarse approximation and the complete expression, removes both the peaks corresponding to the source in  $\theta_0 = \theta_1$  and the peaks generated by the correlation between sources, so correlated sources can be handled as if they were uncorrelated. This is especially useful for echo localization, where several correlated sources have to be found.

It is worth mentioning that, in both examples, we supposed that the first source was perfectly located and, since the auto-correlations of the sources are really narrow, small errors on it would have dramatically degraded its cancellation. However, the estimation of the sources is more accurate as the autocorrelation of the sources becomes narrower and, for wider autocorrelation, the accuracy of the estimation is not so critical since both the original source and the terms we are subtracting in (2.17) and (2.20) are wider. This effect is further studied in section 2.4.2.

#### 2.3.5.3 Two speech sources with reverberation

Finally, we used Lehmann's implementation [133] of the image source method (ISM) algorithm [61] to simulate a linear array with 5 microphones and 10 cm of intermicrophone distance in a room with reverberation time  $T_{60} = 0.1$  s; although this is a quite low reverberation, it allows us to check that our technique still works when the delay introduced by the propagation does not have a perfectly linear phase, i.e. each frequency component of the signal suffers a slightly different delay. Fig 2.7 (a) shows the CCs between the signals received by the microphones 1 and 5 when a speech source is placed in front of the array and another one at 45°; we can see how the source in front of the array (whose peak is indicated with a red circle) mask the effect of the second source (whose peak should be at the red star). As we can see in Fig. 2.7 (b), the truncated version of our cancellation technique removes too much energy from CCs and we need to apply the complete equation, Fig. 2.7 (c), to properly identify the peak corresponding to the second source.

We analyze higher reverberant rooms in section 2.4.3, but only locating peaks in the CCs is not enough to deal with them, so we need to employ more sophisticated DOA estimation algorithms.



a) Original Cross-Correlation Functions

Figure 2.6: CCs between sensors in the scenario described in the example 2.3.5.2 (a), CCs after applying the coarse approximation to cancel the first source (b) and CCs after applying the complete equation of the proposed cancellation technique (c). The red circles are at  $\Delta \tau_{nm}(\boldsymbol{\theta}_1)$  and the red stars at  $\Delta \tau_{nm}(\boldsymbol{\theta}_2)$ .



**Figure 2.7:** CC between sensors 1 and 5 in the scenario described in the example 2.3.5.3 (a) and the result of applying the coarse approximation (b) and the complete equation (c) over it. The theoretical positions of the peaks generated by the first and the second source are indicated with a circle and with a star, respectively.

# 2.4 An iterative multi-source localization algorithm based on source cancellation and SRP-PHAT

# 2.4.1 Algorithm

Several multiple-source localization algorithms may be proposed by combining the SRP-PHAT power maps with the source cancellation technique presented in the previous section. Here, we propose a simple iterative algorithm similar to the one used for narrowband arrays that, despite its simplicity, is able to prove the power of the proposed source cancellation technique.

The algorithm, detailed in Fig. 2.8, first computes the CCs (2.11) or GCCs (2.4) and uses them to compute the SRP or SRP-PHAT power map (2.3). Then, it iteratively searches for the strongest source and cancels it to compute a new power map. It also allows to iteratively apply the source cancellation in the same position if a new maximum is not found; however, as proved in sections 2.4.2 and 2.4.3, only a low



Figure 2.8: Iterative algorithm to find several sources using source cancellation and SRP techniques.

number of iterations, controlled by the IT\_MAX parameter in Fig. 2.8, is needed.

For the sake of simplicity, it is assumed that the number of sources (N\_SOURCES) is known, but it would be easy to modify the algorithm to dynamically obtain the number of sources. To do so, a threshold could be applied to the power of each new maximum to determine if it is a new source or if there are no more sources to be removed.

Implementing both (2.17) and (2.3) in a discrete-time system has the problem that  $\Delta \tau_{nm}(\boldsymbol{\theta}_0)$  may not be multiple of the sampling period. In order to deal with that, we found that using a linear interpolation between adjacent samples in (2.17) and just taking the closest one in (2.3) provides good enough results.

## 2.4.2 Simulations

#### 2.4.2.1 Uncorrelated white sources

Simulations were conducted to compare the performance of the conventional SRP method with our proposal. We carried out 1000 simulations with 1024 samples of 2 Gaussian sources (without correlation between them) in random positions, capturing the signals with a circular array with 16 sensors and a diameter of 0.5 meters in an anechoic environment. We repeated the 1000 simulations with 31 different power levels of the second source from  $-30 \,\mathrm{dB}$  to  $0 \,\mathrm{dB}$  relative to the first source. The signal-tonoise ratio (SNR) was always 10 dB with reference to the first source, i.e. the SNR was from  $-20 \,\mathrm{dB}$  to  $10 \,\mathrm{dB}$  with reference to the second source.

The SRP equation (2.3) was computed over a grid of spherical coordinates with 129 different values of polar angle  $\theta$  and 129 of azimuthal angle  $\phi$ , where the reference system  $(\hat{x}, \hat{y}, \hat{z})$  has its origin in the center of the array with  $\hat{z}$  normal to the array plane and  $\hat{x}$  pointing to the first sensor. After that, the two principal local maxima were located assuming a minimum separation of 2°.

To prove the performance of our source cancellation technique, the algorithm previously proposed (SRP SC) was used with  $IT_MAX = 2$ . In order to compute the probability of finding the second source, for both methods we assumed that the second source had been found if the localization error was fewer than 5°. In addition, we



**Figure 2.9:** Probability of finding the second source in a 2 random sources scenario with  $P_{s1} = 0 \,\mathrm{dB}$ ,  $\mathrm{SNR}_{s1} = -10 \,\mathrm{dB}$  and different values of  $P_{s2}$ .

repeated the simulation using the phase transform in the original GCCs (SRP-PHAT and SRP-PHAT SC) and with the truncation of (2.17) proposed in section 2.3.3: SRP SC(T) and SRP-PHAT SC(T). Finally, we implemented the source cancellation technique proposed in [129] (deeGCF). The authors did not provide any method to find the optimum parameters for their de-emphasis function, but for our array geometry we found that the best results were obtained for  $p \in [0.5, 1]$  and that b has a low impact on the performance of the algorithm; we finally used p = 0.75 and b = 2.

As shown in Fig. 2.9, our algorithm suffers less degradation when the power of the second source decreases and, surprisingly, truncating (2.17) makes our proposal even more robust. When combining the proposed source cancellation technique with the SRP algorithm, we can locate sources about 5 dB weaker than when using only the SRP algorithm, and the coarse approximation proposed in section 2.3.3 achieves an additional gain of 5 dB. Finally, the results show that, as could be expected, the phase transform does not generate great improvements for uncorrelated white sources since it does not modify white signals.

The truncated source cancellation technique (SC(T)) seems to perform better than

### 2.4 An iterative multi-source localization algorithm based on source cancellation and SRP-PHAT

the original (SC) because it removes more energy from the maps and sometimes applying the complete equation leaves residual energy around the source that may be identified as a new source. To solve this issue, it would be possible to modify the algorithm to not only repeat the source cancellation when the new maximum is in the same position as the original source, but also when it is closer than a threshold. Doing this would impose a trade-off between finding the weakest sources and the minimum necessary separation between them.

Multiplying the CC functions by a de-emphasis function as proposed in [129] gives a similar performance to the truncated version of our technique, but with our technique we do not need to fine-tune any parameters.

#### 2.4.2.2 Uncorrelated speech sources

In order to analyze the performance of the proposed technique with more complex sources, we replaced the white Gaussian sources with segments of 1024 samples randomly taken from a database of speech signals without silences sampled at 44100 Hz created by us from audio-book recordings in Spanish. We reduced IT\_MAX to 1 as increasing it did not improve the results.

As shown in Fig. 2.10, the algorithm with a better performance in this scenario is our source cancellation technique applied over GCCs with PHAT, followed by its coarse approximation. As expected for speech sources, the phase transform improves the performance of all algorithms.

Contrary to other cases, there is a non-negligible probability that the SRP algorithm without PHAT cannot locate any of the sources (about 25% for  $P_{s2} = P_{s1}$ ). This is due to the wide auto-correlation that some voice segments have, which leads to oversized peaks in the SRP maps. However, even in these cases where the estimation of the position of the first source is quite inaccurate, the proposed technique can eliminate the first source. Interestingly, the same issue that is responsible for the inaccurate localization of the first source, i.e. the wideness of the auto-correlation of the source, causes that the inaccuracy in the estimation of  $\theta_0$  does not lead to dramatic changes in the result of (2.17). Fig. 2.11 shows an example taken from a simulation where the second source was hidden by the first, but our technique was able to find it even after



**Figure 2.10:** Probability of finding the second source in a 2 speech sources scenario with  $P_{s1} = 0 \,\mathrm{dB}$ ,  $\mathrm{SNR}_{s1} = -10 \,\mathrm{dB}$  and different values of  $P_{s2}$ .

an inaccurate estimation of  $\boldsymbol{\theta}_0$ . In this situation, the cancellation technique proposed in [129] fails, since a notch function wide enough to eliminate the first source would also eliminate the second one.

The degradation SRP SC technique when the power of the second source approach the power of the first one is due to the fact that, when the two sources are too close, the second one is removed when the first one is canceled. This does not occur with SRP-PHAT SC since the phase transform narrows the wide of the peaks of the GCCs.

Although our cancellation technique is quite robust to errors in the estimation of the position of the first source, there are some cases where the two sources are too close and, if they have similar power, it removes both if used without PHAT. Applying the phase transform, it is possible to separate the two sources and, when our cancellation technique is applied, only the desired one is removed. However, we can see in Fig. 2.10 how SRP SC outperforms SRP-PHAT SC when the power of the second source decreases. Analyzing these cases, we found that in the original power maps (both with and without PHAT) the second source was completely hidden even when it was quite far from the first one. We also found that when applying the proposed technique, the



**Figure 2.11:** SRP power map resulting from the simulation of two speech sources with equal power and the SRP power map after applying our source cancellation technique at its maximum. The black circles indicate the actual position of the sources and the red circle the maximum of the map. The first source was successfully removed even after an inaccurate estimation of its position.

second source emerged from the SRP map but not from the SRP-PHAT map. This could be because the phase transform has been proven to approach the maximum likelihood solution in low-noise reverberant environments [132], but here the power of the second source is under the noise level ( $P_N = P_{s1} - 10 \text{ dB}$ ). An example of this situation is shown in Fig. 2.12.

With speech signals, it is not feasible to use the coarse approximation over GCCs without PHAT. This is due, again, to the wideness of the auto-correlation of the speech segments, causing the coarse approximation to remove too much information from the GCCs which needs to be recovered with the last term of (2.17). As expected, the de-emphasis function also has this problem; we tried to modify its parameters but we could not improve its results.

#### 2.4.2.3 Correlated speech sources

Finally, to test the effect of the correlation between sources, we repeated the previous simulation using the same signal (but with different power levels) for both sources, as if the second source was an echo of the first one. In this situation, the SRP without



Figure 2.12: SRP and SRP-PHAT power maps resulting from the simulation of two speech sources with  $P_{s2} = P_{s1} - 13 \text{ dB}$  and the SRP and SRP-PHAT power maps after applying our source cancellation technique. The black circles indicate the actual position of the sources and the red circle the maximum of the map. The first source was successfully removed both with and without PHAT, but the second source only appeared in the second case.



**Figure 2.13:** Probability of finding the second source in a 2 correlated speech sources scenario with  $P_{s1} = 0 \text{ dB}$ ,  $\text{SNR}_{s1} = -10 \text{ dB}$  and different values of  $P_{s2}$ .

PHAT is unable to find the first source and, even with PHAT, when the sources are too close and have similar power the maximum of the map is between them and it is impossible to locate them (this happens in about 30% of cases for  $P_{s2} = P_{s1}$ ).

As shown in Fig. 2.13, the SRP-PHAT algorithm can find the second source only if both sources have similar power, but with the proposed source cancellation technique we significantly improve the dynamic range of the algorithm. Due to the strong correlation between the sources, the SRP maps present a third maximum in the middle, which leads to the poor performance of SRP SC when both sources have similar power. Fortunately, if we use our cancellation technique on this spurious maximum it disappears without affecting the maxima that correspond to the sources, so it would be possible to find the two sources if our technique is applied twice. If we apply the cancellation technique to a maximum that corresponds to a source, the spurious maximum is also removed, so this could be exploited to differentiate the spurious maximum is also removed, so this could be exploited to differentiate the spurious maximum is attenuated, making it easier to find the second source simply by canceling the maximum of the original SRP-PHAT map. As with previous scenarios,



Figure 2.14: Probability of finding the second source in a 2 correlated speech sources scenario with  $P_{s1} = 0 \,\mathrm{dB}$ ,  $\mathrm{SNR}_{s1} = -10 \,\mathrm{dB}$  and different values of  $P_{s2}$  using a longer window size (K = 4096).

the technique proposed in [129] performs similarly to the truncated version of our algorithm.

To test the effect of increasing the window size, we repeated the simulation with speech segments of 4096 samples and achieved the results shown in Fig. 2.14. As expected, using longer window sizes improves the performance of all the algorithms, but the relations between them remain constant.

## 2.4.3 Recordings

Finally, some recordings were made in a highly reverberant room (see Fig. 2.15) in order to validate the effectiveness of the algorithm in real environments. The dimensions of the room were  $6 \text{ m} \times 8.5 \text{ m} \times 3 \text{ m}$  and we measured a reverberation time  $T_{60}$ of 1 second and a noise level of  $55 \text{ dB}_{\text{SPL}}$ . There was no acoustic foam in the walls or any other system to reduce reflections. Four loudspeakers were placed 80 cm apart, 4 meters in front of the array. The height of the array and the speakers were 1.05 m and 0.64 m, respectively.


Figure 2.15: Set-up of the recordings.

### 2.4.3.1 White sources

The power map in Fig. 2.16 (a) was obtained by playing white noise in only 2 loudspeakers, with 70 dB<sub>SPL</sub> and 65 dB<sub>SPL</sub> at the array position, and applying the conventional SRP-PHAT algorithm. After identifying the global maximum (which had an error of 2.04° regarding the first source position), the proposed technique (without truncation) was applied and we obtained the power map shown in Fig. 2.16 (b). The new global maximum was not the second loudspeaker, but the floor reflection of the first one. Applying our algorithm again to cancel this new direction, we obtained Fig. 2.16 (c), where the maximum is the reflection at the right wall. Finally, after canceling this direction, we found the second loudspeaker in Fig. 2.16 (d) with an error of only  $1.97^{\circ}$ . In order to locate the second source directly from the original SRP-PHAT map, we would have had to take its third local maxima as the second source (the second maxima was the floor reflection) which was at the same position as the maximum of Fig. 2.16 (d).

Fig. 2.17 and Fig. 2.18 show the same process for 3 and 4 loudspeakers, respectively. In all cases, we can see the ability of our algorithm to cancel the stronger sources, but sometimes some reflections are found before the weakest sources are located. In the first scenario (see Fig. 2.17), the localization error of each source were  $1.63^{\circ}$ ,  $2.01^{\circ}$  and  $0.81^{\circ}$ , while the second maximum in the original map was  $2.10^{\circ}$  apart from the second source and the peak closest to the third source was the 9<sup>th</sup> most energetic peak (so it would hardly have been considered a source if we did not know there had to be one in that direction) and its error was  $2.11^{\circ}$ . In the second scenario



Figure 2.16: Power maps obtained with SRP-PHAT (a) and applying our algorithm at the position of the maximum of the previous map (b, c, d), for 2 loudspeakers with  $70 \, dB_{SPL}$  and  $65 \, dB_{SPL}$  of white noise at the array position. The black circles are the actual positions of the loudspeakers and the red circle the maximum of each power map. After removing two reflections of the first source, we found the second source in (d).



Figure 2.17: Power maps obtained with SRP-PHAT (a) and applying our algorithm at the position of the maximum of the previous map (b, c, d), for 3 loudspeakers with  $75 \,dB_{SPL}$ ,  $70 \,dB_{SPL}$  and  $65 \,dB_{SPL}$  of white noise at the array position. The black circles are the actual positions of the loudspeakers and the red circle the maximum of each power map.



Figure 2.18: Power maps obtained with SRP-PHAT (a) and applying our algorithm at the position of the maximum of the previous map (b, c, d), for 4 loudspeakers with  $65 \,dB_{SPL}$  of white noise at the array position. The black circles are the actual positions of the loudspeakers and the red circle the maximum of each power map.

(see Fig. 2.18), the errors were 1.63°, 2.10°, 1.43° and 1.21° while the maximum of the original SRP-PHAT map closest to the sources was at the same positions, but to find the fourth source we would have needed to take into account the 6 strongest maxima.

### 2.4.3.2 Speech sources

Finally, we replaced the white noise recordings with speech recordings. Due to the non-stationary nature of the speech signal, the power generated by each loudspeaker may be strongly different in each 1024-sample frame, even after removing the silences from the recordings.

Fig. 2.19 and Fig. 2.20 had been obtained using 2 and 3 loudspeakers respectively and selecting a frame when the sources had power enough to locate all of them. In order to implement a complete speaker localization system, some tracking algorithm would be needed so it can deal with the fluctuations of the power of the sources and with the presence of estimated DOAs corresponding to reflections instead of sources (this issue is common to most of the DOA estimation algorithms when working with



**Figure 2.19:** Power maps obtained with SRP-PHAT (a) and applying our algorithm at the position of the maximum of the previous map (b, c, d), for 2 loudspeakers playing speech signals. The black circles are the position of the loudspeakers and the red circle the maximum of each power map.



**Figure 2.20:** Power maps obtained with SRP-PHAT (a) and applying our algorithm at the position of the maximum of the previous map (b, c, d), for 3 loudspeakers playing speech signals. The black circles are the actual positions of the loudspeakers and the red circle the maximum of each power map.

speech signals and this is the reason why tracking algorithms are implemented).

The localization error in Fig. 2.19 were  $1.04^{\circ}$  and  $0.38^{\circ}$ . The distance between the second source and the 4<sup>th</sup> peak of the original SRP-PHAT map (the closest one) was  $0.99^{\circ}$ , but it was surrounded by multiple maxima of similar power, so it would have been difficult to choose which of them are sources and which are not if we had not known the number of sources, while with our technique most of these maxima were cleaned. In Fig. 2.20, the localization error was  $3.84^{\circ}$ ,  $3.07^{\circ}$  and  $1.77^{\circ}$  while using only the SRP-PHAT algorithm we would have only been able to find two sources (with errors  $2.58^{\circ}$  and  $3.07^{\circ}$ ) since none of the 20 strongest peaks of the original map were close to the 3<sup>rd</sup> source.

We can see with these examples how our technique does not improve the accuracy of the location of the sources that could be found with the SRP-PHAT algorithm but makes it possible to find sources that were hidden in the original maps. The main issue in using our cancellation technique (or the proposed in [129]) for speaker localization applications would be dealing with the estimated DOAs which correspond to reflections of an already found source rather than the direct path of a new one. An approach to classify the reflections could be using a beamformer steered to each estimated DOA and studying the correlation between the sources received from each direction, but the computational cost would be prohibitive for most applications. A setup that could facilitate the classification of the reflections would be arranging multiple arrays in the room, so only the DOAs of the direct path from each source to each array would intersect at one point since each array would receive different reflections; another approach could be using near-field arrays whose spherical wave propagation model is more robust against wall reflections. Finally, a survey on the multi-source tracking algorithms should be performed to find those that better fit with the DOA estimates that our cancellation technique provides.

### 2.5 Conclusions

Sound source localization (SSL) is a problem that has been extensively studied using classical signal processing techniques and a broad range of techniques have been pro-

### 2. THE SRP-PHAT POWER MAPS

posed to solve it, each one with different advantages and disadvantages. Among them, SRP-PHAT provides an efficient algorithm to compute acoustic power maps whose peaks allow us to estimate the DOAs of the sound sources recorded with a microphone array. Despite having many advantages, like being more robust to reverberation than other techniques, SRP-PHAT still has some drawbacks, like the need of high resolution maps to ensure that their peaks really correspond to the DOAs, which increases its computational cost, and the issue of strong sources masking the weakest ones, which complicates its use for multi-source applications.

In order to solve this second issue, a new technique has been presented that allows broadband sensor arrays to cancel the effect of a source at an affordable computational cost. This technique is an extension of previous narrowband techniques with an efficient implementation that can be used iteratively to cancel several sources. It is compatible with other broadband techniques, like the SRP-PHAT, or with most of the multi-source tracking algorithms already proposed. It can be used in arrays with any geometry, no assumptions about the sources need to be made and it can deal with correlated sources.

The proposed algorithm is capable of locating multiple sources even when the second source is much weaker than the first one. The simulations made with a microphone array obtained an increase between 5 dB and 10 dB in the dynamic range in comparison with the conventional SRP-PHAT algorithm and the recordings demonstrate that the algorithm also works in real environments. This is especially important when the sources are not sparse and the weakest source will always be masked, both in time and frequency, by the strongest one. We have proved through simulations that the coarse approximation of our algorithm has a similar performance to the technique proposed in [129] but without needing to fine-tune any parameter and that, working with the complete equation, we clearly outperform it. This coarse approximation, and the technique proposed in [129], has a good performance for white sources but suffers a strong degradation when dealing with more complex sources like speech. It may be useful in applications where we have control over the sound sources, such as in the acoustic characterization of rooms, or when we want to locate noise sources that we already know are white, but for speaker localization applications the complete equation should be employed.

It should be noted that the new technique cancels the desired direction, its diffraction lobes, and the artifacts generated by the correlation between sources. But, when used in a reverberant real acoustic environment, it does not eliminate the room reflections generated by the same source. In cases where the aim is only to detect sound sources, further stages will be required in order to classify and detect the reflections. However, if the objective is to study the acoustics of a room, the reflections need to be located and, therefore, must not be eliminated.

### 3

# An infinite-size synthetic dataset for sound source localization and tracking

Due to the difficulty of obtaining an accurately hand-labeled dataset of moving sources recorded with microphone arrays, we opted to train our models with simulated signals. Another approach might have been using measured room impulse responses (RIRs) convolved with speech signals, but this would have reduced the amount of different acoustic conditions seen by the model during training, increasing the possibility of overfitting to those conditions and not generalizing.

Since we simulate our training signals from completely random parameters, we can simulate them as they are needed during the training obtaining an infinite-size dataset. This makes the training slower, but has two important advantages: i) we increase the different acoustic conditions that the models see during the training, which has proven to increase the model accuracy [63], and ii) we have higher flexibility to modify the probability distribution of the parameters of the simulation, such as the signal-to-noise ratio or the reverberation time, during training so we can perform curriculum learning strategies [134].

In this chapter, we first describe the procedure that we follow to simulate moving sources in section 3.1 and then we present the GPU implementation of the image source method (ISM) that we used to accelerate these simulations.

This chapter includes the reproduction of figures and text fragments from [V] and [IV] with the permission of the copyright holders.

### 3.1 Moving sources simulation

### 3.1.1 Trajectory generation

In order to generate trajectories that can be used to train tracking systems, we need to randomly generate continuous trajectory points that can be tracked, but we also need to ensure that they have enough diversity to avoid the network to learn how they are generated and overfit to them.

In order to do so, after randomly choosing the room dimensions, we randomly select two points within the room boundaries to be the starting  $(\mathbf{p}_0 = [p_{x0}, p_{y0}, p_{z0}]^T)$  and ending  $(\mathbf{p}'_L = [p'_{xL}, p'_{yL}, p'_{zL}]^T)$  points of the trajectory and add to the straight line that connects them a sinusoidal function in each axis with random frequencies  $(\boldsymbol{\omega} = [\omega_x, \omega_y, \omega_z]^T)$  and amplitudes  $(\mathbf{A} = [A_x, A_y, A_z]^T)$  ensuring that no more than 2 oscillations are performed during the trajectory in each axis and that the amplitude is low enough to avoid the source to exit the room:

$$\mathbf{p}_{i} = \mathbf{p}_{0} + \frac{i}{L-1}(\mathbf{p}_{L}' - \mathbf{p}_{0}) + \mathbf{A} \circ \sin(\boldsymbol{\omega} i), \qquad (3.1)$$

where L is the number of points of the trajectory,  $\circ$  stands for the pointwise product and the sin function also operates pointwise. Although the generation model is quite simple, it generates quite diverse trajectories (some examples are shown in Fig. 3.1) and, since the network only sees the azimuth and elevation coordinates and has a limited temporal perceptive field, the model should not overfit to it. In order to confirm this, we always test our models with the recordings of the LOCATA dataset (see section 4.1.4.3).

With this generation algorithm, we do not have direct control over the speed of the sources and it depends on the size of the rooms and the duration of the trajectories.



Figure 3.1: Examples of source trajectories used to train the models. The red dots are the trajectory points and the gray points represent the microphones.

The histogram in Fig. 3.2 shows the distribution of the source speed generated by the parameters employed in all the experiments presented in this thesis.

### 3.1.2 Trajectory simulation

As shown in Fig. 3.3, we use LibriSpeech utterances as sound sources. The LibriSpeech corpus [135] contains 960 hours of speech sampled at  $f_s = 16$  kHz extracted from audiobooks. Although audiobooks could be expected to contain quite clean speech signals, we found that some of them have a strong background noise that, after filtered by the RIRs, would be located in the same position as the source and would facilitate its localization and tracking in silent segments. To prevent our network learning to exploit this fact, which will not be present in actual recordings, we use the WebRTC voice activity detector (VAD) [136] to detect silent segments and clean them by completely removing the signal in those frames.

The sizes of the rooms are randomly selected from the range  $3 \text{ m} \times 3 \text{ m} \times 2.5 \text{ m}$  to  $10 \text{ m} \times 8 \text{ m} \times 6 \text{ m}$  and the array is randomly placed inside the room, being restricted to have a separation from the walls of a 10% of the room size in each dimension and



Figure 3.2: Distribution of the source speed in the synthetic dataset.



Figure 3.3: Dataset generation process. Italic letters represent variables and regular letters represent processes. Right-angled boxes represent deterministic processes and round boxes represent stochastic variables or processes.

be in the lower half of the room for the vertical axis. The signal-to-noise ratio (SNR) and reverberation time  $(T_{60})$  are also randomly selected from the ranges 5 dB to 30 dB and 0.2 s to 1.3 s respectively. Uniform distributions over the specified ranges are used for all the random parameters of the dataset.

In order to be able to evaluate the models trained with this synthetic dataset with actual recordings, we use the same microphone geometry as the microphone array with 12 sensors designed to be mounted over an NAO robot head employed in the LOCATA dataset [137]. The minimum and maximum inter-microphone distance of the array are 1.3 cm and 12.1 cm and the actual position of each microphone can be found in [138].

Once we have the position of the trajectory points and the microphones and the properties of the room, we can use the image source method (ISM) to obtain the RIRs for every trajectory position and convolve the source signal with them using the overlap-add method to obtain the sound signals received at the microphone positions. After that, we finally add an omnidirectional Gaussian noise to obtain the desired SNR. In order to compute the noise power needed to obtain the desired SNR, we compute the average energy of the sound signal during the whole trajectory excluding the silent segments (including the silent segments in the average would have led to lower noise levels for the same SNR).

### 3.1.3 Multi-source scenes

For the acoustic scenes with a varying number of sources employed to train the models presented in chapter 5, we need to model how new sources appear (born) and how existing sources disappear (die). In order to allow several sources to appear in the same time frame, we model the number of sources that are born in a time frame using a Poisson distribution:

$$P\{k \text{ new sources born in the time frame } t\} = \frac{e^{-\lambda}\lambda^k}{k!},$$
(3.2)

where  $\lambda$  is the rate of the distribution, which is equal to its mean and its variance. This is similar to the birth assumption typically used in multi-object trackers based on probabilistic models [139, 140] but, in order to control the maximum number of



Figure 3.4: Distribution of the number of sources in every time frame.

sources simultaneously active, we made the birth rate  $\lambda$  dependent on the number of sources active in the previous frame and truncated the probability of (3.2) to avoid exceeding the chosen maximum of sources. In the experiments shown in chapter 5, we allowed up to 3 sources simultaneously active and employed a birthrate of 0.06, 0.04, or 0.02 depending if the number of active sources in the previous frame was 0, 1, or 2.

Following again the typical assumptions used in multi-object trackers, we model the death probability of every source in every time frame with a Bernoulli distribution, but in order to avoid sources that lived too short, we added a minimum life length:

$$P\{\text{the active source } i \text{ dies in the time frame } t\} = \begin{cases} p & t - t_i^b > T_{min} \\ 0 & else \end{cases}, \quad (3.3)$$

where p is the death probability,  $T_i^b$  the time frame when the source *i* was born and  $T_{min}$  the minimum life of the sources. In the experiments shown in chapter 5 we used p = 0.02 and  $T_{min} = 200$  ms.

This model, with the parameters used in chapter 5, generates the distribution shown in Fig. 3.4 for the number of sources in every time frame.

# 3.2 Room impulse response simulation with GPU acceleration

The technique presented in the previous section allowed us to generate infinite acoustic scenes to train our models, but the simulation of the room acoustics was too slow to simulate them as they were needed during the training. This issue has typically been overcome by pre-generating enough acoustic scenes to train the models, but that would have meant fixing the size of the dataset. Oppositely to this approach, we opted by exploiting the power of the graphics processing units (GPUs) to implement a faster version of the image source method (ISM).

### 3.2.1 Introduction

The simulation of the acoustics of a room is needed in many fields and applications of audio engineering and acoustic signal processing, such as training robust speech recognition systems [141] or training and evaluating sound source localization [142] or speech enhancement [143] algorithms. Although there are many low-complexity techniques to simulate the reverberation effect of a room in real time, such as the classic Schroeder Reverberator [144], some applications require an accurate simulation of the reflections causing the reverberation. The information of all those reflections is gathered in the room impulse response (RIR) between the source and the receiver positions, which allows the simulation of the reverberation process by filtering the source signal with it. The image source method (ISM) is probably the most used technique for RIR simulation due to its conceptual simplicity and its flexibility to modify parameters such as the room size, the absorption coefficients of the walls, and the source and receiver positions. We can simulate any level of reverberation by modifying the room size and the absorption coefficients, but the computational complexity of the algorithm grows fast as the number of reflections to simulate increases.

Initially developed to support the graphics computations of video games, graphics processing units (GPUs) are today one of the best and cheapest ways to increase the speed of many algorithms that can be expressed in a parallel form. Despite parallelizing most of the stages of the ISM is quite straightforward, to the best of our knowledge,

	RIR generator [146]	pyroomacoustics [147]	[133]	[145]	gpuRIR
Open source library (language)	✓(Matlab and Python)	✓(Python)	✓(Matlab)	X	✓(Python)
Implementation language	C++	Python and C++	Matlab	CUDA	CUDA
Fractional delays	1	1	1	1	1
Negative reflection coefficients	x	X	1	X	1
Diffuse reverberation model	x	x	1	X	1
GPU acceleration	x	X	x	1	1
Lookup table implementation	x	1	x	X	1
Mixed precision implementation	X	X	X	×	1

Table 3.1: Comparison of some state-of-the-art ISM implementation.

only [145] proposed to implement it in GPUs. Although they showed that using GPUs it was possible to speed up the RIR simulations, they did not provide the code of their implementation and the acoustic signal processing and audio engineering communities have not embraced their approach. In addition, they used an overlap-add strategy with atomic operations to combine the contributions of each image source, which strongly reduces the level of parallelism. In this chapter, we present a new GPU implementation with a higher degree of parallelism, which allows us to achieve higher speed-ups with cheaper GPUs. Motivated by the performance boost obtained with the use of lookup tables (LUTs) in the CPU implementations, we also study its use in our GPUs implementation. Finally, we propose a 16-bit precision implementation which can increase even more the simulation speed in the newer GPUs with mixed precision support.

Table 3.1 shows some state-of-the-art implementations of the ISM and compares some of their main characteristics. We can see how our implementation is the only one with GPU acceleration that is available as a free and open source library and how it includes some features (further explained in section 3.2.2.2) that are not included in other Python libraries. Using our library does not require any knowledge about GPU programming, but just having a CUDA compatible GPU and the CUDA Toolkit, and it can be installed and used as any CPU RIR simulation library.

### 3.2.2 The Image Source Method (ISM)

The method of images has been widely used in many fields of physics to solve differential equations with boundary conditions, but its application for RIR estimations was originally proposed by Allen and Berkley [61]. In this section, we first review



Figure 3.5: Image sources for a two-dimensional room. The red square and the blue dot represent the receiver and the source and the blue circumferences represent the image sources. The solid green line represents one of the multiple reflection paths and the dashed green line is the direct path of the equivalent image source. The black dot is the origin of the coordinates system.

their original algorithm and then explain some of the improvements that have been proposed to improve both its accuracy and computational performance.

### 3.2.2.1 Original Allen and Berkley algorithm

The main idea behind the ISM is to compute each wave-front that arrives at the receiver from each reflection off the walls as the direct path received from an equivalent (or image) source. In order to get the positions of these image sources, we need to create a 3D grid of mirrored rooms with the reflections of the room in each dimension; as shown in Fig. 3.5 simplified to 2D for an example.

If the number of images we want to compute for each dimension is  $N_x$ ,  $N_y$  and  $N_z$ , then we define a grid  $\mathbb{N}$  of image sources  $\mathbf{n} = (n_x, n_y, n_z) : \lceil -N_x/2 \rceil \leq n_x < \lceil N_x/2 \rceil, \lceil -N_y/2 \rceil \leq n_y < \lceil N_y/2 \rceil$  and  $\lceil -N_z/2 \rceil \leq n_z < \lceil N_z/2 \rceil$  (where  $\lceil \cdot \rceil$  stands for the round toward positive infinity operator). The coordinates of the position of each image  $\mathbf{p_n} = (x_n, y_n, z_n)$  are calculated using its grid indices, the position of the source

and the dimensions of the room; as an example, the component x would be calculated as

$$x_{\mathbf{n}} = \begin{cases} n_x L_x + x_s & \text{if } n_x \text{ is even} \\ (n_x + 1)L_x - x_s & \text{if } n_x \text{ is odd} \end{cases},$$
(3.4)

where  $\mathbf{L} = (L_x, L_y, L_z)$  is the size of the room and  $\mathbf{p}_s = (x_s, y_s, z_s)$  is the position of the original source. The y and the z coordinates can be obtained similarly.

The distance  $d_{\mathbf{n}}$  from the image source  $\mathbf{n}$  to a receiver in the position  $\mathbf{p}_r = (x_r, y_r, z_r)$ , and therefore the delay of arrival  $\tau_{\mathbf{n}}$ , is trivial if we know the image source position:

$$d_{\mathbf{n}} = ||\mathbf{p}_r - \mathbf{p}_s||, \tag{3.5}$$

$$\tau_{\mathbf{n}} = \frac{d_{\mathbf{n}}}{c},\tag{3.6}$$

where  $|| \cdot ||$  denotes the Euclidean norm and c is the speed of sound.

In order to calculate the amplitude with which the signals from each image source arrive at the receiver, we need to take into account the reflection coefficients of the walls of the room. We define  $\beta_{x0}$  as the reflection coefficient of the wall parallel to the x axis closest to the origin of the coordinates system and  $\beta_{x1}$  as the farthest;  $\beta_{y0}$ ,  $\beta_{y1}$ ,  $\beta_{z0}$  and  $\beta_{z1}$  are defined equivalently. Finally, if we define  $\beta_{\mathbf{n}}$  as the product of the reflection coefficients of each wall crossed by the path from the image source **n** to the receiver, its amplitude factor will be

$$A_{\mathbf{n}} = \frac{\beta_{\mathbf{n}}}{4\pi \cdot d_{\mathbf{n}}}.\tag{3.7}$$

Knowing the amplitude and the delay for each image, we can easily obtain the RIR as the sum of the contribution of each image source:

$$h(t) = \sum_{\mathbf{n}\in\mathcal{N}} A_{\mathbf{n}} \cdot \delta(t - \tau_{\mathbf{n}}), \qquad (3.8)$$

where  $\delta(t)$  is the Dirac impulse function.

#### 3.2.2.2 Improvements to the original algorithm

#### Fractional delays

In order to implement (3.8) in the digital domain, we need to deal with the fact that the values of  $\tau_{\mathbf{n}}$  may not be multiples of the sampling period. The original algorithm proposed to just approximate the fractional delays by the closest sample, however, the error introduced by this approximation is too high for some applications, such as sound source localization (SSL) with microphone arrays. In [148], Paterson proposed to replace the Dirac impulse function with a sinc windowed by a Hanning function:

$$\delta'(t) = \begin{cases} \frac{1}{2} \left( 1 + \cos \frac{2\pi t}{T_{\omega}} \right) \operatorname{sinc}(2\pi f_c t) & \text{if} - \frac{T_{\omega}}{2} < t < \frac{T_{\omega}}{2} \\ 0 & \text{otherwise} \end{cases},$$
(3.9)

where  $f_c$  is the cut-off frequency,  $T_{\omega}$  is the window length, and the sinc function is defined as  $\operatorname{sinc}(x) = \operatorname{sin}(x)/x$ . This is motivated by the low pass anti-aliasing filter that would be used if the RIR was recorded with a microphone in the real room. A window duration of  $T_{\omega} = 4 \,\mathrm{ms}$  and a cut-off frequency equal to the Nyquist frequency, i.e.  $f_s/2$ , are typically used.

Using the Paterson approach with  $T_{\omega} = \infty$  is equivalent to computing (3.8) in the frequency domain as the sum of complex exponential functions as proposed in [149] [150], but using shorter window lengths reduces the computational complexity of the algorithm.

#### Negative reflection coefficients

Using positive reflection coefficients as proposed in [61] generates a low-frequency artifact that must be removed using a high-pass filter. In addition, while a RIR recorded in a real room has both positive and negative peaks, all peaks generated by the ISM are positive. Using negative reflection coefficients as proposed in [150] solves both problems without the need for adding any posterior filter to the ISM algorithm.

#### **Diffuse reverberation**

In order to properly simulate a RIR, we need to use values of  $N_x$ ,  $N_y$ , and  $N_z$  high enough to get all the reflections that arrive in the desired reverberation time. Since

the delays of the signals of each image source are proportional to their distance to the receiver, and the distance is to the image index, the number of images to calculate for each dimension grows linearly with the reverberation time, and, therefore, the number of operations in (3.8) grows in a cubic way.

A popular solution to allow the simulation of long reverberation times in a reasonable time is decomposing the RIR in two parts: the early reflections and the late, or diffuse, reverberation. While the early reflections need to be correctly simulated with the ISM method to avoid losing spatial information, the diffuse reverberation can be modeled as a noise tail with the correct power envelope. In [133], Lehmann and Johansson propose using noise with logistic distribution and the technique introduced in [151] to predict the power envelope.

Although the technique presented in [151] generates better predictions of the power envelope obtained in real rooms, its computational complexity is quite high. Therefore, for the sake of computational efficiency, we decided to use a simple exponential envelope following the popular Sabine formula [152]. According to this model, the reverberation time  $T_{60}$  that takes for a sound to decay by 60 dB in a room, is

$$T_{60} = \frac{0.161V}{\sum S_i \alpha_i},$$
(3.10)

where V is the volume of the room and  $S_i$  and  $\alpha_i = 1 - \beta_i^2$  are the surface area and the absorption coefficient of each wall<sup>1</sup>; and the power envelope of the RIR is

$$P(t) = \begin{cases} A \exp\left(\log_{10}\left(\frac{T_{60}}{20}\right)(t-t_0)\right) & \text{if } t > t_0\\ 0 & \text{otherwise} \end{cases}.$$
 (3.11)

Therefore, knowing  $T_{60}$ , we can easily estimate A from the early reflections simulated with the ISM and then multiply the logistic-distributed noise by  $\sqrt{P(t)}$  to simulate the diffuse reverberation.

<sup>&</sup>lt;sup>1</sup>It should be noted that, as done in [61], we are defining the absorption ratio  $\alpha$  as a quotient of sound intensities (energies) while the reflection coefficient  $\beta$  is defined as a quotient of pressures (amplitudes).

CUDA functions	Description	Time $(\%)$
calcAmpTau_kernel	Equations $(3.6)$ and $(3.7)$	0.68
$generateRIR\_kernel$	Since computation and initial sum $(3.8)$	90.34
$reduceRIR_kernel$	Parallel sum $(3.8)$	1.07
envPred_kernel	Power envelope prediction	0.03
generate_seed_pseudo	cuRAND function (diffuse reverberation)	7.78
gen_sequenced	cuRAND function (diffuse reverberation)	0.01
diffRev_kernel	Diffuse reverberation computation	0.01
CUDA memcpy	[CPU to GPU]	0.00
CUDA memcpy	[GPU to CPU]	0.06

Table 3.2: Kernels and functions of the CUDA implementation.

### 3.2.3 Parallel implementation

As shown in Fig. 3.6, the parallel computation of the delays and the amplitudes of arrival for the signals from each image source and their sinc functions is straightforward since there are not any dependencies between each image source, and computing RIRs for different source or receiver positions in parallel is also trivial. However, the parallelization of (3.8) involves more problems, as the contributions of all the image sources need to be added into the same RIR.

It is worth mentioning that, though it would be possible to compute RIRs from different rooms in parallel, we choose to implement only the parallelization of RIRs corresponding to the same room. This was because the number of image sources to be computed depends on the room dimensions and the reverberation time and, to compute different rooms in parallel, we would have needed to use the worst-case scenario (i.e. the smallest room and higher reverberation time) for all of them, which would have decreased the average performance.

In order to implement the ISM in GPUs, we decided to use the Nvidia<sup>TM</sup>'s Compute Unified Device Architecture (CUDA) [153] and divide our code into the kernels<sup>1</sup> listed in Table 3.2. For illustrative purposes, we show in Table 3.2 the average proportion of time employed by each kernel to compute a standard case of 6 RIRs with  $T_{60} = 1$  s using the ISM method for the 250 first milliseconds and the diffuse model for the

<sup>&</sup>lt;sup>1</sup>A CUDA kernel is a function that, when is called, is executed N times in parallel by N different CUDA threads in the GPU. For more details, see the CUDA programming guide: https://docs.nvidia.com/cuda/cuda-c-programming-guide/



Figure 3.6: ISM parallel implementation. Our library actually computes some of the since sequentially, which leads to more efficient memory use. The reduction sum is detailed in Fig. 3.7

following 750ms using a Nvidia<sup>TM</sup> GTX 980Ti. It can be seen how the bottleneck is located at the beginning of the computation of (3.8), which is due to the high amount of sinc functions that are needed to be computed. The following sections provide further details about the implementation of the different parts of the algorithm.

#### 3.2.3.1 Amplitudes and delays computation

For computing (3.6) and (3.7), we use calcAmpTau\_kernel, which computes sequentially each RIR but parallelizes the computation for each image source. Although parallelizing the computations for each RIR would have been possible, since  $N_x \cdot N_y \cdot N_z$ is generally greater than the number of RIRs to compute, the level of parallelization is already quite high and, as shown in Table 3.2, further optimizations of this kernel would have had a slight impact on the final performance of the simulation.

#### **3.2.3.2** Computation and sum of the contribution of each image source

The computation of (3.8) is the most complex part of the implementation as it implies a reduction operation (the sum of the contributions of each image source into the final RIR), which is hard to parallelize since it would imply several threads writing in the same memory address, and the calculation of a high number of trigonometric functions. We can see it as creating a tensor with 3 axes (each RIR, each image source, and each time sample) and summing it along the image sources axis. However, the size of this tensor would be huge and it would not fit in the memory of most GPUs.

To solve this problem, we first compute and sum a fraction of the sources contributions sequentially, so the size of the tensor we need to allocate in the GPU memory is reduced; we do that through generateRIR\_kernel. Specifically, each parallel thread of this kernel performs sequentially the sum of 512 images for a time sample of a RIR. This sequential sum reduces the degree of parallelism of the implementation but, since the number of threads is already high enough to keep the GPU always busy, it does not decrease the performance. It should be noted that, although all the threads can potentially run in parallel, the number of threads which actually run in parallel is limited by the number of CUDA cores of the GPU and, if we have more threads than CUDA cores, many threads will be queued and will run sequentially.



Figure 3.7: Parallel reduction sum of the since (each level is performed by a call to reduceRIR\_kernel). The sum must be performed pairwise to prevent several threads from concurrently writing in the same variable. The sums of each time sample are also performed in parallel.

After that, we use **reduceRIR\_kernel** recursively to perform the reduction in parallel by pairwise summing the contribution of each group of images as shown in Fig. 3.7. Performing the whole sum in parallel would lead to all the threads concurrently writing in the same memory positions, which would corrupt the result.

It can be seen in Table 3.2 how most of the simulation time is expended in generateRIR\_kernel, this is due to the high amount of sinc functions that need to be computed and it also happens in the sequential implementations. However, thanks to the computing power of modern GPUs, we can compute many sinc functions in parallel and therefore reduce the time we would have needed to sequentially compute them in a CPU. We analyze the implementation of these sinc functions using lookup tables (LUTs) in section 3.2.3.5 and its performance in section 3.2.5.2.

#### 3.2.3.3 Diffuse reverberation computation

For the diffuse reverberation, we first use envPred\_kernel to predict in parallel the amplitude and the time constant of each RIR. After that, we use the cuRAND library included in the CUDA Toolkit to generate a uniformly distributed noise (the functions generate\_seed\_pseudo and gen\_sequenced in Table 3.2 belong to this library) and we finally transform it to a logistic distributed noise and apply the power envelope through diffRev\_kernel, which parallelizes the computations of each sample of each RIR. The function generate\_seed\_pseudo generates the seed for the cuRAND random number generator and it is only called when the library is imported, not every time a new RIR is calculated.

### 3.2.3.4 Simulating moving sources

In order to simulate a moving source recorded by a microphone array, we need to compute the RIR between each point of the trajectory and each microphone of the array and filter the sound source by them using the overlap-add method. In sequential libraries, the complexity of the filtering is negligible compared to the RIR simulation; however, in our library, thanks to the performance of the GPUs, we found that we also needed to parallelize the filtering process if we did not want to be limited by it (especially for short reverberation times). To solve this problem, our library is able to compute multiple convolutions in parallel using the cuFFT library (included in the CUDA Toolkit) and a custom CUDA kernel to perform the pointwise complex multiplication of the FFTs.

#### 3.2.3.5 Lookup tables (LUTs)

Motivated by the performance increase that the CPU implementations achieve by using lookup tables (LUTs) to calculate the sinc functions (see section 3.2.5), we also implemented it in our GPU library.

Our LUT stores the values of a sinc oversampled in a factor Q = 16 multiplied by a Hanning window:

$$LUT[n] = \frac{1}{2} \left( 1 + \cos \frac{2\pi n}{QT_{\omega}} \right) \operatorname{sinc} \left( \pi \frac{n}{Q} \right) \quad \text{for } n \in \left\{ \frac{-T_{\omega}}{2} Qf_s, \dots, \frac{T_{\omega}}{2} Qf_s \right\}$$
(3.12)

and then we use linear interpolation between the closest entries of the table to compute each sample of the sinc functions of each image source.

The main design choice we must make is to define the type of memory that will be used to place the LUT. CUDA GPUs have, in addition to the registers of each thread, 4 different memories: shared, global, constant, and texture memory. On the one hand, shared memory is shared only between threads of the same block and it has the fastest access, however, it is generally lower than 100 kB. On the other hand, global memory is shared by all the threads and usually has several gigabytes, but it has the lowest bandwidth and the highest latency. Finally, constant and texture memories are read-only cached memories, constant memory being optimized for several threads accessing the same address and texture memory being optimized for memory access with spatial locality. Although constant memory has a lower latency than texture memory, texture memory implements some features like several accessing modes and hardware interpolation, which are extremely useful for the implementation of LUTs. We implemented the windowed sinc LUT both in shared memory and texture memory and obtained better performance with the texture memory thanks to the hardware interpolation.

### 3.2.3.6 Mixed precision

Since the Pascal architecture, the Nvidia<sup>TM</sup> GPUs include support for 16-bit precision floats and are able to perform two 16-bit operations at a time. To exploit this feature, we developed the kernels generateRIR\_mp\_kernel and reduceRIR\_mp\_kernel, which compute two consecutive time samples at a time so we can halve the number of threads needed. We focused on these kernels and did not optimize the others because, as shown in Table 3.2, most of the simulation time is spent on them.

CUDA provides the data type half2, which contains 2 floating point numbers of 16 bits, and several intrinsics to operate with it. These intrinsics allow us to double the number of arithmetic operations that we can perform per second; however, we found that the functions provided to compute two 16-bit trigonometric functions were not as fast as computing one 32-bit function. To increase the simulation speed, we developed our own sinpi(half2) and cospi(half2) functions. For the sine function, we first reduce the argument to the range [-0.5, 0.5], then we approximate the sine function in this range by

$$\sin(\pi x) \approx 2.326171875x^5 - 5.14453125x^3 + 3.140625x \tag{3.13}$$

and finally, multiply the result by -1 if the angle was in the second or the third quadrant. The coefficients of the polynomial are the closest numbers that can be represented with half-precision floats to those of the optimal polynomial in a leastsquares sense. Equivalently, for the cosine function, we used the polynomial:

$$\cos(\pi x) \approx -1.2294921875x^6 + 4.04296875x^4 - 4.93359375x^2 + 1 \tag{3.14}$$

with the advantage that, since we only used it for computing the Hanning window in (3.8), we do not need to perform argument reduction or sign correction.

The polynomial evaluation can be efficiently performed with Horner's method:

$$b_n = a_n$$
  
 $b_{n-1} = a_{n-1} + b_n x$   
...  
 $p(x) = b_0 = a_0 + b_1 x$ 
(3.15)

where  $a_i$  is the coefficient of the *n* degree polynomial p(x) we want to evaluate and the computation of  $b_i$  can be done in parallel for two different values of *x* using the CUDA intrinsic \_\_hfma2(half2) that performs the fused multiply-add operation of the two elements of three half2 variables at a time. More information about the polynomial approximation of transcendental functions can be found in [154].

Obviously, working with half-precision representation reduces the accuracy of the results. We found that the most critical part was in subtracting  $t - \tau_n$ . Working with 16-bit precision floats, we can only represent 3 significant figures accurately, so, when t grows, we lose precision in the argument of the sinc function which leads to an error that increases with time; when t grows we expend the precision in the integer part and we do not represent accurately the fractional part. To solve this issue, we perform the subtraction with 32 bits arithmetic and then we transform the result to 16-bit

precision. Working this way, we have always maximum precision in the center of the sinc and the lower accuracy is outside the Hanning window.

Unfortunately, the hardware interpolation of the texture memory does not support 16-bit arithmetic, so the mixed precision implementation is not compatible with the LUT.

### 3.2.4 Python library

We have included the previous implementation in a Python library<sup>1</sup> that can be easily compiled and installed using the Python packet manager (pip) and be used as any CPU library. The library provides a function that takes as parameters the room dimensions, the reflections coefficients of the walls, the position of the source and the receivers, the number of images to simulate for each dimension, the duration of the RIR in seconds, the time to switch from the ISM method to the diffuse reverberation model, and the sampling frequency and it returns a 3D tensor with the RIR for each pair of source and receiver positions. Information about the polar pattern of the receivers and their orientation can be also included in the simulation.

We also provide some python functions to predict the time when some level of attenuation will be reached, to get the reflections coefficients needed to get the desired reverberation time (expressed in terms of  $T_{60}$ , i.e. the time needed to get an attenuation of 60 dB), and to get the number of image sources to simulate in each dimension to get the desired simulation time without loss reflections. Finally, we include a function to filter a sound signal by several RIRs in order to simulate a moving source recorded by a microphone array. In the repository of the library, some examples can be found about how to simulate both isolated RIRs and moving sources.

Since the use of the LUT to compute the sinc function improves the performance in most of the cases and the precision loss is negligible (see section 3.2.5.2), its use is activated by default, but the library provides a function to deactivate it and use the CUDA trigonometric functions instead. In order to exploit the mixed precision capabilities of the newer GPUs, it has a function to activate it and use the 16-bit

 $<sup>^1 \</sup>rm The$  code, the documentation, the installation instructions, and examples can be found in <code>https://github.com/DavidDiazGuerra/gpuRIR</code>

precision kernels instead of the 32-bit; activating it automatically deactivates the use of the LUT.

### 3.2.5 Results

#### 3.2.5.1 Base implementation

In order to show the benefits of using GPUs for RIR simulation, we have compared our library against three of the most employed libraries for this purpose: the Python version of the RIR Generator library presented in [146], whose code is freely available in [155] and has been used, for example, in [143, 156, 157]; the Python package pyroomacoustics presented in [147] that has been employed in [158, 159, 160] among others; and the Matlab<sup>TM</sup> library presented in [133], whose code is freely available in [161], and that has been used, for example, in [142, 162, 163]. Since all the libraries are based on the ISM, whose acoustical accuracy is well known, we focus on the computation time of each library.

Neither RIR Generator nor pyroomacoustics implement any kind of diffuse reverberation model, so they are expected to have worse performance than the Matlab<sup>TM</sup> library and our GPU library if we use it. The Matlab<sup>TM</sup> library uses the formula presented in [151] to model the power envelope of the diffuse reverberation, which is more complex than our exponential envelope model, so, for the sake of a fairer comparison, we modified the Matlab<sup>TM</sup> implementation to use an exponential model. The simulations with the sequential libraries and the ones with the Nvidia<sup>TM</sup> GTX 980Ti and the RTX 3090 were performed in a computer with an Intel<sup>TM</sup> Core i7-6700 CPU and 16 GB of RAM, while the simulations with the Nvidia<sup>TM</sup> Tesla P100 and V100 were performed in an n1-highmem-4 instance in the Google Cloud Platform<sup>TM</sup> with 4 virtual CPUs cores and 26 GB of CPU memory; more details about the GPUs employed for the simulations can be found in Table 3.3.

Fig. 3.8 represents the runtime of the different libraries for computing different numbers of RIRs in a room with size  $3 \text{ m} \times 4 \text{ m} \times 2.5 \text{ m}$  and  $T_{60} = 0.7 \text{ s}$ . It can be seen how our library can simulate a hundred times more RIRs in a second than the Matlab<sup>TM</sup> library even with a GPU designed for gaming (the Nvidia<sup>TM</sup> GTX 980

GPU model	Release year	Architecture	Memory	Single Precision FLOP/s	Memory Bandwidth
GTX 980 Ti	2015	Maxwell	$6 \mathrm{GB}$	5.6 TeraFLOP/s	$337~\mathrm{GB/s}$
Tesla P100	2016	Pascal	16 GB	9.5 TeraFLOP/s	732  GB/s
Tesla V100	2017	Volta	16 GB	14.9  TeraFLOP/s	$900 \ \mathrm{GB/s}$
RTX 3090	2020	Ampere	24 GB	29.3  TeraFLOP/s	$936~\mathrm{GB/s}$

 Table 3.3: GPUs employed for the performance analysis.



**Figure 3.8:** Runtime of each library for computing different numbers of RIRs ( $M_{src}$ ) in a room with size  $3 \text{ m} \times 4 \text{ m} \times 2.5 \text{ m}$  and  $T_{60} = 0.7 \text{ s}$ . For the gpuRIR library, the solid line times were obtained with the GTX 980 Ti GPU, the dashed lines with the Tesla P100, the dotted lines with the Tesla V100, and the dash-dot lines with the RTX 3090.

Ti). Using our library without any kind of diffuse reverberation modeling, we have a similar execution time to the Matlab<sup>™</sup> library, which only computes the ISM until the RIR has an attenuation of 13 dB, and we are also about a hundred times faster than the RIR Generator library. Finally, it is worth noting how pyroomacoustics performs quite similarly to our library when we use a GTX 980 Ti and compute the whole RIR with the ISM without using any diffuse reverberation model; this is due to the use of LUTs to compute the sinc functions by pyroomacoustics (to confirm this hypothesis we modified the code of pyroomacoustics to avoid the use of LUTs and its performance degraded to the same results than RIR Generator). However, using a faster GPU, i.e. the Tesla V100 or the RTX 3090, our library can compute ten times more RIRs in a second than pyroomacoustics even without using LUTs, since we can set at full performance all the parallelization mechanisms presented in section 3.2.3.

Comparing the performance of our library using different GPUs, we can see how the lower results are obtained using the GTX 980 Ti and the Tesla P100 and the best results are obtained with the Tesla V100 and the RTX 3090 (being between 5 and 10 times faster than the GTX 980 Ti). It is worth noting how, in less than 5 years, a gaming-oriented GPU as the RTX 3090 has matched and ever surpassed the performance of the professional GPU Tesla V100. We can expect the number of RIRs that we can compute per second with our parallel implementation to continue increasing at a good peace, something that we cannot expect from most of the sequential implementations designed to run in CPUs.

In Fig. 3.9 we show the runtime of the different libraries for computing 128 RIRs in a room with size  $3 \text{ m} \times 4 \text{ m} \times 2.5 \text{ m}$  and different reverberation times. We can see again how our library is about two orders of magnitude faster than the sequential alternatives which do not use LUTs. It must be said that our library has some limitations because calculating a large number of RIRs with high reverberation times may require more memory than what is available in the GPU; however, using the diffuse reverberation model, this limitation appears only for really high number of RIRs and reverberation times. Furthermore, it would be always possible to batch the RIRs in several function calls to circumvent this problem.



Figure 3.9: Runtime of each library for computing 128 RIRs in a room with size  $3 \text{ m} \times 4 \text{ m} \times 2.5 \text{ m}$  and different reverberation times. For the gpuRIR library, the solid line times were obtained with the GTX 980 Ti GPU, the dashed lines with the Tesla P100, the dotted lines with the Tesla V100, and the dash-dot lines with the RTX 3090.

Number of RIRs		Diffuse reverberation model					Full ISM		
			1	16	128	1024	1	16	128
Ma	tlab I	Library [ms]	$221,\!52$	$1,\!643.20$	$12,\!252.67$	96,208.58	-	-	-
pyroom acoustics [ms]		-	-	-	-	242.35	$3,\!6409.16$	$28,\!646.86$	
	Τ:	Base [ms]	4.98	17.43	117.60	898.54	283.88	2,601.82	19,630.60
	GTX 980	LUT [ms]	5.19	16.64	109.38 x1.08	834.38	279.28	2,434.33	18,547.03
gpuRIR		${\mathop{\rm MP}\limits_{\rm speedup}} [{\mathop{\rm ms}\limits_{ m speedup}}]$	-	-	-	-	-	-	_
	00	Base [ms]	5.81	13.86	79.28	596.02	115.57	1,661.35	12,879.31
	a P1	$\underset{\rm speedup}{\rm LUT} [ms]$	$\underset{\scriptscriptstyle \rm x0.97}{5.97}$	$\underset{\scriptscriptstyle \mathrm{x1.14}}{12.14}$	$\underset{\scriptscriptstyle \rm x1.24}{63.90}$	$\underset{\scriptscriptstyle \rm x1.27}{471.16}$	$\underset{\scriptscriptstyle \times 1.33}{86.86}$	$1,\!235.64_{\scriptscriptstyle {\rm x1.35}}$	9,397.40
	Tesh	$\mathop{\rm MP}_{\rm speedup}[ms]$	$\underset{\mathrm{x1.05}}{5.52}$	$9.45 \\ \scriptstyle \times 1.47$	$\underset{\scriptscriptstyle \times 1.74}{45.49}$	$\underset{\scriptscriptstyle \mathrm{x1.84}}{324.12}$	$\underset{\scriptscriptstyle \mathrm{x1.94}}{59.46}$	$\underset{\scriptscriptstyle \times 1.96}{847.74}$	$6,\!$
	00	Base [ms]	4.76	7.13	28.14	195.69	37.62	447.04	3,403.60
	V1	LUT [ms]	5.01	6.79	23.66	156.91	30.66	394.54	2,595.97
	Tesla	${\mathop{\rm MP}\limits_{ m speedup}}\left[{\mathop{\rm ms}\limits_{ m speedup}} ight]$	4.55 x1.05	6.29 x1.13	19.57 x1.44	128.72 x1.52	21.76 x1.73	253.03 x1.77	1,900.52
	06	Base [ms]	5.95	8.04	23.27	139.41	46.75	351.10	2,638.73
	ξ 306	$\mathop{\rm LUT}_{\rm speedup}[{ m ms}]$	6.02	$\underset{\mathrm{x0.99}}{8.12}$	$\underset{\scriptscriptstyle \times 1.09}{21.42}$	$\underset{\scriptscriptstyle \mathrm{x1.10}}{127.14}$	$\underset{\scriptscriptstyle \rm x1.33}{35.02}$	$\underset{\scriptscriptstyle \times 1.06}{331.02}$	$2,\!$
	RTY	$\mathop{\rm MP}_{\rm speedup}[{\rm ms}]$	$\underset{\mathrm{x1.04}}{5.69}$	$\underset{\scriptscriptstyle \times 1.14}{7.06}$	$\underset{\mathrm{x1.43}}{16.22}$	$\underset{\scriptscriptstyle \times 1.65}{84.33}$	$\underset{\scriptscriptstyle \rm x1.68}{27.79}$	$\underset{\scriptscriptstyle \mathrm{x1.74}}{200.90}$	$1,\!$

**Table 3.4:** Lookup table (LUT) and mixed precision (MP) simulation times and speedups for computing different numbers of RIR with  $T_{60} = 0.7$  s.

#### 3.2.5.2 Lookup tables

Motivated by the huge speedup generated by the use of LUTs in the CPU implementations (a factor 5 in Fig. 3.8) we replaced the trigonometric computations with a LUT as described in section 3.2.3.5. Tables 3.4 and 3.5 show the speedup (defined as the runtime without using the LUT divided by the runtime using it) for several numbers of RIRs and reverberation times using different GPUs.

We can see how our library obtains a speedup much lower than the one obtained by pyroomacoustics over CPUs. This is due to the high computation power of the GPUs, which makes the computation of trigonometric functions quite efficient and therefore they are not so benefited by replacing computation tasks with memory calls. Despite that, we can see how using LUTs is faster than computing the trigonometric functions, i.e. the speedup is higher than 1.0, in most of the cases, especially when the number of RIRs or the reverberation time increases.

Among the studied GPUs, the Tesla P100 obtains the higher speedups since it has

	$T_{co}$ [s]	Diffuse reverberation model					Full ISM		
- 00 [5] -		0.3	0.7	1.1	1.5	1.9	0.3	0.7	1.1
Mat	lab Library [ms]	2,573.67	12,078.52	39,330.40	94,946.73	$136,\!522.39$	-	-	-
pyro	omacoustics [ms]	-	-	-	-	-	1,854.08	$23,\!253.22$	$90,\!960.54$
Ë	Base [ms]	8.90	118.00	627.15	2,016.40	5,073.48	731.59	19,657.05	-
980	LUT [ms]	$\frac{8.62}{x1.03}$	$\underset{\scriptscriptstyle \times 1.07}{109.89}$	$\underset{\scriptscriptstyle \times 1.06}{588.90}$	$1,\!896.48$	4,769.68	$\underset{\scriptscriptstyle \times 1.05}{694.10}$	$18,\!466.15$	-
tIR Tesla P100   GTX	$\mathop{\mathrm{MP}}_{\mathrm{speedup}}[\mathrm{ms}]$	-	-	-	-		-	-	-
	Base [ms]	8.97	80.78	416.57	1,349.47	$3,\!289.13$	494.33	12,875.14	76,383.87
	LUT [ms]	$\underset{\scriptscriptstyle \rm x1.21}{7.39}$	$\underset{\scriptscriptstyle \times 1.25}{64.90}$	$\underset{\scriptscriptstyle \times 1.29}{321.97}$	$1,\!023.39$	$2,\!$	$\underset{\scriptscriptstyle \rm x1.26}{391.39}$	$9,\!406.31$	$55,\!402.08$
	$\mathop{\rm MP}_{\rm speedup}[{\rm ms}]$	$\underset{\mathrm{x1.35}}{6.64}$	$\underset{\mathrm{x1.79}}{45.18}$	$\underset{\scriptscriptstyle \rm x1.91}{218.18}$	$\underset{\mathrm{x1.93}}{699.95}$	$1,\!698.17_{\scriptscriptstyle \rm x1.94}$	$\underset{\scriptscriptstyle \rm x1.91}{258.96}$	$6,\!$	$38,\!393.03$
Ind 00	Base [ms]	5.80	28.81	125.02	379.13	896.97	141.86	3,400.95	19,935.71
g RTX 3090 Tesla V10	LUT [ms]	5.95	$\underset{\scriptscriptstyle \times 1.18}{24.43}$	101.85	$\underset{\scriptscriptstyle \mathrm{x1.14}}{332.35}$	$\underset{\scriptscriptstyle \rm x1.30}{690.02}$	$\underset{\scriptscriptstyle \rm x1.22}{117.55}$	$2,\!$	$15,\!363.05$
	$\mathop{\mathrm{MP}}_{\mathrm{speedup}}[\mathrm{ms}]$	$\underset{\scriptscriptstyle \rm x1.14}{5.08}$	$\underset{\scriptscriptstyle \times 1.46}{19.80}$	$\underset{\scriptscriptstyle \times 1.63}{\textbf{76.71}}$	$\underset{\scriptscriptstyle \times 1.72}{220.66}$	$\underset{\scriptscriptstyle \mathrm{x1.73}}{519.46}$	$\underset{\mathrm{x1.62}}{87.48}$	$1,\!901.21_{_{\rm x1.79}}$	$\underset{_{\rm x1.80}}{11,052.52}$
	Base [ms]	6.02	24.31	96.60	295.95	694.41	112.20	2,585.85	$15,\!498.97$
	LUT [ms]	6.09	22.38	$88.22 \\ \times 1.09$	273.19	$\underset{\mathrm{x1.09}}{639.95}$	103.85	2,380.35	14,225.71
	${\mathop{\rm MP}\limits_{ m speedup}}[{\mathop{\rm ms}\limits_{ m speedup}}]$	$5.80$ $_{\times 1.04}$	$\underset{\scriptscriptstyle \mathrm{x1.41}}{17.21}$	$57.32$ $_{\rm x1.69}$	170.21	$401.21 \\ {}_{\rm x1.73}$	63.81 x1.73	$1,\!477.57$	8,544.19

**Table 3.5:** Lookup table (LUT) and mixed precision (MP) simulation times and speedups for computing 128 RIRs with different reverberation times.

a higher memory bandwidth compared with its computing power. The GTX 980 Ti gets really humble speedups due to its low memory bandwidth and the Tesla V100 and the RTX 3090, though they have the higher bandwidth, do not reach the speedups obtained by the Tesla P100 due to their huge computing power.

Fig. 3.10 shows the first 0.5 seconds of the RIR of a room with  $T_{60} = 1$  s computed with our GPU implementation working with single (32-bit) precision trigonometric functions and the error introduced by replacing them with our LUT. We can see how, as could be expected, the error introduced by the use of the LUT is negligible: three orders of magnitude lower than the amplitude of the RIR.

#### 3.2.5.3 Mixed precision

In the case of using the 16-bit precision kernels, we are reducing the accuracy of the simulation, so we need to analyze its impact. Fig. 3.10 also shows the error introduced by computing the same RIR using our half (16-bit) precision kernels. We can see how the error is 3 orders of magnitude lower than the amplitude of the RIR at



**Figure 3.10:** RIR computed with single (32-bit) precision trigonometric functions and the errors introduced due to computing it using a lookup table (LUT) and half (16-bit) precision functions (Mixed Precision).

the beginning, which should be acceptable for most of the applications; however, since the error does not decrease with time as much as the RIR does, the signal-to-error ratio deteriorates with the time. Fortunately, this higher error corresponds with the diffuse reverberation, where its perceptual importance is lower.

Theoretically, a twofold speedup could be expected from working with 16-bit precision floats instead of 32-bit floats, however, this speedup is generally not reachable as the number of operations is not the only limiting factor of many GPU kernels and some half2 functions are not as fast as their equivalent single functions. Tables 3.4 and 3.5 show the speedup that our mixed precision implementation achieves for several numbers of RIRs computed in parallel and several reverberation times. We can see how the speedup is higher when the workload increases, especially for long reverberation times where the operations per second are the main limiting factor of its performance and how the speedup achieved with the mixed precision implementation is always higher than the achieved with the LUTs.

The mixed precision support was introduced with the Pascal architecture, so it is not available in older models like the GTX 980 Ti. The Tesla P100 achieves speedups really close to 2 for high workloads and the Tesla V100 and the RTX 3090 also achieve better speedups than with the use of LUTs although not as high as the Tesla P100.

### 3.3 Conclusions

Due to the difficulty of recording a hand-labeled dataset of moving sources large enough to train a neural network, we have introduced a new procedure for generating random trajectories and simulating them as they are needed for training. With it, we have an infinite-size dataset whose parameters can be easily modified during training to accelerate the convergence or during test to analyze the performance of the model in specific scenarios. To reduce the simulation time and allow the generation of the trajectories as they are needed during the training, we have implemented the image source method (ISM) using CUDA so we can reduce the simulation time by two orders of magnitude using GPUs.
We have published our GPU implementation of the ISM as a free and open-source library that has proven to be about one hundred times faster than other state-of-theart CPU libraries. To the best of our knowledge, it is the first library with these features freely available on the Internet, and it could allow the acoustic signal processing community, for example, to generate huge datasets of moving speaker speech signals in a reasonable computation time or to compute the acoustics of a virtual reality scene in real time.

We have studied different methods to increase the speed of our GPU implementation, concluding that the best strategy is using 16-bit arithmetic, but this is only compatible with the newer GPUs. On the other hand, using LUTs stored in the GPU's texture memory, though it generates lower speedups, is compatible with most of the CUDA GPUs, so we have chosen to use this implementation as our library default.

4

# Robust single source localization

In this chapter we present two sound source localization (SSL) models for scenarios with a single source, both using steered response power with phase transform (SRP-PHAT) power maps as inputs and being completely causal; i.e. they could be implemented in real-time and provide a new direction of arrival (DOA) estimate for every new input map. Both are fully convolutional models, the first one is based on a 3D convolutional neural network (CNN) and the second one combines icosahedral convolutions with 1D temporal convolutions.

Using both synthetic scenarios and real recordings, we prove that the use of SRP-PHAT power maps as input features of our models provides more robust estimations under adverse conditions and that the use of network architectures that fit the symmetries of the problem allows us to improve the localization accuracy even reducing the number of trainable parameters and the resolution of the input maps.

This chapter includes the reproduction of figures and text fragments from [V] and [VI] with the permission of the copyright holders.

## 4.1 Robust sound source localization using SRP-PHAT and 3D convolutional neural networks

As explained in chapter 2, we can use the SRP-PHAT algorithm to compute acoustic power maps from the signals received by a microphone array. These maps have traditionally been used for SSL by just choosing the position of their maximum as the estimated DOA but, even being more robust than other classic techniques, they still have some limitations especially when reverberation and noise increase. In this section, we propose the use of CNNs over SRP-PHAT power maps, performing the convolutions over the dimensions of the maps and the temporal dimension.

Any kind of SRP-PHAT power maps could be employed with this approach depending on the geometry of the array but, as we focus on compact arrays, we use 2D spherical power maps and therefore, since we include the temporal dimension, 3D CNNs. The extension to 4D CNNs over 3D SRP-PHAT maps to perform 3D SSL with distributed arrays would be straightforward.

Using conventional CNNs over equiangular projections of spherical maps is not the optimal solution since these projections generate deformations around the poles that need to be learned by the network instead of just being part of the equivariances of its architecture. However, this is still a good first approach to process this kind of maps with neural networks due to its simplicity and efficient implementation. In section 4.2 we study the use of icosahedral convolutions as an extension of this work.

Many of the state-of-the-art CNN architectures include bidirectional recurrent units at the last layers of the model. Recurrent neural networks (RNNs), as recurrent linear filters, make the output at any time instant dependent on the values of the input at every previous time instant and, therefore, applying them in the backward direction is extremely non-causal. Obviously, any tracking system can greatly benefit from the information on the future positions of the source but, in order to make our system feasible for real-time applications, we opt for using only causal convolutional layers. By using 3D convolutions, the localization is done including the information of the previous time frames, which does not happen in most of the classical systems where the localization is done using only the information of the current time frame and then the localization result is filtered in a posterior tracking stage considering the localization results of the previous frames. Integrating the information in this way should lead to more robust estimates since we do not lose any information between the localization and tracking stages.

#### 4.1.1 Preprocessing

The proposed model takes as input a 4-dimensional tensor (**M**) with size  $C \times T \times N_{\theta} \times N_{\varphi}$ , whose first channel  $\mathbf{M}_{1,t,i,j}$  contains SRP-PHAT power maps. In order to compute them, we first window the signals of a microphone array sampled at 16 kHz using Hanning windows of length K = 4096 samples (i.e. 256 ms) with a hop size of 3K/4. After that, for the T resulting frames, we apply (2.3) at a rectangular grid with  $N_{\theta}$  equispaced polar angles in the range  $\theta \in [0, \pi]$  and  $N_{\varphi}$  equispaced azimuth angles in  $\varphi \in [-\pi, \pi)$  to obtain the power map of each window; for planar arrays, the same model could be used sampling the polar angle only in  $\theta \in [0, \frac{\pi}{2}]$ . Finally, we normalize the maps by subtracting their mean and dividing them between their maximum to approximately fit them to the range [-1,1]. For the sake of computational efficiency, we do not perform any kind of interpolation in the computation of (2.3) and just approximate the fractional delays to the nearest sample.

Although the model must learn more complex patterns in order to exploit all the information available in the SRP-PHAT, it is obvious that one of the main sources of information about the DOA of the source is the position of the maximum of each map. Since it did not cause a significant increase in the computational complexity of the system, we decided to explicitly indicate to the network the position of the maximum of each map. After trying to introduce this information in different layers, we found that the best results were obtained including it in the input of the network, using C = 3 with  $\mathbf{M}_{2,t,i,j} = \hat{\theta}_t^{SRP}$  and  $\mathbf{M}_{3,t,i,j} = \hat{\varphi}_t^{SRP}$  for any  $t \in \{1, ..., T\}$ ,  $i \in \{1, ..., N_{\theta}\}$ , and  $j \in \{1, ..., N_{\varphi}\}$ , where  $\hat{\theta}_t^{SRP}$  and  $\hat{\varphi}_t^{SRP}$  are the polar and azimuth angles corresponding to the position of the maximum of the maximum of the maximum of the range [0,1].

Finally, we found that, since the synthetic dataset presented in chapter 3 does not include any directional noise, the models trained with it are very sensitive to directional noise sources. For example, in some of the recordings of the LOCATA dataset [137], the noise of a fan is present and, although its power is very low, the models tracked it when it was the only active sound source. In order to avoid this issue, we use the WebRTC voice activity detector (VAD) to determine in which frames the speech source is active. We first tried to include the VAD information as an additional input channel to the network. However, during the training, the VAD sometimes failed and classified frames with speech information as silent so the network learned that even frames classified as silent could contain useful tracking information as long as they contained a directional source and, therefore, it ignored the VAD input. In order to prevent the network from tracking directional noise sources, we finally opted to turn to zero the maps corresponding to frames classified as silent by the VAD so no directional information is seen by the network when no speech source is active.

#### 4.1.2 Model architecture

The first layer of our model is a 3D convolutional layer with 32 kernels of size  $5 \times 5 \times 5$  and parametric rectified linear unit (PReLU) activations [164]. For the temporal axis, we always use causal convolutions, so this model could be used in real-time applications generating a new DOA estimation for each new power map available without introducing any delay.

Pooling layers are typically used in CNNs to progressively reduce the size of the input and make the model generalize; not using them, means that the fully connected layers used at the end of most of the convolutional models would have a huge number of trainable parameters which would surely overfit. When the desired output of the CNN is a summary of the information of the whole input, e.g. in image classification tasks, increasing the number of channels with convolutional layers and reducing their size with pooling layers progressively reduces the spatial information and gets higher-level representations of the input. However, since our desired output is not only related to the presence of some patterns but especially to their position, we must be careful when using them.

In order to get the benefits of pooling layers but allowing the spatial information to reach the last layers of the model, we opted to, as shown in Fig. 4.1, split the model into two branches and apply max pooling in a different dimension in each one.



4.1 Robust sound source localization using SRP-PHAT and 3D convolutional neural networks

Figure 4.1: Model architecture. The noted sizes correspond to a model for 16x32 maps and an input sequence of length 103. For the sake of simplicity, we represented it with only 1 input channel, although it actually has 3.

Working this way, the branch which pools the  $\varphi$  axis can retain positional information about the  $\theta$  coordinate of the maps and vice versa. Specifically, each branch has 4 layers with a convolution with 32 kernels of size  $5 \times 3 \times 3$ , PReLU activations, and a max-pooling with a kernel size of  $1 \times 1 \times 2$  and  $1 \times 2 \times 1$  respectively. If the input power maps have less than 16 points in the  $\theta$  or the  $\varphi$  axes, it would not be possible to perform so many pooling layers; in those cases, we reduce the 4 layers to the maximum number possible:  $\log_2(\min(N_{\theta}, N_{\phi}))$ . Due to the use of 3D convolutional layers and these perpendicular branches, we named this model Cross3D.

After the 3D convolutional layers, we concatenate the results of each branch and reshape them so we have a temporal sequence of length T for each one of the elements of each channel and spherical coordinates. Each one of these temporal sequences is used as the input channels of a 1D causal convolutional layer with 128 kernels of length 5 and PReLU activations. Finally, the resultant 128 time sequences are passed through another 1D causal convolutional layer with only 3 kernels of length 5 and tanh activations. These layers are similar to the fully connected layers that most of the CNN architectures have, but we include a temporal convolution so they can still

exploit the tracking information. We use a dilation factor of 2 in order to allow the tracking to take into account a longer context without increasing the complexity of the network. With all the temporal convolutions included in the model, each DOA estimation is computed from the last 37 SRP-PHAT maps, i.e. the temporal receptive field and the tracking memory is 7.17 s.

The result of this process are 3 time sequences of length T whose elements are in the range (-1, 1), which are considered to be the XYZ coordinates of a unitary vector pointing in the direction of the source in each time frame.

Tables detailing the network architecture of Cross3D for several SRP-PHAT map resolutions can be found in appendix A.

#### 4.1.3 Training

We trained our model to minimize the Euclidean distance between the output of the network and the 3 time sequences obtained from the coordinates of the unitary vectors steering at the direction where the sound source was simulated in each time window. Similarly to the results reported in [14, 53, 54], we obtained better results using this approach than trying to directly obtain the spherical coordinates from the network even when using the great-circle distance between the output and the ground-truth DOA angles as cost function.

Although using an infinite-size dataset the term "epoch" does not have the same meaning as in most of the machine learning systems, we define an epoch as 585 trajectories (the number of book chapters in the LibriSpeech train-clean-100 subset). We employed 80 epochs with trajectories of 20 s, i.e. 103 SRP-PHAT maps, to train the model with the Adam algorithm [165] using Pytorch [166].

As explained in chapter 3, the training dataset contains scenes with reverberation times and signal-to-noise ratios (SNRs) uniformly distributed from 0.2 s to 1.3 s and from 5 dB to 30 dB respectively. However, we found that the training converged faster with higher SNRs. Therefore, we followed a curriculum learning strategy [134] using batches of 5 trajectories with SNR=30 dB for the first 20 epochs and for the following epochs we employed the full range of SNRs, increased the batch size to 10 trajectories, and reduced the learning rate from  $1 \times 10^{-4}$  to  $1 \times 10^{-5}$ .

#### 4.1.4 Evaluation

#### 4.1.4.1 Baseline methods

In order to analyze the convenience of using SRP-PHAT maps as input features of CNNs for DOA estimation, we developed some alternative CNNs to use them as baselines. We designed them to be as similar as possible to our proposed model and to have the same temporal perceptive field so they have the same tracking information.

Since we are including the position of the maximum of each map into the input of the network, we should verify if our model is actually exploiting the additional information that is within the SRP-PHAT maps or if it is only using the position of its maximums. To do that, we designed a 1D CNN which takes as input 2 time sequences with the coordinates of the maximum of each map normalized to the range [0,1] and applies to them 7 layers of 1D causal convolutions with PReLU activations and without any pooling. All the layers had a kernel size of 5 and the last two layers used a dilation factor of 2, so its temporal receptive field is 37 frames as in Cross3D. The number of channels of each layer was  $\{1024, 512, 512, 512, 128, 3\}$ . The results shown in the following sections were obtained by training this network with the same process described in section 4.1.3 and using the coordinates of maps with resolution  $64 \times 128$ .

One of the most common input features employed by the first DOA estimation techniques based on neural networks were the generalized cross-correlations (GCCs). They typically employed fully connected perceptrons with not too many hidden layers and, since they only used the GCCs computed in a temporal window, did not perform any kind of tracking. Following this idea, but with the aim of including tracking information into the network, we used the same 1D causal CNN that we used over the map maximums but using as input sequences the temporal evolution of each element of the GCCs which represented an inter-microphone delay lower than the maximum inter-microphone distance divided by the speed of sound.

Although, as explained in chapter 1, the use of 2D CNNs over spectrograms may not be optimal, we also implemented a model following this approach since it is quite popular in the literature. For a fair comparison, we used causal convolutions with a

Model	Input		Trainable parameters	Temporal perceptive field	Window length	Causal
Cross3D	Power maps	$\begin{array}{c} 4\times8\\ 8\times16\\ 16\times32\\ 32\times64\\ 64\times128 \end{array}$	$526\ 372\\946\ 340\\1\ 693\ 988\\5\ 626\ 148\\21\ 354\ 788$	$\begin{array}{c} 5.63\mathrm{s}\\ 6.40\mathrm{s}\\ 7.17\mathrm{s}\end{array}$	4096	Yes
1D CNN	GCCs Maximums (64x128)		$\frac{11282436}{6899716}$	$7.17\mathrm{s}$	4096	Yes
2D CNN SELDnet [14]	Spectrograms		$\frac{1882372}{104643}$	$7.17 \mathrm{s}$ $\infty$	$4096 \\ 512$	Yes No

Table 4.1: Models employed for the evaluation.

similar architecture to Cross3D: one convolution with  $256.5 \times 5$  kernels, four convolutions with  $256.5 \times 5$  kernels with  $1 \times 4$  pooling, a reshape to transform the remaining features into temporal sequences and two 1D causal convolutions with kernel size 5 and dilation factor 2, the first one with 128 channels and the last one with 3. For computing the spectrogram, we used the same windows as for computing the SRP-PHAT maps, extracted the magnitude and phase of each frequency of the Fourier transform (FT), and finally normalized the magnitude of each window to its maximum and the phase to the range [-1,1].

Finally, we also trained with our simulation procedure a replica of SELDnet [14] but without including the sound event detection (SED) output and with only a DOA output since we were only interested in tracking one source. This model takes as inputs the magnitude and phase of the spectrograms and has three 2D convolutional layers followed by two bidirectional gated recurrent units (GRUs) [167] and two fully connected layers. It is worth saying that this model, due to the bidirectional GRUs, is non-causal and that it uses shorter analysis windows than the other analyzed methods.

For the models that use spectrograms as input features, we found that they did not train properly with the full range of reverberations described in section 3.1, and we got the best results training them with values of  $T_{60}$  randomly selected from the range 0 s to 0.3 s.

All the models employed for the evaluation are summarized in Table 4.1 and tables detailing their architectures can be found in appendix A.

#### 4.1.4.2 Simulated dataset

We trained different models for several power map resolutions with the whole range of reverberation times and SNRs, and then we tested their performance for several specific values of  $T_{60}$  and SNR in order to analyze the robustness of the proposed tracking system.

Since we are using SRP-PHAT power maps as the input of our algorithm, we started our evaluation by comparing our model with the classic SRP-PHAT algorithm. SRP-PHAT does not perform any kind of tracking, so, for a fairer comparison, we did not take into account the silent frames when computing the root mean squared angular errors (RMSAEs) shown in Fig. 4.2. As we can see in this figure, when working with high-resolution power maps in almost anechoic rooms with high SNR, using our 3D CNN over the SRP-PHAT maps does not improve the results compared to just taking the maximum of each map; actually, our system seems to slightly degrade the DOA estimation, probably due to the effect of applying an unneeded tracking. However, when the room conditions deteriorate, we can see how Cross3D is robust enough to get its performance degraded in only 5° when the  $T_{60}$  increases to 1.5 s (which is higher than any reverberation seen during the training) while the SRP-PHAT algorithm is just unable to perform a proper estimation. They are also surprising the results obtained with maps of only 4x8 resolution, which only perform a SRP-PHAT measurement each  $45^{\circ}$  in the azimuth and  $60^{\circ}$  in the elevation and, since  $P(\theta, \varphi) = P(\theta, 0) \ \forall \varphi \in [0, 2\pi)$ if  $\theta = 0$  or  $\pi$ , only needs to perform 18 computations of (2.3).

Fig. 4.3 shows a couple of examples of simulated trajectories and their estimated DOA. In Fig. 4.3a we can see how, for scenarios with high reverberation and low SNRs, the maximum of the SRP-PHAT maps becomes really noisy but our proposed system is able to maintain the estimated DOA quite close to the actual one. In linear systems, robust tracking with noisy estimations usually comes with the cost of being slow to track fast changes, at least with casual systems, but we can see how our model was able to follow the sudden change in the azimuth of the source at the fifth second of the trajectory. In Fig. 4.3b we can see how, when working with low-resolution power maps, our system can predict the DOA with much higher precision than the



**Figure 4.2:** Localization root mean squared angular error (RMSAE) for several power map resolutions, SNR and reverberation times. The silent frames were not included in the computation of the RMSAE



Figure 4.3: Examples of the DOA estimated in a scenario with  $T_{60}=0.9$  s and SNR=5 dB using maps with  $32 \times 64$  resolution (a) and in a scenario with  $T_{60}=0.3$  s and SNR=30 dB using maps with  $4 \times 8$  resolution (b). The solid line represents the actual DOA of the source, the dashed line the estimated DOA and the crosses represent the maximum of each SRP-PHAT power map. Grey segments indicate silent frames.

maximums of the maps. This could not be done with a two-step DOA estimation and tracking algorithm that performed the tracking based only on the maximum of the maps. Our system is able to analyze the whole maps and it was able to learn to exploit the patterns in the SRP-PHAT maps to achieve higher resolution than the grid used to compute the maps.

Finally, we also tested the baseline methods under different reverberations and SNRs to compare the robustness of each model. In this case, since all the methods include tracking capabilities, we did not exclude the silent frames when we computed the RMSAEs shown in Fig. 4.4. We can see how the best results are obtained using our method with high-resolution power maps but, even reducing the resolution, its performance is still competitive. Using 1D CNN over the coordinates of the maximums of  $64 \times 128$  SRP-PHATs maps performs worse than using our 3D CNN over  $4 \times 8$  maps, so we can conclude that our model is exploiting the patterns present in the SRP-PHAT maps and not only using the information of the position of its maximums (this was also suggested by Fig. 4.3b). Using a 1D CNN over the GCCs —which is an approach, to the best of the authors' knowledge, unpublished— have a performance between using



**Figure 4.4:** Tracking root mean squared angular error (RMSAE) of Cross3D with several power map resolutions and the baseline methods for SNR=30 dB (a) and SNR=5 dB (b) and several reverberation times. The silent frames were also included in the computation of the RMSAE.

3D CNNs over  $4 \times 8$  and  $8 \times 16$  maps and may be an interesting approach when a lower computational cost is needed. Finally, the models that use spectrograms as inputs perform well in favorable scenarios (SELDnet even outperforms our proposal in lownoise anechoic chambers) but they are not very robust against noise and reverberation.

#### 4.1.4.3 LOCATA dataset

In order to confirm that, although it was trained with a simulated dataset, our system is general enough to track sound sources recorded in real rooms, we tested it with the LOCATA challenge dataset [137], which contains several recordings with the same array that we had simulated to train the models. We used the development dataset and we focused on tasks 1, 3, and 5 of the challenge: a static loudspeaker recorded with a static array, a moving talker recorded with a static array, and a moving talker recorded with a moving array; it is worth mentioning that the array was static in all the simulations employed to train the model. For the robot head microphone array that we simulated in the training dataset, the development dataset contains 3 recordings for each task and its ground-truth positions.

	Model:		Cross3D			11	1D CNN		SELDnet	
Input:		4x8	SRF 8x16	P-PHAT 16x32	' maps 32x64	64x128	GCCs Maximums		Spectrograms	
Task 1	Recording 1 Recording 2 Recording 3	$17.93 \\ 18.90 \\ 10.35$	$11.92 \\ 7.68 \\ 6.34$	$8.30 \\ 6.68 \\ 2.98$	$4.62 \\ 4.90 \\ 3.25$	$5.16 \\ 3.91 \\ 2.24$	$16.18 \\ 12.60 \\ 11.57$	$7.54 \\ 5.19 \\ 5.09$	$93.76 \\ 64.18 \\ 140.21$	$29.70 \\ 38.44 \\ 54.81$
	Average	15.72	8.65	5.99	4.26	3.77	13.45	5.94	99.38	40.98
Task 3	Recording 1 Recording 2 Recording 3	23.06 20.97 21.05	$18.11 \\ 13.71 \\ 12.74$	$13.79 \\ 10.01 \\ 9.83$	$12.43 \\ 8.36 \\ 7.69$	9.92 9.22 6.60	$\begin{array}{c} 13.59 \\ 14.17 \\ 15.21 \end{array}$	$14.04 \\ 12.02 \\ 13.29$	70.86 83.42 82.48	50.57 48.71 57.29
	Average	21.69	14.85	11.21	9.49	8.58	14.32	13.12	78.92	52.86
Task 5	Recording 1 Recording 2 Recording 3	$     \begin{array}{r}       11.93 \\       20.92 \\       23.57     \end{array} $	10.83 16.16 18.25	7.25 16.08 13.58	5.74 12.18 15.64	5.49 13.59 15.49	$     \begin{array}{r}       10.93 \\       17.33 \\       20.14     \end{array} $	$     10.53 \\     17.42 \\     23.58 $	58.33 41.98 66.91	37.24 73.17 66.50
	Average	18.81	15.08	12.31	11.19	11.52	16.13	17.18	55.74	58.97
Average		18.74	12.86	9.83	8.31	7.96	14.64	12.08	78.01	50.94

**Table 4.2:** RMSAE  $[\circ]$  of the DOA estimated for the LOCATA dataset with Cross3D using several map resolutions and the baseline tracking methods. The silent frames were included in the computation of the RMSAE.

The only modification to the proposed technique that we made after seeing its performance with the LOCATA dataset was the use of a VAD. All the hyperparameters of the model and the acoustical properties of the training dataset were selected according only to the results obtained with simulated datasets. In other words, we used simulated signals for training and validation and the LOCATA recordings only for testing.

Table 4.2 shows the RMSAE of estimating the DOA of the source of each recording using our technique and with the baseline methods. Although it is difficult to draw conclusions from such a low number of recordings, we can see how the proposed tracking system clearly outperforms the baseline methods that use spectrograms as inputs and that it also outperforms the 1D CNN methods when we use maps with at least  $16 \times 32$  resolution.

According to [137], the reverberation time of the room where the recordings were performed was  $T_{60} \approx 0.5$  s, so we can observe some degradation in the performance of Cross3D when it is used over low-resolution power maps compared with the results obtained with the simulated dataset (see Fig. 4.4), but it disappears when the res-



Figure 4.5: DOA estimated for the second recording of the second task of the LOCATA challenge using maps with Cross3D over  $32 \times 64$  maps and the baseline methods.

olution of the maps increases; actually we even reach lower errors in the LOCATA dataset than with the simulated test dataset. Using a 1D CNN also suffers a similar degradation, but its most dramatic impact is on the methods which use spectrograms as inputs. In contrast, the use of a 1D CNN over the coordinates of the maximums of high-resolution SRP-PHAT maps does not suffer almost any degradation; but it may not be an interesting approach since, having computed the  $64 \times 128$  resolution maps, we can obtain far better results using the whole maps as inputs of Cross3D.

As an example, Fig. 4.5 shows the DOA estimation of the second recording of the third task of the LOCATA dataset, where all the methods obtained an RMSAE quite close to their average. We can see how Cross3D performs the best estimation of the analyzed methods both for the elevation and for the azimuth. We can also see that the LOCATA dataset has longer silences than the ones present in the simulated dataset, which could also explain why some of the methods obtained lower results with this dataset. In order to make the methods based on CNNs more robust against longer silences, we should include them in the simulation of the training dataset and, probably, increase the temporal receptive field of the models, which could be done by increasing the number of layers, the temporal size of its kernels, or including longer temporal dilations in the convolutions.

## 4.2 A sound source localization model equivariant to the rotations of the source and the array

In the previous section, we have presented a new model based on 3D CNNs applied over SRP-PHAT power maps that has been proved to be more robust than other state-of-the-art models for SSL. However, this model is not really equivariant to the rotations of the source or the array, since the projection of the spherical SRP-PHAT maps into the 2D equiangular grid employed as input of the convolutions generates position-depending deformations that are not intrinsically compensated by the 3D convolutions (though the model could learn to do it during the training). In addition, this equiangular sampling is not the optimal way to sample the SRP-PHAT maps since it oversamples the regions close to the poles.

Many new kinds of CNNs have been proposed in recent years to obtain equivariance to spherical rotations. In [77] and in [168], it was proposed to perform the convolutions in the spherical harmonic domain and then transform their result back to the spatial domain to pointwise apply the nonlinear activations. Since then, several modifications of this approach have been published, some of them proposing nonlinear activations in the harmonic domain [169, 170]. Another approach is to analyze the spherical signal as a graph where each point is connected to its neighbors [171, 172, 173]. By working this way, they avoid the need to work in the harmonic domain, but, in most cases, their kernels are restricted to being isotropic, i.e. to having circular symmetry.

In this section, we propose the use of a third approach: the icosahedral CNNs presented in [76]. These networks are only strictly equivariant to the 60 rotational symmetries of the icosahedron instead of the continuous space of spherical rotations, but they have a much more efficient implementation based on standard 2D convolutions. They have been proven to smoothly generalize to the continuous space of spherical rotations when these rotations are shown during the training of the model and their hexagonal kernels are not restricted to be isotropic thanks to saving the results of every possible orientation as separate channels.

In addition, in order to preserve the equivariance of the whole model, we present a new layer, which we call soft-argmax, to replace the fully connected layers that are usually employed at the end of the convolutional models. This layer interprets the output of the last convolutional layer as the probability distribution of the source position and computes its expected value, so allows us to convert a classification output into a regression output in a differentiable and interpretable way without adding any trainable parameters to the model.

#### 4.2.1 Icosahedral CNNs

Several techniques have been recently proposed to extend the translation equivariance of conventional CNNs to spherical rotations, most of them based on the spherical harmonics domain. Although they are only equivariant to the 60 icosahedral rotations instead of the continuous space of spherical rotations SO(3), in this work we use the icosahedral CNNs proposed in [76] due to its efficient implementation based on conventional bi-dimensional CNNs. The icosahedron is the platonic solid with the highest number of faces and it allows us to approximate a sphere with lower error than other geometric shapes with a lower number of faces while being able to define a convolution operation on a hexagonal grid without needing any kind of interpolation and implement it using a conventional 2D convolution. The details of this implementation and some experiments proving its good performance approximating spherical signals (and its rotations) can be found in [76], but we summarize some of the main ideas in this section.

The icosahedral grid used to sample the spherical signals is built recursively starting from the vertices of the icosahedron (which we define as having a vertex in the south and another in the north pole as shown in Fig. 4.7a) and then subdividing each triangular face into 4 smaller triangles by introducing a new point in the center of each edge as shown in Fig. 4.6. Repeating this process r times, we obtain a grid with  $5 \cdot 2^{2r+1} + 2$  points. As we can see in Figs. 4.7a and 4.7c, apart from the vertices of the icosahedron, which only have 5, every one of these inner sampling points has 6 neighbors, so we can see them as hexagonal pixels. In [76], due to their pentagonal shape, it is proposed to keep the corners values to 0 to preserve the equivariance of the model but, in order to avoid artifacts around them after applying the convolutions,



Figure 4.6: Sampling points of a face of the icosahedron for different resolutions.

we replace that 0 with the average value of their 5 neighbors (which also preserves the equivariance of the model).

Due to the hexagonal shape of the pixels defined this way, the icosahedral CNNs use hexagonal kernels which can be stored in  $3 \times 3$  bi-dimensional kernels. If we want the model to be equivariant to the icosahedral rotations without restricting the kernels to be isotropic, we have to consider their 6 possible rotations so, for each kernel in the convolution, we have to work with 6 channels instead of with just one as it is done in conventional convolutions. Finally, in order to implement the icosahedral convolutions using standard 2D convolutions, a projection of the whole icosahedral grid into a  $5 \cdot 2^r \times 2^{r+1}$  rectangular grid (Fig. 4.7b) is presented in [76] along with a way to circularly pad it into a  $5 \cdot (2^r + 2) \times 2^{r+1} + 2$  extended grid to preserve the equivariance of the model.

To sum up, we can apply a convolution with C kernels over an icosahedral grid of resolution r using a conventional 2D convolutional layer with  $3 \times 3$  kernels and 6Cchannels over a  $5 \cdot (2^r + 2) \times 2^{r+1} + 2$  image.

In [76], it is not detailed how the pooling layers are implemented. In our implementation, an icosahedral pooling layer reduces an icosahedral grid of resolution r to a new one of resolution r-1 where each new hexagonal pixel is computed as the average of the pixel which has its same center and its 6 neighbors. This can be seen as the icosahedral equivalent of a 2D pooling with kernel size  $3 \times 3$  and stride  $2 \times 2$ .



Figure 4.7: Example of an icosahedral SRP-PHAT power map with resolution r = 3 in a high reverberation low noise scenario:  $T_{60}=1.0$  s and SNR=30 dB (a), the 2D representation employed for the convolution implementation (b) and a spherical projection for visualization purposes (c). The green sphere/circle indicates the actual DOA of the sound source.

#### 4.2.2 Soft-argmax regression

In the model presented in the previous section using 3D convolutions, in order to perform the regression of the source coordinates after the convolutional layers, we flattened the activation maps of the last layer and fed it to several fully connected layers (actually, we used 1D convolutions in order to allow the model to also take into account the previous frames). Even when employing pooling layers along with the convolutional layers to reduce the number of activations that reach the fully connected layers, this approach has a high computational cost and highly increases the number of trainable parameters of the model, which increases its memory consumption and its risk of overfitting.

For our model based on icosahedral convolutions, we replace those fully connected layers with a new soft-argmax function, where we use a soft-max layer to ensure that the sum of the whole activation map is 1.0 and then we just sum the results of multiplying each hexagonal pixel by the coordinates that they represent in the icosahedral grid:

soft-max 
$$(P(\boldsymbol{x})) = \frac{e^{P(\boldsymbol{x})}}{\sum_{\boldsymbol{x}\in\mathcal{X}}e^{P(\boldsymbol{x})}} = \frac{e^{P(\boldsymbol{x})-\max(P(\boldsymbol{x}))}}{\sum_{\boldsymbol{x}\in\mathcal{X}}e^{P(\boldsymbol{x})-\max(P(\boldsymbol{x}))}}$$
 (4.1)

soft-argmax 
$$(P(\boldsymbol{x})) = \sum_{\boldsymbol{x} \in \mathcal{X}} \boldsymbol{x}$$
 soft-max  $(P(\boldsymbol{x}))$ , (4.2)

where  $P(\mathbf{x})$  is the output of the last convolutional layer of the model and  $\mathbf{x} \in \mathcal{X}$ are the coordinates of the points of the icosahedral grid  $\mathcal{X}$  where it is sampled. The subtractions of max  $(P(\mathbf{x}))$  inside the exponential functions are done for numerical stability reasons without affecting the analytical result.

This way, the output of the icosahedral convolutions, after being normalized with the soft-max function, can be seen as the probability distribution of the coordinates of the source and the output of the soft-argmax function as its expected value. Another advantage of this approach, apart from the model interpretability and the reduction in the computational and memory costs, is that we avoid introducing any non-equivariant layer to the model. Although we could directly estimate the spherical coordinates of the source defining  $\boldsymbol{x}$  in spherical coordinates, we estimate the 3D coordinates of the unitary vector pointing at the direction of the source by defining  $\boldsymbol{x}$  in 3D Cartesian coordinates. As previously explained, it has been proven that this brings better results than directly inferring the elevation and azimuth angles [53, 54] and, in addition, continuing with the interpretation of the output of the CNN as a probability distribution function, minimizing the mean squared error (MSE) of this vector does not only imply reducing the distance between its expected value and the source position but also reduces its variance, since a completely unitary vector would only be possible if only one pixel is activated. Therefore, we can see the norm of the output vector as a measurement of the confidence in the DOA estimation, being closer to 1 when the confidence is higher as done with the activity-coupled Cartesian direction of arrival (ACCDOA) representation [58].

In Fig. 4.8 we can see an example of the output of the last convolutional layer of the trained model after being normalized with the soft-max function and how it represents a probability distribution whose expected value accurately estimates the actual DOA of the sound. It is worth noting how the precision of the estimated DOA exceeds the grid used to sample the probability distribution, overcoming the main issues of the classification approaches to SSL.

#### 4.2.3 Proposed technique

#### 4.2.3.1 Model architecture

Since the regression of the estimated DOA is performed with our soft-argmax layer, which already integrates the coordinates of every map point, the input of the model has only one channel with the SRP-PHAT map and it does not include the additional channels that we used in section 4.1 to indicating the coordinates of the maximum of every map. Including these channels would have broken the rotational equivariance of the model and would have made it more difficult its extension to multi-source scenarios.

As shown in Fig. 4.9, we combine icosahedral convolutions with one-dimensional convolutions operating in the time dimension in order to take the temporal context



Figure 4.8: The output of the last convolutional layer after passing through the softmax function, corresponding to the input maps shown in Fig 4.7, and the result of the soft-argmax function converted into spherical coordinates (red cross). The green circle indicates the actual DOA of the source.

into account when performing the DOA estimation while being equivariant to both icosahedral rotations and temporal translations. Each convolutional unit is composed of an icosahedral and a temporal convolution followed by layer normalization [174] and rectified linear unit (ReLU) activation. The temporal convolution has a kernel of size 5 operating causally, i.e. its receptive field only includes past maps, and both the icosahedral and the temporal convolution have 32 kernels. To preserve the equivariance of the model, the 6 kernel-orientation channels of every icosahedral kernel are seen as 6 independent signals by the temporal convolutions. The impact of the number of convolutional kernels in the performance of the model is analyzed in section 4.2.4.2.

In [174], it was hypothesized that layer normalization did not provide relevant improvements in convolutional neural networks since the hidden units close to the boundaries of the images did not follow the same distribution as the rest of the hidden units. However, our inputs do not have boundaries and we found that, adding layer normalization to our model, it converged faster and more robustly during the training. To keep the model equivariant to the icosahedral rotations, we have implemented a layer normalization that normalizes the inputs along the 32 channels and its 6 kernel orientations but the scale weights are tied for the 6 kernel orientations in the affine transformation.



**Figure 4.9:** Architecture of the proposed model. *B* is the batch size, *T* is the number of temporal frames of the trajectory,  $H = 2^r$  and  $W = 2^{r+1}$  are the height and the width of the projections of the icosahedral grid, and N = r - 1 is the number of down-sampling units used for that input resolution.

#### 4.2 A sound source localization model equivariant to the rotations of the source and the array

We concatenate two of these convolutional units to an icosahedral pooling to build a down-sampling unit and stack as many of these down-sampling units as needed to get an r = 1 icosahedral map (in the case of using maps with r = 1 as input we do not use any down-sampling unit). When we have a minimal-size icosahedral map, we concatenate 5 convolutional units (the last one with only an output channel for the 1D convolution and without layer normalization and the ReLU activation). This way, we ensure that the receptive field of all the output cells of the icosahedral CNN includes all the cells of the input map independently of its resolution. Finally, we use a maxpooling layer over the 6 kernel-orientation channels and feed the resulting icosahedral maps to the soft-argmax layer explained in section 4.2.2.

#### 4.2.3.2 Training

For the training dataset, we used again the technique described in chapter 3 to generate random source trajectories and simulated them using our GPU implementation of the Image Source Method [61] at a sample rate of 16 kHz using utterances from the LibriSpeech train-clean-100 dataset [135] as source signals. We simulated the same 12-microphones array included in the LOCATA dataset [137, 138] as we did for Cross3D, which has a minimum and maximum inter-microphone distances of 1.3 cm and 12.1 cm respectively.

As done for the Cross3D models, we computed the SRP-PHAT maps using frames of length K = 4096 samples (i.e. 256 ms) with a hop size of 3K/4. We normalized the maps by subtracting their mean and dividing them between their maximum and used the Voice Activity Detector of the Web Real-Time Communication (WebRTC) project [136] to turn to 0 the maps corresponding to silent frames.

As explained in the previous section, using this approach we have an infinite-size dataset, but we define an epoch as 585 random trajectories of 20 s, each one with an utterance randomly taken from one of the 585 chapters present in the LibriSpeech train-clean-100 subset. We used Pytorch [166] to train the model using the Adam algorithm [165] over 50 epochs. Similar to the curriculum learning [134] strategy employed to train the Cross3D models, we keep fixed the SNR of the simulations to

Model	Input		SRP-PHAT computations	Trainable parameters	Temporal receptive field
IcoCNN	Icosahedral SRP-PHAT maps	$r=1 \\ r=2 \\ r=3 \\ r=4$	$30 \\ 150 \\ 630 \\ 2550$	$193441\\289953\\386465\\482977$	4.10 s 5.63 s 7.17 s 8.70 s
Cross3D	Equiangular SRP-PHAT maps	$\begin{array}{c} 4\times8\\ 8\times16\\ 16\times32\\ 32\times64\\ 64\times128 \end{array}$	18 98 450 1922 7938	$526\ 372\\946\ 340\\1\ 693\ 988\\5\ 626\ 148\\21\ 354\ 788$	$\begin{array}{c} 5.63\mathrm{s}\\ 6.40\mathrm{s}\\ 7.17\mathrm{s}\end{array}$
1D CNN	GCC	Ċs	0	11282436	$7.17\mathrm{s}$

 Table 4.3:
 Models employed for the evaluation.

30 dB during the first 25 epochs and then we employed uniformly distributed random values from 5 dB to 30 dB in the following epochs.

#### 4.2.4 Evaluation

#### 4.2.4.1 Baseline techniques

Since, in the previous section, they were the techniques that proved to be the most robust against reverberation, we compared the proposed technique with Cross3D and the 1D CNN operating over the GCC coefficients.

As we can see in Table 4.3, the proposed model has a much lower number of trainable parameters than the other models. This is because the final regression is performed with the soft-argmax function (without trainable parameters) instead of with fully connected layers.

#### 4.2.4.2 Simulated dataset

In order to analyze the performance of the proposed model under different acoustic conditions, we first evaluated our model using synthetic signals simulated following the same procedure employed for the training dataset. We found that some utterances included a short period of silence at the beginning which artificially biased the results, so we have not taken into account the localization error of the first 5 frames, i.e. the



Figure 4.10: Localization root mean squared angular error (RMSAE) under several simulated conditions for the proposed and the baseline techniques.

first second, of each trajectory even if they were used to train the models. All the root mean localization errors described in this section include the frames where the source was silent, but the models are able to continue estimating its position thanks to the temporal context included in their receptive fields through the temporal convolutions.

As we can see in Fig. 4.10, the proposed models clearly outperform the baselines even using maps of lower resolution and having far less trainable parameters. To make this even clearer, Fig. 4.11 plots the localization error represented as a function of the number of computations of the SRP-PHAT functional (2.3), needed to compute the input maps of each model. It is worth noting that the reverberation times  $T_{60} = 0.0$  s and  $T_{60} = 1.5$  s are out of the range of reverberation times used during the training, but the model generalizes well to them continuing with the same tendency shown with the rest of reverberation times.

We can also see how using icosahedral maps with a resolution higher than r = 2 does not seem to improve the accuracy of the DOA estimations. Considering that those models had a higher number of down-sampling units, and therefore more trainable parameters and longer temporal receptive fields, we can conclude that the maps with r = 2 already contain all the information useful for tracking. This limit is probably determined by the size of the array we are using to compute the maps and that limits their spatial bandwidth; using bigger arrays would probably allow us to obtain even better results from higher-resolution maps.

In Fig. 4.12, we can see an example of a random trajectory in a scenario with



Figure 4.11: Localization root mean squared angular error (RMSAE) vs the number of computations of equation (2.3) needed to compute each input map. The semitransparent area indicates the whole reverberation interval from  $T_{60} = 0.2$  s to 1.3 s and the solid line indicates its average value.

high reverberation time and low noise using maps with resolution r = 2. We can see that the maximums of the SRP-PHAT maps are in spurious positions in many frames and, even in those where they are in the grid position closest to the ground truth, they are quite far due to their low resolution. However, we can see that the estimation of the proposed model stays always closer to the actual DOA of the source even during the silent frames since the icosahedral convolutions allow the model to analyze the whole maps instead of just taking only the information of the position of their absolute maximums and thanks to the temporal context provided by the temporal convolutions.

In Fig. 4.8 we have seen the r = 1 probability distribution inferred by the model for the r = 3 maps depicted in Fig. 4.7. We can see how the model is able to accurately adjust the probability assigned to every hexagonal pixel around the pentagonal vertex so the result of the soft-argmax function is precisely displaced from the center of the closest hexagonal pixel to the actual DOA of the simulated source. It is worth saying that, even if the vertex values were replaced by the average value of their neighbors during the convolutional layers, they are turned to 0 before the soft-argmax function.

Since that was the only difference that we had introduced in our implementation of the icosahedral convolutions with respect to the original implementation described in [76], we also studied the effect of replacing the vertices of the icosahedron by zeros



Figure 4.12: DOA estimation for a random trajectory simulated with  $T_{60}=1.0$  s and SNR=30 dB. The solid lines represent the ground-truth DOA and the dashed lines the estimation performed by the proposed method using input maps of resolution r = 2. The dots indicate the position of the maximum of those maps.

**Table 4.4:** Mean RMSE  $[\circ]$  in the simulated dataset with the different strategies to handle the vertices in the icosahedral convolutions.

Vertices values	Map resolution					
	r=1	r=2	r=3			
zeros	11.02	7.81	7.31			
neighbors' average	10.89	7.81	7.57			

instead of by the average of their neighbors. As we can see in Table 4.4, the difference between both strategies is negligible but, considering that the impact of the averaging in the computational cost of the model is also negligible, we still think that our strategy is preferable since it reduces the artifacts around the vertices in the hidden layers of the model.

In Table 4.5, we can see the mean RMSAE for all the reverberation times and SNRs of the evaluation dataset for several numbers of convolutional kernels. It should be noted that, if we need to reduce the computational complexity and memory consumption of the model, we can reduce the number of convolutional kernels without having a too large impact on the localization accuracy. This is possible thanks to the equivariances of the model since, contrary to conventional 2D CNNs, the same kernel

Convolutional	Map resolution				
kernels	r=1	r=2	r=3		
4	14.16	12.78	10.19		
8	12.63	9.82	8.84		
16	11.89	8.61	7.69		
32	10.89	7.81	7.57		
64	10.21	7.38	6.71		

Table 4.5: Mean RMSAE [°] in the simulated dataset with different numbers of convolutional kernels.

can model the different rotations of its pattern. This fact, combined with avoiding the distortions that the 2D projection of the maps generated in Cross3D, allows the proposed model to extract even more information than the previous model using less convolutional kernels.

#### 4.2.4.3 LOCATA dataset

To confirm that the models trained with simulated signals are general enough to work with signals recorded in real environments, we have tested them using the recordings of tasks 1, 3, and 5 (i.e. the tasks with only one source) of the evaluation partition of the LOCATA dataset [137]. Table 4.6 and Fig. 4.13 show the average results for every task and for the whole evaluation partition and we have included the root mean squared error (RMSE) of every recording in appendix B.

As we can see in Fig. 4.14, some of the recordings of the LOCATA dataset include long periods of silence when the sound source was moving. During these silent periods, the models obviously cannot keep tracking the movements and if they constitute a high percentage of a recording, its RMSAE will be strongly biased by them; therefore, Table 4.6 also includes the RMSAEs without taking into account these silent frames. In addition, it is worth mentioning that the size of the dataset is quite small, with the 23 recordings used for this test adding less than 7 minutes after having removed the initial silences, which makes it very sensitive to any anomalous circumstance that could appear even if it is only present for a short period of time.

We can see that the difference between the model using icosahedral convolutions

**Table 4.6:** Mean RMSAE  $[\circ]$  of the DOA estimated for every task of the evaluation partition of the LOCATA dataset with the icosahedral CNNs using several map resolutions and the baseline tracking methods. The second (gray) numbers indicate the RMSAE without taking into account the frames when the sound source was silent.

Model:	IcoCNN				Cross3D					1D CNN
Input:	Icosah r=1	r=2	SRP-PH r=3	$\operatorname{AT\ maps}_{r=4}$	Equ 4x8	uiangul 8x16	ar SRP- 16x32	PHAT 32x64	maps 64x128	GCCs
Task 1	$\underset{\scriptstyle 8.34}{8.34}$	$5.88$ $_{6.08}$	$\underset{\scriptscriptstyle{5.78}}{5.57}$	$\underset{\scriptscriptstyle{5.23}}{5.23}$	$\underset{\scriptscriptstyle 27.09}{28.62}$	$\underset{\scriptscriptstyle{22.16}}{22.16}$	$\underset{\scriptstyle 14.32}{13.89}$	$9.84 \\ \scriptscriptstyle 8.51$	$\underset{\scriptscriptstyle{5.48}}{5.48}$	$\underset{\scriptscriptstyle{12.80}}{12.80}$
Task 3	$\underset{\scriptscriptstyle 8.97}{10.53}$	8.94 7.29	$\underset{\scriptstyle{6.93}}{10.07}$	$\underset{\scriptscriptstyle{7.87}}{9.57}$	18.51 17.98	17.56 17.50	12.76 12.16	11.18 10.33	$\underset{\scriptstyle 8.86}{9.92}$	$\underset{\scriptscriptstyle{11.62}}{12.09}$
Task 5	$\underset{\scriptscriptstyle{13.03}}{15.48}$	$\underset{\scriptscriptstyle 8.87}{11.13}$	$\underset{\scriptscriptstyle 9.66}{11.54}$	$\underset{\scriptscriptstyle 8.49}{10.95}$	$\underset{\scriptscriptstyle 15.82}{19.13}$	$\underset{\scriptscriptstyle{13.38}}{17.49}$	$\underset{\scriptscriptstyle 10.75}{13.03}$	$\underset{\scriptscriptstyle 10.63}{12.20}$	$\underset{\scriptscriptstyle 10.73}{12.49}$	$\underset{\scriptscriptstyle{13.31}}{16.47}$
Average	$\underset{\scriptstyle{9.50}}{10.20}$	$\underset{\scriptstyle 6.97}{7.69}$	$7.85$ $_{6.87}$	$7.43$ $_{6.51}$	$\underset{\scriptscriptstyle 23.47}{24.90}$	$\underset{\scriptstyle 19.24}{20.32}$	$\underset{\scriptscriptstyle 13.03}{13.03}$	9.84 9.52	7.86	$\underset{\scriptscriptstyle 12.65}{13.30}$
Median	$\underset{9.94}{9.91}$	$7.17$ $_{6.79}$	$\underset{\scriptstyle{6.53}}{6.66}$	$\underset{\scriptstyle 6.45}{6.45}$	18.51 18.25	$\underset{\scriptstyle 14.43}{15.32}$	$9.90$ $_{7.65}$	7.58	$\underset{\scriptstyle{5.74}}{5.74}$	$\underset{\scriptscriptstyle{12.00}}{12.00}$
Standard deviation	$\underset{\scriptstyle 4.17}{5.00}$	$\underset{\scriptstyle 2.40}{\textbf{3.53}}$	$\underset{\scriptscriptstyle 2.69}{3.82}$	$\underset{\scriptscriptstyle 2.26}{3.51}$	$\underset{\scriptscriptstyle 14.39}{15.65}$	$\underset{\scriptscriptstyle 13.24}{13.24}$	$\underset{\scriptscriptstyle{13.26}}{12.87}$	$\underset{8.07}{7.71}$	$\underset{\scriptscriptstyle{5.37}}{5.37}$	$5.67$ $_{5.44}$



Figure 4.13: Average localization root mean squared angular error (RMSAE) in the LOCATA dataset vs the number of computations of equation (2.3) needed to compute each input map, along the whole signal (a) and without taking into account the silent frames (b). The semitransparent area indicates the average  $\pm$  the standard deviation.



Figure 4.14: DOA estimation for the third recording of task 5 of the LOCATA dataset. The solid lines represent the ground-truth DOA and the dashed lines the estimation performed by the proposed method using input maps of resolution r = 2. The dots indicate the position of the maximum of those maps.

and Cross3D is not as great in this dataset as in the synthetic one, though the model using icosahedral convolutions still clearly outperforms the baseline model, especially when using low-resolution power maps. This could be due to the differences between the signals simulated with the image source method (ISM) method used to train the models and the signals recorded in a real room used for this test. It seems that we might be approaching the accuracy limit imposed by this dataset difference, so the accuracy with real recordings can not improve even when we improve the models. In recent years, several domain adaptation techniques have been proposed to improve the accuracy of models trained with simulated signals [46, 64, 65, 66] and it would be interesting to conduct further studies along these lines.

In any case, as can be seen in Fig. 4.13, the proposed model working with maps of resolution r = 2 still has in the LOCATA dataset an accuracy comparable with Cross3D using maps of much higher resolution, reducing in almost two orders of magnitude both the number of SRP-PHAT computations and the number of trainable parameters, which might be crucial in applications where the sound source localization must be done in low-cost devices in real time. We can also see how the proposed



Figure 4.15: Example of an input map from the 1st recording of the 5th task of the LOCATA evaluation dataset (a) and the output of the last convolutional layer after passing through the soft-max function and the result of the soft-argmax function converted into spherical coordinates (red cross) (b). The green circle indicates the actual DOA of the source.

model provides far more consistent results than the baseline (whose results have a much higher variance), which also suggests a better generalization from the simulated dataset used for training to the actual recordings of the LOCATA dataset.

Finally, Fig. 4.15 shows an example of an input SRP-PHAT map extracted from the first recording of the fifth task of the evaluation partition of the LOCATA dataset and its corresponding model output. As in Fig. 4.8, we can see how the energy of output of the CNN is distributed in a way that, after normalized with the soft-max function and interpreted as a probability distribution, its expected value approaches the actual DOA of the sound.

#### 4.3 Conclusions

In this chapter, we have presented two new models for sound source localization, both of them using fully convolutional architectures over SRP-PHAT power maps. The first one uses a 3D CNN and the second one uses an icosahedral CNN. Since they do not use any non-causal element, such as bidirectional recurrent layers, they are feasible for real-time applications and provide a new DOA estimate with every new input map; i.e., every 192 ms. The experiments performed show that the SRP-PHAT maps are a good input feature to be used in tracking systems based on deep learning, being much more robust to reverberation and noise than the use of spectrograms as proposed in most of the recent literature. They also prove that it is possible to obtain a good tracking performance using only causal convolutional layers and that non-causal recurrent layers are not needed.

In addition, our second model also shows how using architectures that fit the equivariances of the problem that we want to solve allows us to increase the accuracy of the models while reducing the number of trainable parameters. The model is completely equivariant to time shifts and to the 60 rotational symmetries of the icosahedron, which is a good approximation of the continuous space of spherical rotations. It can be implemented using conventional 2D convolutional layers and has a low number of trainable parameters thanks to replacing the fully connected layers that are typically employed after the convolutional layers with a differentiable version of the argmax function, which, in addition, allows us to interpret the output of the convolutional layers as a probability distribution whose expected value is the DOA estimation.

Finally, it is worth mentioning that, even if we have focused on using SRP-PHAT maps as inputs for our models, any other spatial pseudo-spectrum could also be used (such as those computed using the MUSIC algorithm), so further studies comparing different input features and even combining them as different input channels would also be interesting.  $\mathbf{5}$ 

# Permutation invariant multi-source tracking

In this chapter we present our works on multi-source tracking. We first introduce the tracking problem and its invariance to the permutations of the sources in section 5.1 and our attempts to develop an iterative multi-source localization system based on source cancellation in section 5.2. Then we present our main proposals for sound source tracking (SST): the permutation invariant gated recurrent units (PI-GRUs) and the sliding permutation invariant training (sPIT) in sections 5.3 and 5.4 respectively. Finally, we evaluate these proposals in section 5.5 and we end the chapter with some conclusions in section 5.6.

This chapter includes the reproduction of figures and text fragments from [VII] with permission of the copyright holders.

# 5.1 Sound source tracking (SST) and the permutation invariance

In the previous chapter, we focused on providing direction of arrival (DOA) estimates as accurate as possible in scenarios where we knew that one and only one source was always present. To train deep-learning models to do this in a supervised manner, we could easily define loss functions by comparing the estimates of the model with the ground-truth DOAs. However, when moving from single to multiple sound source localization (SSL), we need to assign every estimate to one of the ground-truth DOAs. If there are no possible criteria to classify or order the sources, we cannot expect the estimates to follow the same order as they have in the ground-truth labels and, therefore, any permutation of the estimates is equally valid. If we do not take into account this permutation invariance when defining our loss function, our models will converge to useless solutions as estimating all the sources in the middle of the actual DOAs since, not being able to match the order of the sources in the ground-truth labels, this is the feasible solution that minimizes the average localization error of the sources.

In addition, when moving from SSL to sound source tracking (SST), apart from providing DOA estimates, we must also detect when a new source appears or disappears and be able to associate every new estimation to the correct tracked trajectory avoiding identity switches (IDSs). In classical tracking systems, this is typically solved in two steps following what is usually called tracking-by-detection: they first try to detect every source present in a time frame and then they filter these estimated DOAs associating them to the tracked trajectories and determine if new sources have appeared or some of the tracked sources have disappeared. Although the deep-learning models for SST usually do not split the tracking into two completely separated steps, many of them use recurrent convolutional neural network (RCNN) where, according to the temporal receptive field of every layer and the information that they receive, we can expect the convolutional layers of the model to specialize in the detection and the recurrent layers in the tracking.

As shown in Fig. 5.1, we can see a tracking-by-detection system as a recurrent process that, at every time frame, combines the new DOA estimates provided by the detector with the last tracking output to update the tracked trajectories. Under this model, we would expect the tracking system to be invariant to the permutations of the DOA estimates generated by the detector (since they are an unordered set) and equivariant to the permutations of the tracked trajectories (since they are also an unordered set but the new estimates must be assigned to the correct trajectory).


Figure 5.1: Diagram of a tracking-by-detection system seen as a recurrent process.

**Table 5.1:** Mean RMSAE  $[\circ]$  simulating just one source and simulating two sources and analytically canceling one of them.

Number of sources	Map resolution		
	r=1	r=2	r=3
One	11.02	7.81	7.31
Two (one analytically canceled)	31.49	26.96	26.03

# 5.2 Iterative multi-source localization through source cancellation using deep learning

In order to build a tracking system following a tracking-by-detection approach, we first need to have a detector that outputs a set of DOAs for every time frame. Having developed a technique that is able to cancel a source from the generalized cross-correlation (GCC) functions of a microphone array and a deep-learning model for single-source localization, we tried to combine both elements to design an iterative SSL system with the structure shown in Fig. 5.2a.

We first tried to simply use the analytical cancellation technique presented in chapter 2 to obtain the canceled maps. However, as can be seen in figures from Fig. 2.16 to Fig. 2.20, this cancellation has a too aggressive effect on the steered response power with phase transform (SRP-PHAT) maps when used in reverberant environments, which leads to the very poor performance of the model presented in the previous chapter even when trained using canceled maps (see Table. 5.1). Therefore, we tried to use deep learning techniques to improve the cancellation technique.



Figure 5.2: Structure of iterative multi-source localization systems using source cancellation over the GCCs (a) and over the SRP-PHAT maps (b).

The cancellation technique presented in chapter 2 computes the canceled GCCs as a linear combination of time-shifted versions of the original GCCs, where the scale coefficients applied to every GCC in the linear combination are constant and the time shifts are calculated based on the array geometry and the DOA of the source we want to cancel. Our first attempt to include neural networks into this process was letting a multilayer perceptron (MLP) compute these coefficients and time shifts with the idea of being able to apply multiple iterations of the cancellation technique using different sets of coefficients and time shifts generated with a neural network for the same canceled DOA.

First, in order to confirm that an MLP was able to, at least, replicate the analytical solution, we trained an MLP to replicate it as shown in Fig. 5.3a and we found that a small MLP with just one hidden layer was indeed able to easily replicate it. After that, we implemented the cancellation technique in the frequency domain so it was easier to track gradients through it but, when we tried to train the same MLP to minimize the error of the canceled GCCs (Fig. 5.3b) or the SRP-PHAT maps (Fig.

#### 5.2 Iterative multi-source localization through source cancellation using deep learning

5.3c) generated using it, we found that the training did not converge to any useful solution. We also tried to train the MLP to obtain the GCCs and the maps obtained without including the source that we want to cancel into the acoustic simulation but, as could be expected after having seen the previous results, this did not converge either. We think this is due to the peaks of the GCCs being too narrow, especially when using the phase transform (PHAT), which led to the GCCs with the random time shifts generated by the MLP at the beginning of the training not having any overlap with the GCCs needed for the cancellation and therefore not generating any useful gradients to train the model.

After concluding that the previous approach was not feasible, we tried to use a residual U-net [71] architecture over the SRP-PHAT maps to cancel the effect of a source in order to develop an iterative system as the one represented in Fig. 5.2b. We first tried to train it to replicate the maps obtained without including the canceled source in the acoustic simulation, but then we found that we could obtain better results by training it to generate maps whose maximum, using the soft-argmax layer presented in section 4.2.2, was at the DOA of the remaining source. This led us to believe that, especially if we use 3D convolutions in the U-Net to increase the temporal receptive field of the model, finding the second source is easier for the models than canceling the first one since the maps generated by this model did not have any similarity with the original maps and looked as if the model had located the second source and created a completely new map for it.

All these results dissuaded us from keep trying this iterative approach for multisource localization and we decided to move to models that could locate several sources at the same time. Using the icosahedral convolutional neural network (icoCNN) presented in the previous chapter, this is easily done by just increasing the number of output channels of the last convolutional layer from 1 to the maximum number of sources that we want our model to be able to locate and then applying the softargmax layer independently to every one of these channels so each one can provide a different DOA for a different source.

### 5. PERMUTATION INVARIANT MULTI-SOURCE TRACKING



Figure 5.3: Experiments training an MLP to replicate the analytical coefficients and time shifts of the source cancellation technique presented in chapter 2.

# 5.3 Permutation invariant gated recurrent units (PI-GRUs)

Since tracking can be seen as a recursive process, using recurrent neural networks (RNNs) at the end of our models seems to be a natural option, and this is indeed what most of the state-of-the-art models do [14, 26, 38, 41, 42, 58, 62]. Conventional RNNs, such as the long short-term memory (LSTM) layers [175] or the gated recurrent units (GRUs) [176], use a  $\mathbf{h}(t) \in \mathbb{R}^{d_h}$  vector to store the tracking state, which is updated at every time frame based on an input vector  $\mathbf{x}(t) \in \mathbb{R}^{d_x}$  and, since these updates are usually computed using fully connected perceptrons, their computational complexity and their number of trainable parameters grow with the square of the length of  $\mathbf{h}(t)$ . When applied to multi-object tracking (MOT), the information corresponding to all the tracked objects is stored in  $\mathbf{h}(t)$  without any predefined order or structure, which makes it hard to interpret and impose a compromise between the number of tracked objects, the amount of information that is stored for every one of them, and the computation complexity and the number of trainable parameters of the model. Furthermore, in addition to this compromise, if we stack the information of every object provided by the detector into a single vector, any change in the order of these objects can dramatically change the network output, so the network needs to learn the equivariance of the problem during the training.

In order to overcome the aforementioned issues of the conventional RNN for tracking applications, we propose, following the Geometric Deep Learning philosophy, the design of a new kind of recurrent architecture that shares the permutation invariances and equivariances of the tracking problem. As analyzed in section 5.1, if a tracking system is taking as inputs the estimates of a SSL system, we can see these estimates as an unordered set and therefore we should expect the tracking output to be the same regardless the order in which they are presented; i.e., the tracking system should be invariant to the permutation of its input estimates. On the other hand, if we want the identities assigned to every trajectory to be consistent during the time, once a source is randomly assigned to one of the tracking outputs, it should always be assigned to the same output but, in the case of having being assigned to a different output,



Figure 5.4: Architecture of the proposed permutation invariant gated recurrent unit (PI-GRU).

the result should be exactly the same except for the output where that trajectory is presented; i.e., the tracking system should be equivariant to the permutation of the tracked trajectories.

In contrast to these conventional RNNs, we propose replacing the input and state vectors  $\mathbf{x}(t)$  and  $\mathbf{h}(t)$  with sets of vectors  $\mathbf{X}(t) = {\mathbf{x}_1(t), \mathbf{x}_2(t), ..., \mathbf{x}_{M_x}(t)}$  and  $\mathbf{H}(t) = {\mathbf{h}_1(t), \mathbf{h}_2(t), ..., \mathbf{h}_{M_h}(t)}$  where every element  $\mathbf{x}_i(t) \in \mathbb{R}^{d_x}$  and  $\mathbf{h}_i(t) \in \mathbb{R}^{d_h}$  contains the information corresponding to only one input detection or one tracked trajectory respectively. Since the most efficient way to computationally represent data for deeplearning applications is in form of matrices, we can represent the input and state sets as two  $M_x \times d_x$  and  $M_h \times d_h$  matrices as far as we ensure that every row of  $\mathbf{H}(t)$ is updated in a way that is invariant to the permutations of the rows of  $\mathbf{X}(t)$  and equivariant to the permutations of the rows of  $\mathbf{H}(t-1)$ . For the sake of notation simplicity, we will consider  $d_x = d_h = d$  in the remainder of this thesis, but the proposed architecture could easily be adapted to use different embedding sizes for the input and the state elements with just some minor modifications.

Interestingly, one of the most popular deep-learning modules nowadays already has these desired symmetries: the multi-head attention employed in transformer architectures [84]. The multi-head attention operation takes as input three matrices, typically called queries, keys and values, and its output, in the case of using the same matrix for the key and the value inputs, is invariant to the permutations of the rows of the key and value matrix and equivariant to the permutations of the rows of the query matrix. In order to combine these properties with the gated structures that have been proved to offer good results in conventional RNNs, we propose the architecture presented in Fig. 5.4. This architecture is based on the conventional gated recurrent units (more precisely, on a simplified version of the Minimal GRU presented in [177]), so we call it permutation invariant gated recurrent unit (PI-GRU).

Instead of concatenating the input and the previous state as done in conventional GRUs, the PI-GRU uses multi-head attention [84] to obtain a new set  $\mathbf{C}(t) = {\mathbf{c}_1(t), \mathbf{c}_2(t), ..., \mathbf{c}_{N_h}(t)}$  based on the input set and the previous state set:

$$\mathbf{C}(t) = \text{MultiHead}(\mathbf{H}(t-1), \ \mathbf{X}(t) \cup \mathbf{H}(t-1), \mathbf{X}(t) \cup \mathbf{H}(t-1))$$
(5.1)

$$MultiHead(\mathbf{Q}, \mathbf{K}, \mathbf{V}) = Concat(head_1, ..., head_{N_{heads}})$$
(5.2)

$$\mathbf{head}_i = \operatorname{Attention}(\mathbf{QW}_i^Q, \mathbf{KW}_i^K, \mathbf{VW}_i^V)$$
(5.3)

Attention
$$(\mathbf{Q}_i, \mathbf{K}_i, \mathbf{V}_i) = \text{soft-max}\left(\frac{\mathbf{Q}_i \mathbf{K}_i^T}{\sqrt{d_k}}\right) \mathbf{V}_i$$
 (5.4)

where  $\mathbf{W}_{i}^{Q} \in \mathbb{R}^{d \times d_{k}}$ ,  $\mathbf{W}_{i}^{K} \in \mathbb{R}^{d \times d_{k}}$ , and  $\mathbf{W}_{i}^{Q} \in \mathbb{R}^{d \times d_{c}}$  are learnable projection matrices and the sizes of the keys and values  $d_{k}$  and  $d_{c}$  are hyperparameters of the model. It is worth mentioning that (5.2) is typically defined including a linear output projection [84] but we do not include it here since we are using the output of the multi-head attention to directly feed fully connected perceptrons instead of a residual sum as is done in transformer architectures.

Finally, after obtaining the set  $\mathbf{C}(t)$ , we use two fully connected perceptrons (or gates) to update every element  $\mathbf{h}_i(t)$  of  $\mathbf{H}(t)$  depending only on  $\mathbf{c}_i(t)$  and  $\mathbf{h}_i(t-1)$ :

$$\mathbf{h}_i(t) = [1 - \mathbf{z}_i(t)] \odot \mathbf{h}_i(t - 1) + \mathbf{h}_i(t)$$
(5.5)

$$\mathbf{z}_i(t) = \sigma(\mathbf{c}_i(t)\mathbf{W}^z) \tag{5.6}$$

$$\tilde{\mathbf{h}}_i(t) = \tanh(\mathbf{c}_i(t)\mathbf{W}^h), \tag{5.7}$$

where  $\mathbf{W}^{z}, \mathbf{W}^{h} \in \mathbb{R}^{N_{\text{heads}}d_{c} \times d}$  are learnable projection matrices,  $\odot$  stands for pointwise multiplication, and  $\sigma(\cdot)$  is the logistic sigmoid function.

We can interpret the role of the multi-head attention in the PI-GRU as reordering the information of the input set to make it match the order of the tracking trajectories in the state set. Actually,  $\mathbf{Q}_i \mathbf{K}_i^T$  in (5.4) is the cross-correlation matrix between linear projections of the elements of  $\mathbf{H}(t-1)$  (i.e., the information of every tracked trajectory) and linear projections of the elements of  $\mathbf{X}(t) \cup \mathbf{H}(t-1)$  (i.e., the information of every new detection and every tracked trajectory) and therefore we can expect it to, after passing through the soft-max function, act as some kind of soft permutation matrix that, when applied over  $\mathbf{V}_i$  (i.e., linear projections of  $\mathbf{X}(t) \cup \mathbf{H}(t-1)$  again), generates a new set whose i-th element is built with the information of the elements of  $\mathbf{X}(t) \cup \mathbf{H}(t-1)$  most related to the i-th element of  $\mathbf{H}(t-1)$ .

With this architecture, the computational complexity of the PI-GRU has a linear dependency with the number of tracked objects  $M = M_h$  and new detections  $M_{det} = M_x$  and the number of trainable parameters does not depend on them and, since every  $\mathbb{R}^d$  vector in  $\mathbf{X}(t)$  and  $\mathbf{H}(t)$  only have to store information about one detection or one tracked trajectory, we would expect to need lower values of  $d_x$  and  $d_h$  than in conventional GRUs.

To the best of our knowledge, the PI-GRU is the first recurrent layer that works with unordered sets instead of with ordered vectors. The closest proposal in the literature is probably the TrackFormer [178], a model for multi-object tracking (MOT) on video signals that is based on the DETR transformer [179, 180], a model for object detection on images. DETR uses a set of learnable *object queries* as input for their transformer decoder that, through the multi-head attention layers of the decoder, become the object detections. In the TrackFormer, a recurrent loop is built around the decoder by using as *object queries* of every video frame the outputs of the decoder in the previous frame, so it can track the objects that have already been detected. Compared with the TrackFormer, the PI-GRU is not a model but a layer that can easily be integrated into many different models and that is based on an architecture, the conventional GRU, that, unlike the transformer, was designed to be used in recurrent loops. In addition, several PI-GRU layers can easily be stacked to build deeper models with more recurrent loops, while the natural way of increasing the depth of the TrackFormer is increasing the number of layers of its transformer encoder and decoder, which does not increase the number of recurrences of the model.

# 5.4 Permutation invariant training (PIT)

Regardless we are taking into account the permutation invariance of the problem in the network architecture or not, we cannot ignore it during its training. When we want to train a multi-source localization or tracking model in a supervised manner and we cannot apply any criteria to classify and order the sources, we have to face the permutation invariance of the sources; i.e., we cannot directly compare the m-th trajectory estimated by the model  $\hat{\mathbf{y}}_m(t)$  with the m-th trajectory of our ground-truth dataset  $\mathbf{y}_m(t)$  since the model cannot infer the ground-truth order of the trajectories.

With the exception of [38], where a training strategy for MOT in computer vision tasks [181] is adapted to SSL, most multi-source localization models are trained using permutation invariant training (PIT). All the PIT strategies propose finding a permutation  $\sigma : m \to \sigma_m$ ,  $\forall m \in \{0, ..., M - 1\}$  according to certain optimization criteria to reorder the outputs of the neural network and then use it to compare the estimated and ground-truth trajectories. When using activity-coupled Cartesian direction of arrival (ACCDOA) vectors to represent the DOA and the activity of the sources, we can use the mean squared error (MSE) as the loss function to train our models:

$$L_{PIT} = \frac{1}{TM} \sum_{t=0}^{T-1} \sum_{m=0}^{M-1} \|\mathbf{y}_m(t) - \hat{\mathbf{y}}_{\sigma_m(t)}(t)\|^2,$$
(5.8)

where M is the maximum number of trajectories that the model can estimate, T is the number of time frames in the scene, and  $\|\cdot\|$  is the Euclidean norm operator. In the case of having a number of ground-truth trajectories lower than M, we can just add as many 0-norm padding trajectories as needed.

#### 5.4.1 Frame-level Permutation Invariant Training (fPIT)

The original PIT was first proposed for training speech separation models [182] but, applied to SSL [13, 62], it proposes to find the permutation of the estimated sources



Figure 5.5: Examples of 1D trajectories and the result of applying the different PIT strategies. The dashed lines represent the estimates and their color how every PIT pairs them with the ground-truth trajectories (solid lines). The red arrows represent the gradients of the MSE of every pairing w.r.t. the first estimated trajectory.

that minimizes the matching error between the estimated and the ground-truth DOAs for every time frame:

$$\sigma^{f}(t) = \underset{\sigma \in \Pi_{M}}{\operatorname{arg\,min}} \sum_{m=0}^{M-1} \|\mathbf{y}_{m}(t) - \hat{\mathbf{y}}_{\sigma_{m}}(t)\|,$$
(5.9)

where  $\Pi_M$  is the set of all the permutations  $\sigma : i \to \sigma_i$  of M elements. To solve this optimization problem, we can compute the  $M \times M$  distance matrices  $\mathbf{D}(t)$  with elements  $d_{ij}(t) = \|\mathbf{y}_i(t) - \hat{\mathbf{y}}_j(t)\|$  and apply the Hungarian algorithm [183] over them to find the optimal permutation at every time frame.

We call this approach frame-level permutation invariant training (fPIT) and, although it allows us to solve the permutation invariance problem, it does not penalize at all the IDSs. Instead, as can be seen in Fig. 5.5a, its gradients push the model to do these switches as fast as possible so the estimates are close to a ground-truth trajectory at all time frames. Therefore, if we want to keep the identity of every output stable during tracking, we need to add post-processing stages to fix the IDSs.

### 5.4.2 Utterance-level Permutation Invariant Training (uPIT)

In order to penalize the IDSs, utterance-level permutation invariant training (uPIT) [184] proposes finding the permutation that minimizes the error for a whole speech utterance or some other longer recording unit of interest, instead of a different one for every time frame:

$$\sigma^{u} = \underset{\sigma \in \Pi_{M}}{\operatorname{arg\,min}} \sum_{t=0}^{T-1} \sum_{m=0}^{M-1} \|\mathbf{y}_{m}(t) - \hat{\mathbf{y}}_{\sigma_{m}}(t)\|,$$
(5.10)

where T is the number of time frames of the acoustic scene. In this case, we only need to apply the Hungarian algorithm once per acoustic scene after computing the time average of the matrices  $\mathbf{D}(t)$ .

Replacing  $\sigma^{f}(t)$  by  $\sigma^{u}$  in (5.8) indeed penalizes the presence of IDSs since all the frames where the output ACCDOAs do not follow the main identity assignation compute as completely wrong estimates. However, as we can see in Fig. 5.5b, this penalization is excessive, being able to penalize situations that can not be solved by causal systems or generating gradients too much time after the IDS when, in most situations, it would be preferred to keep tracking the new identities rather than switching them again. Our experiments in using uPIT for multi-source tracking showed that it can easily generate a wide local minimum in the loss function that corresponds to estimating all DOAs in the middle of the active sources. That effect makes effective model training impossible, especially for long scenes of variable multiple sources.

# 5.4.3 Sliding Permutation Invariant Training (sPIT)

To overcome the limitations of both fPIT and uPIT, we propose a new PIT strategy that we call sliding permutation invariant training (sPIT) which consists in choosing, for every time frame, the optimal permutation for the last  $T_{avg}$  frames, i.e., for a causal sliding window of length  $T_{avg}$ :

$$\sigma^{s}(t) = \underset{\sigma \in \Pi_{M}}{\arg\min} \sum_{k=0}^{T_{avg}-1} \sum_{m=0}^{M-1} \|\mathbf{y}_{m}(t-k) - \hat{\mathbf{y}}_{\sigma_{m}}(t-k)\|.$$
(5.11)

In order to obtain  $\sigma^{s}(t)$  for every time frame, we can follow the same procedure as in the fPIT but applying a causal moving average of length  $T_{avg}$  over the elements of  $\mathbf{D}(t)$  before computing the Hungarian algorithm, so the computational complexity is virtually the same. It is worth mentioning that, in the case of training non-causal trackers, we could replace the causal moving window in (5.11) with a centered window.

As shown in Fig. 5.5c, the sPIT penalizes an estimation if it does not follow the main source assignation of the last  $T_{avg}$  time frames, while it stops penalizing an IDS after a maximum of  $T_{avg}$  frames and focuses on maintaining the new identities. Hence, the global minimum of the loss function corresponds to a solution without any IDSs but it also prevents the training from converging to the useless local minima generated by uPIT, estimating all DOAs in the middle point of the active sources.

In addition, when used over ACCDOA vectors, if the number of estimated sources is lower than the actual number, one of the estimated ACCDOA vectors whose norm is lower than the detection threshold will be paired with the ground-truth ACCDOA vector of the missed source and the gradients of (5.8) will pull that estimated ACC-DOA towards it. Similarly, in the case of a false positive, the gradients will pull the false-positive ACCDOA towards 0. Hence, sPIT is able to optimize both the source detections and the consistent source assignments that we expect from a competent SST system.

# 5.5 Evaluation

## 5.5.1 Experiment design

To evaluate the PI-GRUs and the PIT we used the dataset described in chapter 3 to generate acoustic scenes with up to 3 simultaneously active sources. During the 20 s of every scene, sources could appear and disappear, so the models do not only need to estimate the DOA of the sources but also their onsets and offsets.

We used the icoCNN described in chapter 4.2 as the basis of our tracking models. Since the output of this model was a 3D vector pointing towards the DOA of the source and whose norm was proportional to the confidence of the model in the estimation, we can already interpret it as an ACCDOA. We used the model for maps with resolution r = 3 and modified it by increasing the number of kernels of the hidden convolutional



Figure 5.6: Architecture of the permutation invariant RNN used to evaluate the tracking capabilities of the PI-GRUs.

layers from 32 to 128 and the number of kernels of the last convolutional layer (and therefore the number of output ACCDOAs) from 1 to  $M_{det}$ .

After the convolutional part, and once generated  $M_{det}$  permutation invariant AC-CDOAs, we used a small MLP to project every 3D ACCDOA into a space with  $d_x$ dimensions that we used to feed 2 PI-GRUs and then applied a final linear projection to convert every element of the output set into a 3D ACCDOA again as shown in Fig. 5.6. We used  $M = M_{det} = 10$  as the number of ACCDOA outputs since we observed that it was beneficial to use a number higher than the maximum possible number of active sources in the dataset (i.e., 3) and  $d_x = d_k = N_{heads}d_c = 128$  and  $N_{heads} = 4$ . With these hyperparameters, the permutation invariant RNN had 236 035 trainable parameters.

As baseline models, we studied the performance of the same icoCNN model without any RNN after it (so the tracking was done using only the temporal convolutions included in the icoCNN) and we also trained a model replacing the PI-GRUs of our permutation invariant RNN (Fig. 5.6) by conventional GRUs (Fig. 5.7). In order to feed the conventional GRUs with the set generated by the MLP, we needed to concatenate the M elements of the set into only one dimension and, to keep the number of trainable parameters in a reasonable range, we needed to reduce the size of the output of the GRUs. We used  $d_h = 10 \cdot M = 100$  as output size and, even if it meant having a state vector smaller than the one used to represent just a single trajectory with the PI-GRUs, it resulted in an RNN with 509153 trainable parameters, more than the double than with the PI-GRUs. Here, we can clearly see the compromise explained in section 5.3 between the model complexity and the amount of information that it can keep about every source and it would become even more extreme for higher values of M.

#### 5. PERMUTATION INVARIANT MULTI-SOURCE TRACKING



Figure 5.7: Architecture of the RNN used after the icoCNN as baseline in the evaluation.

We trained the three models using both conventional fPIT and the proposed sPIT, we did not include uPIT in the evaluation since it did not converge to any practical solution. We found that the model including the PI-GRUs was especially hard to train and in many cases it did not converge. Following some usual practices when training recurrent models, we used the AdamW algorithm [185] with gradient clipping [186] and reinitialized the state vectors to a learnable initial state every 25 time frames during the first 50 epochs and then every 50 frames during the following 100 epochs. For a fair comparison, we applied the same strategy when training the model with the conventional GRUs and, to facilitate the training of the convolutional part of the models, we also included a fPIT loss over the ACCDOAs generated by the icoCNN. All the results shown in the following section were obtained using  $T_{avg} = 10$  frames (i.e., 2s) for the sPIT, but no large changes were observed with longer or shorter windows.

## 5.5.2 Results

Fig. 5.8 shows the mean angular error (MAE), the identity switch (IDS) rate, and the receiver operating characteristic (ROC) and detection error tradeoff (DET) curves resulting from training each one of the three evaluated models. The MAEs and the IDS, true positive, and false positive rates were calculated following a procedure similar to the one defined by the CLEAR MOT metrics [187] typically used for multi-object tracking in computer vision. In every time frame, each estimate was associated with its closer ground-truth DOA and any estimate that could not be associated with a ground-truth DOA closer to a threshold (which we set at 30° as done in the evaluation of the results of the LOCATA challenge [188]) was considered a false positive. After



Figure 5.8: Evaluation metrics obtained using sPIT and fPIT to train the three analyzed models.

that, in every ground-truth trajectory that had already been assigned to an estimated trajectory in the previous time frame, an IDS was considered to have occurred if now it had been assigned to a different estimated trajectory. Finally, the MAE was defined as the sum of the angular distances of the true positives during the whole dataset divided by the total number of true positives and the IDS, true positive, and false positive rates, as the number of IDSs, true positives, and false positives in the whole dataset divided by the total number of ground-truth DOAs. The MAE and the IDS rate were computed considering a detection threshold of 0.5 over the norm of the estimated ACCDOA vectors and for computing the ROC curve we analyzed the true positive and false positive rates for a range of detection thresholds from 0 to 1.

As could be expected, the metric that is most affected by the PIT strategy is the IDS rate. We can see how using the proposed sPIT we can train purely convolutional models to obtain rates under 0.1 and how it reduces the number of IDSs of the model using conventional GRUs by more than a factor 2. The model using the proposed PI-GRUs had a quite high number of IDSs when trained with the fPIT, but when trained with the proposed sPIT the number was reduced to the same level as the model using conventional GRUs. Since the fPIT does not penalize the IDSs, the IDS rate of the models trained with it depends only on their ability to generate abrupt changes in their outputs because the switches are penalized only if they occur slowly and generate a high localization error till they are completed.

In terms of localization error, all the evaluated models have similar performance, with a difference of only 2° between the best and the worst MAE. For the model doing the tracking using only temporal convolutions and for the one using conventional GRUs the use of sPIT slightly degrades the localization accuracy but, in the case of the model using PI-GRUs, the sPIT improves its localization performance.

Finally, we can see in the ROC and DET curves how the compromise between the true and false positive rates for the models trained with sPIT are quite similar, especially in the models using recurrent layers for tracking. Again, we can see how sPIT slightly degrades the performance of the model using only convolutions and the model using GRUs but improves the results of the model using PI-GRUs. As examples, figures 5.9, 5.10, and 5.11 show trajectories estimated by the evaluated models in acoustic scenes with up to 3 concurrent sources. We can see how the models trained with sPIT do not have the high number of IDSs that the models trained with fPIT have, though the model without recurrent layers still has some switches even when trained with sPIT. We can also see how the models using recurrent layers for tracking clearly outperform the model using only temporal convolutions but there are no clear differences between the model using conventional GRUs and the model using the proposed PI-GRUs. Finally, it is worth mentioning that any of the models seem to really be able to track three concurrent sources.

# 5.6 Conclusions

In this chapter, we have studied the permutation invariance of the multi-source tracking systems and proposed a new permutation invariant training (PIT) strategy to train tracking models and a new recurrent layer that takes an unordered set as input and that is invariant to the permutation of its elements.

We have proven how the proposed sPIT dramatically reduces the number of identity switches (IDSs) compared with the state-of-the-art fPIT, avoiding the need for using additional stages at the output of the tracking models to keep the identity of every tracked trajectory stable. The loss function of sPIT and the gradients that it generates are easy to interpret and it has a similar computational complexity to fPIT. When used over ACCDOAs, sPIT also optimizes the detection accuracy of the model, improving the ratio between the true and false positive rates and therefore optimizing all the features that we would expect from a sound source tracking (SST) system.

About the proposed permutation invariant gated recurrent unit (PI-GRU), even if we can argue that they outperformed the conventional GRUs since the models that used them obtained the same IDS rate and ROC curve as the model with the conventional GRUs and had a lower mean angular error (MAE), this difference probably does not worth the increase in the computational complexity. However, we think that the results obtained are promising since they represent the first attempt at developing

#### 5. PERMUTATION INVARIANT MULTI-SOURCE TRACKING

a recurrent layer for unordered sets and there is still room for improving and optimizing them. In the evaluated models, only geometrical information was provided to the recurrent layers to perform the tracking, but using models that include spectral information of every source would better exploit their ability to increase the size of the vectors representing every detection and tracked trajectory without needing to increase the number of trainable parameters in a quadratic way as happens with the conventional GRUs. Some ideas about how to improve both the PI-GRUs and the whole model are discussed in chapter 6.



**Figure 5.9:** Example of the trajectories estimated by the evaluated models in an evaluation acoustic with up to 2 concurrent sound sources. The solid lines represent the ground-truth trajectories and the dashed lines the estimates.



Figure 5.10: Example of the trajectories estimated by the evaluated models in an evaluation acoustic with up to 3 concurrent sound sources. The solid lines represent the ground-truth trajectories and the dashed lines the estimates.



Figure 5.11: Another example of the trajectories estimated by the evaluated models in an evaluation acoustic with up to 3 concurrent sound sources. The solid lines represent the ground-truth trajectories and the dashed lines the estimates.

6

# Conclusions and future work

In this chapter, we first summarize the conclusions of the thesis (section 6.1) and then we outline some of the research directions that we think have a greater potential to improve the performance the sound source localization (SSL) and sound source tracking (SST) systems based on deep learning in the near future (section 6.2).

# 6.1 Conclusions

- The signal processing community has studied the sound source localization (SSL) problem for decades and the classical techniques that were proposed before the emergence of the deep-learning solutions should not be neglected since they can provide acoustic representations with a higher correlation with the direction of arrival (DOA) of the sources and a better analytical understanding of the problem that should guide the design of the new deep-learning solutions.
- We can eliminate the effects that a source in a given position generates in the inter-microphone generalized cross-correlations (GCCs) by just doing a linear combination of time-shifted versions of the original GCC.
- When using synthetic datasets to train SSL models, we can generate unlimited different random trajectories and therefore we do not need to limit our datasets to a finite number of trajectories. Although this should be used with caution,

with the models employed in this work we have not observed signs of over-fitting to the trajectory generation procedure even if it was analytically quite simple.

- Many of the stages of the image source method (ISM) for room acoustic simulation can be parallelized and efficiently implemented in graphics processing units (GPUs). This is essential if we want to train our models with infinite-size datasets simulated on the fly, since the ISM is computationally expensive and its sequential implementation would slow down the trainings too much. The proposed GPU implementation reduces the simulation time by two orders of magnitude compared with other state-of-the-art implementations.
- The acoustic power maps generated with the classic steered response power with phase transform (SRP-PHAT) algorithm are a robust input representation for neural networks. Though their main source of information for one-source localization is the position of their maximum, neural networks are able to extract additional information from them.
- For one-source tracking, it is possible to obtain competitive results using only causal temporal convolutions avoiding the use of recurrent layers. Bi-directional recurrent layers should be avoided if we want our models to be useful for real-time applications.
- Using models that exploit the rotational invariance of the DOA estimation problem allows us to reduce the size of our models. Several architectures are invariant to the continuous space of spherical rotations, but the icosahedral convolutional neural networks (CNNs) provide a good approximation with their 60 rotational symmetries while having an efficient implementation based on conventional 2D convolutional layers.
- The proposed soft-argmax layer can transform a classification output into a regression one by interpreting the classification output as the probability distribution of the DOA and computing its expected value. It strongly reduces the number of trainable parameters of the model and avoids breaking its rotational equivariance.

- In permutation invariant training (PIT) strategies for multi-source tracking, the temporal context that we consider when choosing the best permutation of the estimated sources is crucial to reduce the number of identity switches (IDSs) in the tracked trajectories. Using a sliding window as we have proposed with the sliding permutation invariant training (sPIT) is a good trade-off between the high number of IDSs obtained with the frame-level permutation invariant training (fPIT) and the convergence issues presented by the utterance-level permutation invariant training (uPIT) for long acoustic scenes.
- We can design permutation invariant recurrent layers for deep learning models that operate over unordered sets instead of over ordered vectors by using a multihead attention module to match the elements of the input set with the elements of the state set and then using element-wise gates to update the information of every state element.
- The use of permutation invariant recurrent layers, such as the proposed permutation invariant gated recurrent unit (PI-GRU), for SST of multiple sources is a promising research direction, but further work is needed to make them competitive with the traditional gated recurrent units (GRUs).

# 6.2 Future work

# 6.2.1 Training data

In section 4.1.4, we saw how all the evaluated models had a worse performance when evaluated in the LOCATA dataset than when evaluated with the synthetic dataset used to train them and the performance gap became ever wider in the case of the model proposed in 4.2 using icosahedral CNNs. The optimal solution would be training the models using actual recordings, but there are no multichannel-audio datasets with ground-truth position labels large enough to do this. Contrary to image datasets, position labels in audio datasets cannot be obtained after the signals are recorded and special hardware is needed to accurately label the source position of moving sources [138]. Therefore, it does not seem to be reasonable to expect the appearance of larger position-labeled datasets in the near future and techniques to improve the performance of models trained with synthetic signals should be found.

There exist some datasets with recorded multichannel room impulse responses (RIRs) and position labels (e.g., [189, 190, 191, 192]) that could be used to replace the simulated RIRs and get more realistic synthetic training datasets, but they are usually not large enough, especially in terms of acoustic conditions. Another approach to improve the quality of the synthetic datasets would be improving RIRs simulated with the ISM by including more acoustic effects [193] such as source directivity [194] or acoustic diffraction on the walls or the array [195].

Finally, an approach that is in growing popularity is reducing the need for position labels by using semi-supervised or weakly-supervised training strategies. For example, some domain adaptation techniques have been proposed where the models are trained with synthetic signals following a supervised approach but then actual recordings without position labels are used to reduce the performance gap between the synthetic and the recorded signals [46, 64, 65, 196]. A different approach is followed in [49, 50], where a large dataset of unlabeled multichannel recordings is used to train an autoencoder to reconstruct the phase of the relative transfer functions between microphones from two latent variables while a smaller dataset with labeled recordings is used to train one of those variables to be the DOA of the acoustic source.

# 6.2.2 Rotation-equivariant models for SSL

In this thesis, we have proposed an SSL model based on applying an icosahedral CNN over SRP-PHAT power maps. However, other approaches could also be followed to design rotation-equivariant models. There are several signal processing techniques, such as the multiple signal classification (MUSIC) algorithm, that can be used to extract positional information of multichannel signals that can be computed on arbitrary grids and that, therefore, could be used to replace the SRP-PHAT power maps as model inputs or to complement them by stacking them as additional input channels. There are also other network architectures designed to be equivariant to the spherical rotations, such as those based on the spherical-harmonic domain [77, 168, 169, 170] or

on graphs [171, 172, 173], so a study comparing the performance of different input representations and network architectures would be interesting.

A more radically different approach would be finding other kinds of acoustic representations where it is possible to exploit the rotational equivariance. For example, a rotation-equivariant network is proposed in [197] that works over the ambisonic representation of the audio signals.

## 6.2.3 Permutation invariant recurrent networks

In section 5.3, we introduced the PI-GRU, which is, to the best of our knowledge, the first recurrent neural layer whose input and recurrent state are unordered sets and which is invariant to the permutations of the elements of the input set and equivariant to the permutations of the state set. We designed the architecture of the PI-GRU based on a simplified version of the minimal GRU [177] but other architectures, such as the ones shown in Fig. 6.1, could also be interesting. A further evaluation of permutation invariant recurrent architectures would be interesting in order to analyze the advantages and disadvantages of every one of them.

Actually, even with the proposed PI-GRU architecture, we did not conduct any systematic analysis of how its hyperparameters (such as the number of heads or the embedding sizes) affected the tracking performance of the model, so this would also need further study in the future.

We also found that the PI-GRUs were considerably harder to train than the conventional GRUs, exhibiting a slower convergence and even not converging to any solution during some trainings. Therefore, it would be also interesting to study which optimization algorithms, regularization techniques, and training strategies could facilitate the training of these models.

Finally, it is worth mentioning that, during the experiments conducted in section 5.5, the recurrent state of the PI-GRUs (and also of the baseline GRUs) was initialized to a learnable set of embedding (or to a learnable vector) at the beginning of each training or evaluation acoustic scene and we let it to evolve during the scene independently of whether new sources were found or not. A different approach that would be worth studying would be, as done with the TrackFormer [178] for visual



Figure 6.1: Different permutation invariant recurrent architectures. (a) is just a minimal recurrent unit, (b) is based on the minimal GRU and (c) is based on a complete GRU.

multi-object tracking (MOT) on videos, keeping always the learned embeddings in the recurrent state set and, after every time frame, adding to it the elements of the output that led to active sources in the previous frame. If the new elements are added to the state set without removing the elements that had generated them as done in [178], this would mean that the size of the state set varies over time, but this is compatible with the PI-GRU architecture presented in section 5.3 and also with the alternative architectures shown in Fig. 6.1.

# 6.2.4 Eliminating the information bottlenecks of the proposed model

In section 4.1, we argued that combining localization and tracking into a single stage by using 3D convolutions that extended their receptive field over the temporal dimension was preferable to first doing the localization using the acoustic information of just one time frame and then filtering this result with a tracking algorithm considering the localization results of the previous frames. However, our models also present some information bottlenecks where we could also be losing potentially useful information.

The first information bottleneck of our model happens at its very beginning: at the input feature selection. By using SRP-PHAT maps as the only input of our model, we are losing all the spectral information of the signals. This could be reasonable for single-source localization since we wanted to locate the only directional source present in the signals regardless of its spectral content, but for multi-source tracking, the spectral information provides crucial information to, for example, avoid IDSs.

One of the easiest ways to integrate spectral information into the model proposed in chapter 5 would be using classic beamformers to obtain the acoustic spectra at every DOA estimated by the icosahedral CNN and then concatenate it to the corresponding ACCDOA so every element in the input set of the PI-GRU contains both spatial and spectral information. This way, the correlation matrix computed in the multi-head attention module would match the elements of the input and state sets not only based on geometrical information but also based on the similarities of their spectrum.

A different approach would be replacing the input power maps with another icosahedral signal that, at every point of the grid, included spectral information obtained using a beamformer steered at the corresponding direction. Considering every frequency bin as a convolution channel, the same convolutional architecture could be easily used, though we would probably need to increase the number of channels in the hidden layers to fully exploit the spectral information. However, feeding the PI-GRU with the spectral information processed by the icosahedral CNN would be more difficult, since every element of the input set should contain information from only one of the detected sources. An option could be increasing the number of channels of the last convolutional layer from M to M + S, so the first M channels could still be processed with the soft-argmax layer to obtain the ACCDOAs of the detected sources and filter each one of the remaining S channels with every one of the first M channels to finally integrate the  $M \times S$  resulting maps over their spatial dimensions so we would obtain M vectors of length S and concatenate every one of them to the corresponding ACCDOA.

# References

- RICHARD W. CARLISLE AND ARNOLD SCHWARTZ. Evaluation of a Stereophonic Loudspeaker by Multiple Microphone Arrays. The Journal of the Acoustical Society of America, 31(10):1348–1351, October 1959. 1
- [2] MICHAEL E. AUSTIN AND RICHARD B. GOMEZ. Azimuth and Elevation Errors Inherent in Sound-Ranging Calculations. The Journal of the Acoustical Society of America, 44(1):25– 27, July 1968. 1
- [3] MICHAEL BRANDSTEIN AND DARREN WARD. Microphone Arrays: Signal Processing Techniques and Applications. Springer Berlin Heidelberg, Berlin, Heidelberg, 2001. 1
- [4] PIERRE-AMAURY GRUMIAUX, SRDAN KITIĆ, LAURENT GIRIN, AND ALEXANDRE GUÉRIN. A Survey of Sound Source Localization with Deep Learning Methods. The Journal of the Acoustical Society of America, 152(1):107–151, July 2022. 3
- [5] DMITRY SUVOROV, GE DONG, AND ROMAN ZHUKOV. Deep Residual Network for Sound Source Localization in the Time Domain, August 2018. 3
- [6] JUAN MANUEL VERA-DIAZ, DANIEL PIZARRO, AND JAVIER MACIAS-GUARASA. Towards Endto-End Acoustic Localization Using Deep Learning: From Audio Signals to Source Position Coordinates. Sensors, 18(10):3418, October 2018. 3
- [7] HADRIEN PUJOL, ÉRIC BAVU, AND ALEXANDRE GARCIA. BeamLearning: An End-to-End Deep Learning Approach for the Angular Localization of Sound Sources Using Raw Multichannel Acoustic Pressure Data. The Journal of the Acoustical Society of America, 149(6):4248–4263, June 2021. 3
- [8] PASI PERTILÄ AND EMRE CAKIR. Robust Direction Estimation with Convolutional Neural Networks Based Steered Response Power. In 2017 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 6125–6129, March 2017. 4, 6
- [9] NELSON YALTA, KAZUHIRO NAKADAI, AND TETSUYA OGATA. Sound Source Localization Using Deep Learning Models. Journal of Robotics and Mechatronics, 29(1):37–48, 2017.
   4
- [10] ZHONG-QIU WANG, XUELIANG ZHANG, AND DELIANG WANG. Robust Speaker Localization Guided by Deep Learning-Based Time-Frequency Masking. *IEEE/ACM Transactions* on Audio, Speech, and Language Processing, 27(1):178–188, January 2019. 4

- [11] WANGYOU ZHANG, YING ZHOU, AND YANMIN QIAN. Robust DOA Estimation Based on Convolutional Neural Network and Time-Frequency Masking. In Interspeech 2019, pages 2703–2707. ISCA, September 2019. 4
- [12] S. CHAKRABARTY AND E. A. P. HABETS. Multi-Speaker DOA Estimation Using Deep Convolutional Networks Trained With Noise Signals. *IEEE Journal of Selected Topics* in Signal Processing, 13(1):8–21, March 2019. 4, 9
- [13] ASWIN SHANMUGAM SUBRAMANIAN, CHAO WENG, SHINJI WATANABE, MENG YU, AND DONG YU. Deep Learning Based Multi-Source Localization with Source Splitting and Its Effectiveness in Multi-Talker Speech Recognition. Computer Speech & Language, 75:101360, September 2022. 4, 6, 10, 121
- [14] S. ADAVANNE, A. POLITIS, J. NIKUNEN, AND T. VIRTANEN. Sound Event Localization and Detection of Overlapping Sources Using Convolutional Recurrent Neural Networks. *IEEE Journal of Selected Topics in Signal Processing*, 13(1):34–48, March 2019. 4, 6, 84, 86, 117
- [15] KARIM GUIRGUIS, CHRISTOPH SCHORN, ANDRE GUNTORO, SHERIF ABDULATIF, AND BIN YANG. SELD-TCN: Sound Event Localization & Detection via Temporal Convolutional Networks. In 2020 28th European Signal Processing Conference (EUSIPCO), pages 16–20, Amsterdam, Netherlands, January 2021. IEEE. 4
- [16] WEIPENG HE, PETR MOTLICEK, AND JEAN-MARC ODOBEZ. Neural Network Adaptation and Data Augmentation for Multi-Speaker Direction-of-Arrival Estimation. IEEE/ACM Transactions on Audio, Speech, and Language Processing, 29:1303–1317, 2021. 4, 6
- [17] CHRISTOPHER SCHYMURA, BENEDIKT BÖNNINGHOFF, TSUBASA OCHIAI, MARC DELCROIX, KEISUKE KINOSHITA, TOMOHIRO NAKATANI, SHOKO ARAKI, AND DOROTHEA KOLOSSA. PILOT: Introducing Transformers for Probabilistic Sound Event Localization. In Interspeech 2021, pages 2117–2121. ISCA, August 2021. 4
- [18] WEIPENG HE, PETR MOTLICEK, AND JEAN-MARC ODOBEZ. Joint Localization and Classification of Multiple Sound Sources Using a Multi-task Neural Network. In Interspeech 2018, pages 312–316. ISCA, September 2018. 4
- [19] GUILLAUME LE MOING, PHONGTHARIN VINAYAVEKHIN, TADANOBU INOUE, JAYAKORN VONGKULBHISAL, ASIM MUNAWAR, RYUKI TACHIBANA, AND DON JOVEN AGRAVANTE. Learning Multiple Sound Source 2D Localization. In 2019 IEEE 21st International Workshop on Multimedia Signal Processing (MMSP), pages 1–6, September 2019. 4
- [20] YIYA HAO, ABDULLAH KÜÇÜK, ANSHUMAN GANGULY, AND ISSA M. S. PANAHI. Spectral Flux-Based Convolutional Neural Network Architecture for Speech Source Localization and Its Real-Time Implementation. *IEEE Access*, 8:197047–197058, 2020. 4
- [21] DANIEL KRAUSE, ARCHONTIS POLITIS, AND KONRAD KOWALCZYK. Comparison of Convolution Types in CNN-based Feature Extraction for Sound Source Localization. In 2020 28th European Signal Processing Conference (EUSIPCO), pages 820–824, Amsterdam, Netherlands, January 2021. IEEE. 4

- [22] X. XIAO, S. ZHAO, X. ZHONG, D. L. JONES, E. S. CHNG, AND H. LI. A Learning-Based Approach to Direction of Arrival Estimation in Noisy and Reverberant Environments. In 2015 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 2814–2818, April 2015. 4, 5
- [23] F. VESPERINI, P. VECCHIOTTI, E. PRINCIPI, S. SQUARTINI, AND F. PIAZZA. A Neural Network Based Algorithm for Speaker Localization in a Multi-Room Environment. In 2016 IEEE 26th International Workshop on Machine Learning for Signal Processing (MLSP), pages 1-6, September 2016. 4
- [24] LUCA COMANDUCCI, FEDERICO BORRA, PAOLO BESTAGINI, FABIO ANTONACCI, STEFANO TUBARO, AND AUGUSTO SARTI. Source Localization Using Distributed Microphones in Reverberant Environments Based on Deep Learning and Ray Space Transform. IEEE/ACM Transactions on Audio, Speech, and Language Processing, 28:2238–2251, 2020. 4
- [25] QINGLONG LI, XUELIANG ZHANG, AND HAO LI. Online Direction of Arrival Estimation Based on Deep Learning. In 2018 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 2616–2620, April 2018. 4
- [26] LAURÉLINE PEROTIN, ROMAIN SERIZEL, EMMANUEL VINCENT, AND ALEXANDRE GUÉRIN. CRNN-Based Multiple DoA Estimation Using Acoustic Intensity Features for Ambisonics Recordings. IEEE Journal of Selected Topics in Signal Processing, 13(1):22–33, March 2019. 4, 117
- [27] PIERRE-AMAURY GRUMIAUX, SRDAN KITIĆ, PRERAK SRIVASTAVA, LAURENT GIRIN, AND ALEXANDRE GUÉRIN. Saladnet: Self-Attentive Multisource Localization in the Ambisonics Domain. In 2021 IEEE Workshop on Applications of Signal Processing to Audio and Acoustics (WASPAA), pages 336–340, October 2021. 4
- [28] THI NGOC THO NGUYEN, NGOC KHANH NGUYEN, HUY PHAN, LAM PHAM, KENNETH OOI, DOUGLAS L. JONES, AND WOON-SENG GAN. A General Network Architecture for Sound Event Localization and Detection Using Transfer Learning and Recurrent Neural Network. In ICASSP 2021 - 2021 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 935–939, June 2021. 4
- [29] R. SCHMIDT. Multiple Emitter Location and Signal Parameter Estimation. IEEE Transactions on Antennas and Propagation, 34(3):276–280, March 1986. 4, 16
- [30] R. ROY AND T. KAILATH. ESPRIT-estimation of Signal Parameters via Rotational Invariance Techniques. IEEE Transactions on Acoustics, Speech, and Signal Processing, 37(7):984–995, July 1989. 4, 16
- [31] JOSEPH HECTOR DIBIASE. A High-Accuracy, Low-Latency Technique for Talker Localization in Reverberant Environments Using Microphone Arrays. PhD thesis, Brown University, 2000.
   4, 16, 19
- [32] JOSEPH HECTOR DIBIASE, HARVEY F. SILVERMAN, AND MICHAEL BRANDSTEIN. Robust Localization in Reverberant Rooms. In Microphone Arrays: Signal Processing Techniques and Applications. Springer Berlin Heidelberg, Berlin, Heidelberg, 2001. 4, 15, 16, 19
- [33] R. TAKEDA AND K. KOMATANI. Discriminative Multiple Sound Source Localization Based on Deep Neural Networks Using Independent Location Model. In 2016 IEEE Spoken Language Technology Workshop (SLT), pages 603–609, December 2016. 4

- [34] THI NGOC THO NGUYEN, WOON-SENG GAN, RISHABH RANJAN, AND DOUGLAS L. JONES. Robust Source Counting and DOA Estimation Using Spatial Pseudo-Spectrum and Convolutional Neural Network. IEEE/ACM Transactions on Audio, Speech, and Language Processing, 28:2626-2637, 2020. 4
- [35] DANIELE SALVATI, CARLO DRIOLI, AND GIAN LUCA FORESTI. Exploiting CNNs for Improving Acoustic Source Localization in Noisy and Reverberant Conditions. IEEE Transactions on Emerging Topics in Computational Intelligence, 2(2):103–116, April 2018. 4
- [36] E. L. FERGUSON, S. B. WILLIAMS, AND C. T. JIN. Sound Source Localization in a Multipath Environment Using Convolutional Neural Networks. In 2018 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 2386–2390, April 2018. 4
- [37] PAOLO VECCHIOTTI, EMANUELE PRINCIPI, STEFANO SQUARTINI, AND FRANCESCO PIAZZA. Deep Neural Networks for Joint Voice Activity Detection and Speaker Localization. In 2018 26th European Signal Processing Conference (EUSIPCO), pages 1567–1571, September 2018. 4
- [38] SHARATH ADAVANNE, ARCHONTIS POLITIS, AND TUOMAS VIRTANEN. Differentiable Tracking-Based Training of Deep Learning Sound Source Localizers. In 2021 IEEE Workshop on Applications of Signal Processing to Audio and Acoustics (WASPAA), pages 211–215, New Paltz, NY, USA, October 2021. IEEE. 4, 6, 117, 121
- [39] DANIEL KRAUSE, ARCHONTIS POLITIS, AND KONRAD KOWALCZYK. Feature Overview for Joint Modeling of Sound Event Detection and Localization Using a Microphone Array. In 2020 28th European Signal Processing Conference (EUSIPCO), pages 31–35, Amsterdam, Netherlands, January 2021. IEEE. 4, 9
- [40] LUCA COMANDUCCI, MAXIMO COBOS, FABIO ANTONACCI, AND AUGUSTO SARTI. Time Difference of Arrival Estimation from Frequency-Sliding Generalized Cross-Correlations Using Convolutional Neural Networks. In ICASSP 2020 - 2020 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 4945–4949, May 2020. 4
- [41] YIN CAO, QIUQIANG KONG, TURAB IQBAL, FENGYAN AN, WENWU WANG, AND MARK PLUMB-LEY. Polyphonic Sound Event Detection and Localization Using a Two-Stage Strategy. In Proceedings of the Detection and Classification of Acoustic Scenes and Events 2019 Workshop (DCASE2019), pages 30–34, New York University, NY, USA, October 2019. 4, 117
- [42] PRANAY PRATIK, WEN JIE JEE, SRIKANTH NAGISETTY, ROHITH MARS, AND CHONGSOON LIM. Sound Event Localization and Detection Using CRNN Architecture with Mixup for Model Generalization. In Proceedings of the Detection and Classification of Acoustic Scenes and Events 2019 Workshop (DCASE2019), pages 199–203. New York University, 2019. 4, 117
- [43] HUY PHAN, L. PHAM, P. KOCH, NGOC Q. K. DUONG, I. MCLOUGHLIN, AND A. MERTINS. On Multitask Loss Function for Audio Event Detection and Localization. undefined, 2020. 5
- [44] WOLFGANG MACK, ULLAS BHARADWAJ, SOUMITRO CHAKRABARTY, AND EMANUËL A. P. HA-BETS. Signal-Aware Broadband DOA Estimation Using Attention Mechanisms. In ICASSP 2020 - 2020 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 4930–4934, May 2020. 5

- [45] CHRISTOPHER SCHYMURA, TSUBASA OCHIAI, MARC DELCROIX, KEISUKE KINOSHITA, TOMO-HIRO NAKATANI, SHOKO ARAKI, AND DOROTHEA KOLOSSA. Exploiting Attention-based Sequence-to-Sequence Architectures for Sound Event Localization. In 2020 28th European Signal Processing Conference (EUSIPCO), pages 231–235, January 2021. 5
- [46] GUILLAUME LE MOING, PHONGTHARIN VINAYAVEKHIN, DON JOVEN AGRAVANTE, TADANOBU INOUE, JAYAKORN VONGKULBHISAL, ASIM MUNAWAR, AND RYUKI TACHIBANA. Data-Efficient Framework for Real-World Multiple Sound Source 2d Localization. In ICASSP 2021 - 2021 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 3425–3429, June 2021. 5, 6, 108, 138
- [47] YIFAN WU, ROSHAN AYYALASOMAYAJULA, MICHAEL J. BIANCO, DINESH BHARADIA, AND PETER GERSTOFT. SSLIDE: Sound Source Localization for Indoors Based on Deep Learning. In ICASSP 2021 - 2021 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 4680–4684, June 2021. 5
- [48] JUAN MANUEL VERA-DIAZ, DANIEL PIZARRO, AND JAVIER MACIAS-GUARASA. Acoustic Source Localization with Deep Generalized Cross Correlations. Signal Processing, 187:108169, October 2021. 5, 6
- [49] MICHAEL J. BIANCO, SHARON GANNOT, AND PETER GERSTOFT. Semi-Supervised Source Localization with Deep Generative Modeling. In 2020 IEEE 30th International Workshop on Machine Learning for Signal Processing (MLSP), pages 1–6, September 2020. 5, 138
- [50] MICHAEL J. BIANCO, SHARON GANNOT, EFREN FERNANDEZ-GRANDE, AND PETER GERSTOFT. Semi-Supervised Source Localization in Reverberant Environments With Deep Generative Modeling. *IEEE Access*, 9:84956–84970, 2021. 5, 6, 138
- [51] SHARATH ADAVANNE, ARCHONTIS POLITIS, AND TUOMAS VIRTANEN. Direction of Arrival Estimation for Multiple Sound Sources Using Convolutional Recurrent Neural Network. In 2018 26th European Signal Processing Conference (EUSIPCO), pages 1462–1466, September 2018. 5
- [52] YINGXIANG SUN, JIAJIA CHEN, CHAU YUEN, AND SUSANTO RAHARDJA. Indoor Sound Source Localization With Probabilistic Neural Network. *IEEE Transactions on Industrial Electronics*, 65(8):6403–6413, August 2018. 5
- [53] LAURÉLINE PEROTIN, ALEXANDRE DÉFOSSEZ, EMMANUEL VINCENT, ROMAIN SERIZEL, AND ALEXANDRE GUÉRIN. Regression Versus Classification for Neural Network Based Audio Source Localization. In 2019 IEEE Workshop on Applications of Signal Processing to Audio and Acoustics (WASPAA), pages 343–347, October 2019. 6, 84, 98
- [54] ZHENYU TANG, JOHN D. KANU, KEVIN HOGAN, AND DINESH MANOCHA. Regression and Classification for Direction-of-Arrival Estimation with Convolutional Recurrent Neural Networks. In Interspeech 2019, pages 654–658. ISCA, September 2019. 6, 84, 98
- [55] SLAWOMIR KAPKA AND MATEUSZ LEWANDOWSKI. Sound Source Detection, Localization and Classification Using Consecutive Ensemble of CRNN Models. In Proceedings of the Detection and Classification of Acoustic Scenes and Events 2019 Workshop (DCASE2019), pages 119–123. New York University, 2019. 6

- [56] SOTIRIOS PANAGIOTIS CHYTAS AND GERASIMOS POTAMIANOS. Hierarchical Detection of Sound Events and Their Localization Using Convolutional Neural Networks with Adaptive Thresholds. In Proceedings of the Detection and Classification of Acoustic Scenes and Events 2019 Workshop (DCASE2019), pages 50–54, New York University, NY, USA, October 2019. 6
- [57] SOOYOUNG PARK. TrellisNet-Based Architecture for Sound Event Localization and Detection with Reassembly Learning. In Proceedings of the Detection and Classification of Acoustic Scenes and Events 2019 Workshop (DCASE2019), pages 179–183, New York University, NY, USA, October 2019. 6
- [58] KAZUKI SHIMADA, YUICHIRO KOYAMA, NAOYA TAKAHASHI, SHUSUKE TAKAHASHI, AND YUKI MITSUFUJI. Accdoa: Activity-Coupled Cartesian Direction of Arrival Representation for Sound Event Localization And Detection. In ICASSP 2021 - 2021 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 915–919, June 2021. 6, 98, 117
- [59] JUNHYEONG PAK AND JONG WON SHIN. Sound Localization Based on Phase Difference Enhancement Using Deep Neural Networks. *IEEE/ACM Transactions on Audio, Speech,* and Language Processing, 27(8):1335–1345, August 2019. 6
- [60] YANKUN HUANG, XIHONG WU, AND TIANSHU QU. A Time-domain Unsupervised Learning Based Sound Source Localization Method. In 2020 IEEE 3rd International Conference on Information Communication and Signal Processing (ICICSP), pages 26–32, September 2020. 6
- [61] JONT B. ALLEN AND DAVID A. BERKLEY. Image Method for Efficiently Simulating Smallroom Acoustics. The Journal of the Acoustical Society of America, 65(4):943–950, April 1979. 7, 29, 56, 59, 60, 101
- [62] DANIEL KRAUSE, ARCHONTIS POLITIS, AND KONRAD KOWALCZYK. Data Diversity for Improving DNN-based Localization of Concurrent Sound Events. In 2021 29th European Signal Processing Conference (EUSIPCO), pages 236–240, August 2021. 7, 117, 121
- [63] YI LUO AND JIANWEI YU. FRA-RIR: Fast Random Approximation of the Image-source Method, August 2022. 7, 49
- [64] RYU TAKEDA, YOSHIKI KUDO, KAZUKI TAKASHIMA, YOSHIFUMI KITAMURA, AND KAZUNORI KO-MATANI. Unsupervised Adaptation of Neural Networks for Discriminative Sound Source Localization with Eliminative Constraint. In 2018 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 3514–3518, April 2018. 7, 108, 138
- [65] PASI PERTILÄ, MIKKO PARVIAINEN, VILLE MYLLYLÄ, ANU HUTTUNEN, AND PETRI JARSKE. Time Difference of Arrival Estimation with Deep Learning – From Acoustic Simulations to Recorded Data. In 2020 IEEE 22nd International Workshop on Multimedia Signal Processing (MMSP), pages 1–6, September 2020. 7, 108, 138
- [66] WEIPENG HE, PETR MOTLICEK, AND JEAN-MARC ODOBEZ. Neural Network Adaptation and Data Augmentation for Multi-Speaker Direction-of-Arrival Estimation. IEEE/ACM Transactions on Audio, Speech, and Language Processing, 29:1303–1317, 2021. 7, 108
- [67] MICHAEL M. BRONSTEIN, JOAN BRUNA, TACO COHEN, AND PETAR VELIČKOVIĆ. Geometric Deep Learning: Grids, Groups, Graphs, Geodesics, and Gauges, May 2021. 7
- [68] Y. LECUN, B. BOSER, J. S. DENKER, D. HENDERSON, R. E. HOWARD, W. HUBBARD, AND L. D. JACKEL. Backpropagation Applied to Handwritten Zip Code Recognition. *Neural Computation*, 1(4):541–551, December 1989. 8
- [69] ALEX KRIZHEVSKY, ILYA SUTSKEVER, AND GEOFFREY E HINTON. ImageNet Classification with Deep Convolutional Neural Networks. In F. PEREIRA, C.J. BURGES, L. BOTTOU, AND K.Q. WEINBERGER, editors, Advances in Neural Information Processing Systems, 25. Curran Associates, Inc., 2012. 8
- [70] KAIMING HE, XIANGYU ZHANG, SHAOQING REN, AND JIAN SUN. Deep Residual Learning for Image Recognition. In 2016 IEEE Conference on Computer Vision and Pattern Recognition (CVPR), pages 770–778, Las Vegas, NV, USA, June 2016. IEEE. 8
- [71] OLAF RONNEBERGER, PHILIPP FISCHER, AND THOMAS BROX. U-Net: Convolutional Networks for Biomedical Image Segmentation. In NASSIR NAVAB, JOACHIM HORNEGGER, WILLIAM M. WELLS, AND ALEJANDRO F. FRANGI, editors, Medical Image Computing and Computer-Assisted Intervention – MICCAI 2015, Lecture Notes in Computer Science, pages 234–241, Cham, 2015. Springer International Publishing. 8, 115
- [72] JONATHAN LONG, EVAN SHELHAMER, AND TREVOR DARRELL. Fully Convolutional Networks for Semantic Segmentation. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, pages 3431–3440, 2015. 8
- [73] TACO COHEN AND MAX WELLING. Group Equivariant Convolutional Networks. In Proceedings of The 33rd International Conference on Machine Learning, pages 2990–2999. PMLR, June 2016. 8
- [74] EMIEL HOOGEBOOM, JORN WT PETERS, TACO S. COHEN, AND MAX WELLING. Hexaconv. arXiv preprint arXiv:1803.02108, 2018. 8
- [75] DANIEL WORRALL AND GABRIEL BROSTOW. Cubenet: Equivariance to 3d Rotation and Translation. In Proceedings of the European Conference on Computer Vision (ECCV), pages 567–584, 2018. 8
- [76] TACO COHEN, MAURICE WEILER, BERKAY KICANAOGLU, AND MAX WELLING. Gauge Equivariant Convolutional Networks and the Icosahedral CNN. In Proceedings of the 36th International Conference on Machine Learning, pages 1321–1330. PMLR, May 2019. 8, 12, 93, 94, 95, 104
- [77] TACO S. COHEN, MARIO GEIGER, JONAS KÖHLER, AND MAX WELLING. Spherical CNNs. In International Conference on Learning Representations, February 2018. 8, 93, 138
- [78] THOMAS N. KIPF AND MAX WELLING. Semi-Supervised Classification with Graph Convolutional Networks. In 5th International Conference on Learning Representations, ICLR 2017, Toulon, France, April 24-26, 2017, Conference Track Proceedings. OpenReview.net, 2017. 8
- [79] MICHAËL DEFFERRARD, XAVIER BRESSON, AND PIERRE VANDERGHEYNST. Convolutional Neural Networks on Graphs with Fast Localized Spectral Filtering. In D. LEE, M. SUGIYAMA, U. LUXBURG, I. GUYON, AND R. GARNETT, editors, Advances in Neural Information Processing Systems, 29. Curran Associates, Inc., 2016. 8

- [80] FEDERICO MONTI, DAVIDE BOSCAINI, JONATHAN MASCI, EMANUELE RODOLA, JAN SVOBODA, AND MICHAEL M. BRONSTEIN. Geometric Deep Learning on Graphs and Manifolds Using Mixture Model Cnns. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, pages 5115–5124, 2017. 8
- [81] PETAR VELIČKOVIĆ, GUILLEM CUCURULL, ARANTXA CASANOVA, ADRIANA ROMERO, PIETRO LIÒ, AND YOSHUA BENGIO. Graph Attention Networks. In International Conference on Learning Representations, 2018. 8
- [82] JUSTIN GILMER, SAMUEL S. SCHOENHOLZ, PATRICK F. RILEY, ORIOL VINYALS, AND GEORGE E. DAHL. Neural Message Passing for Quantum Chemistry. In International Conference on Machine Learning, pages 1263–1272. PMLR, 2017. 8
- [83] MANZIL ZAHEER, SATWIK KOTTUR, SIAMAK RAVANBAKHSH, BARNABAS POCZOS, RUSS R SALAKHUTDINOV, AND ALEXANDER J SMOLA. Deep Sets. In I. GUYON, U. VON LUXBURG, S. BENGIO, H. WALLACH, R. FERGUS, S. VISHWANATHAN, AND R. GARNETT, editors, Advances in Neural Information Processing Systems, 30. Curran Associates, Inc., 2017. 8
- [84] ASHISH VASWANI, NOAM SHAZEER, NIKI PARMAR, JAKOB USZKOREIT, LLION JONES, AIDAN N GOMEZ, ŁUKASZ KAISER, AND ILLIA POLOSUKHIN. Attention Is All You Need. In Advances in Neural Information Processing Systems, 30. Curran Associates, Inc., 2017. 8, 118, 119
- [85] VICTOR GARCIA SATORRAS, EMIEL HOOGEBOOM, AND MAX WELLING. E (n) Equivariant Graph Neural Networks. In International Conference on Machine Learning, pages 9323– 9332. PMLR, 2021. 8
- [86] DANILO COMMINIELLO, MARCO LELLA, SIMONE SCARDAPANE, AND AURELIO UNCINI. Quaternion Convolutional Neural Networks for Detection and Localization of 3D Sound Events. In ICASSP 2019 - 2019 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 8533–8537, May 2019. 10
- [87] MICHELA RICCIARDI CELSI, SIMONE SCARDAPANE, AND DANILO COMMINIELLO. Quaternion Neural Networks for 3D Sound Source Localization in Reverberant Environments. In 2020 IEEE 30th International Workshop on Machine Learning for Signal Processing (MLSP), pages 1–6, September 2020. 10
- [88] BOAZ RAFAELY. Fundamentals of Spherical Array Processing, 8 of Springer Topics in Signal Processing. Springer Berlin Heidelberg, Berlin, Heidelberg, 2015. 16
- [89] DANIEL P. JARRETT, EMANUËL A.P. HABETS, AND PATRICK A. NAYLOR. Theory and Applications of Spherical Microphone Array Processing, 9 of Springer Topics in Signal Processing. Springer International Publishing, Cham, 2017. 16
- [90] C. KNAPP AND G. CARTER. The Generalized Correlation Method for Estimation of Time Delay. IEEE Transactions on Acoustics, Speech, and Signal Processing, 24(4):320–327, August 1976. 16
- [91] H. WANG AND P. CHU. Voice Source Localization for Automatic Camera Pointing System in Videoconferencing. In 1997 IEEE International Conference on Acoustics, Speech, and Signal Processing, 1, pages 187–190 vol.1, April 1997. 16

- [92] M. OMOLOGO AND P. SVAIZER. Use of the Crosspower-Spectrum Phase in Acoustic Event Location. IEEE Transactions on Speech and Audio Processing, 5(3):288-292, May 1997. 16
- [93] C. M. ZANNINI, A. CIRILLO, R. PARISI, AND A. UNCINI. Improved TDOA Disambiguation Techniques for Sound Source Localization in Reverberant Environments. In Proceedings of 2010 IEEE International Symposium on Circuits and Systems, pages 2666–2669, May 2010. 16
- [94] KEVIN D. DONOHUE, JENS HANNEMANN, AND HENRY G. DIETZ. Performance of Phase Transform for Detecting Sound Sources with Microphone Arrays in Reverberant and Noisy Environments. Signal Processing, 87(7):1677–1691, 2007. 16
- [95] B. CHAMPAGNE, S. BEDARD, AND A. STEPHENNE. Performance of Time-Delay Estimation in the Presence of Room Reverberation. IEEE Transactions on Speech and Audio Processing, 4(2):148–152, March 1996. 16
- [96] H. DO AND H. F. SILVERMAN. A Fast Microphone Array SRP-PHAT Source Location Implementation Using Coarse-To-Fine Region Contraction(CFRC). In 2007 IEEE Workshop on Applications of Signal Processing to Audio and Acoustics, pages 295–298, October 2007. 16
- [97] H. DO AND H. F. SILVERMAN. Stochastic Particle Filtering: A Fast SRP-PHAT Single Source Localization Algorithm. In 2009 IEEE Workshop on Applications of Signal Processing to Audio and Acoustics, pages 213–216, October 2009. 16, 19
- [98] L. O. NUNES, W. A. MARTINS, M. V. S. LIMA, L. W. P. BISCAINHO, M. V. M. COSTA, F. M. GONÇALVES, A. SAID, AND B. LEE. A Steered-Response Power Algorithm Employing Hierarchical Search for Acoustic Source Localization Using Microphone Arrays. *IEEE Transactions on Signal Processing*, 62(19):5171–5183, October 2014. 16, 19, 20
- [99] AMPARO MARTI, MAXIMO COBOS, JOSE J. LOPEZ, AND JOSE ESCOLANO. A Steered Response Power Iterative Method for High-Accuracy Acoustic Source Localization. The Journal of the Acoustical Society of America, 134(4):2627–2630, October 2013. 16
- [100] M. V. S. LIMA, W. A. MARTINS, L. O. NUNES, L. W. P. BISCAINHO, T. N. FERREIRA, M. V. M. COSTA, AND B. LEE. A Volumetric SRP with Refinement Step for Sound Source Localization. *IEEE Signal Processing Letters*, 22(8):1098–1102, August 2015. 16
- [101] S. ZHAO, S. AHMED, Y. LIANG, K. RUPNOW, D. CHEN, AND D. L. JONES. A Real-Time 3D Sound Localization System with Miniature Microphone Array for Virtual Reality. In 2012 7th IEEE Conference on Industrial Electronics and Applications (ICIEA), pages 1853– 1857, July 2012. 16
- [102] TAEWOO LEE, SUKMOON CHANG, AND DONGSUK YOOK. Parallel SRP-PHAT for GPUs. Computer Speech & Language, 35(Supplement C):1–13, January 2016. 16
- [103] H. R. ZARGHI, M. SHARIFKHANI, AND I. GHOLAMPOUR. Implementation of a Cost Efficient SSL Based on an Angular Beamformer SRP-PHAT. In 2011 18th IEEE International Conference on Electronics, Circuits, and Systems, pages 49–52, December 2011. 16

- [104] R. ROY, A. PAULRAJ, AND T. KAILATH. Estimation of Signal Parameters via Rotational Invariance Techniques - ESPRIT. In IEEE Military Communications Conference - Communications-Computers: Teamed for the 90's, 1986. MILCOM 1986, 3, pages 41.6.1– 41.6.5, October 1986. 16
- [105] M. A. DORAN, E. DORON, AND A. J. WEISS. Coherent Wide-Band Processing for Arbitrary Array Geometry. IEEE Transactions on Signal Processing, 41(1):414-, January 1993. 17
- [106] A. JOHANSSON, G. COOK, AND S. NORDHOLM. Acoustic Direction of Arrival Estimation, a Comparison between Root-Music and SRP-PHAT. In TENCON 2004. 2004 IEEE Region 10 Conference, B, pages 629–632, November 2004. 17
- [107] J. P. DMOCHOWSKI, J. BENESTY, AND S. AFFES. Broadband Music: Opportunities and Challenges for Multiple Source Localization. In 2007 IEEE Workshop on Applications of Signal Processing to Audio and Acoustics, pages 18–21, October 2007. 17
- [108] J. KROLIK AND D. SWINGLER. Multiple Broad-Band Source Location Using Steered Covariance Matrices. IEEE Transactions on Acoustics, Speech, and Signal Processing, 37(10):1481–1494, October 1989. 17
- [109] M. SOUDEN, J. BENESTY, AND S. AFFES. Broadband Source Localization From an Eigenanalysis Perspective. IEEE Transactions on Audio, Speech, and Language Processing, 18(6):1575–1587, August 2010. 17
- [110] SATISH MOHAN, MICHAEL E. LOCKWOOD, MICHAEL L. KRAMER, AND DOUGLAS L. JONES. Localization of Multiple Acoustic Sources with Small Arrays Using a Coherence Test. The Journal of the Acoustical Society of America, 123(4):2136–2147, April 2008. 17
- [111] M. REN AND Y. X. ZOU. A Novel Multiple Sparse Source Localization Using Triangular Pyramid Microphone Array. *IEEE Signal Processing Letters*, 19(2):83–86, February 2012. 17
- [112] D. PAVLIDI, A. GRIFFIN, M. PUIGT, AND A. MOUCHTARIS. Real-Time Multiple Sound Source Localization and Counting Using a Circular Microphone Array. *IEEE Trans*actions on Audio, Speech, and Language Processing, 21(10):2193–2206, October 2013. 17
- [113] O. NADIRI AND B. RAFAELY. Localization of Multiple Speakers under High Reverberation Using a Spherical Microphone Array and the Direct-Path Dominance Test. *IEEE/ACM Transactions on Audio, Speech, and Language Processing*, 22(10):1494–1505, October 2014. 17
- [114] H. PESSENTHEINER, M. HAGMÜLLER, AND G. KUBIN. Localization and Characterization of Multiple Harmonic Sources. IEEE/ACM Transactions on Audio, Speech, and Language Processing, 24(8):1348–1363, August 2016. 17
- [115] MAOSHEN JIA, JUNDAI SUN, CHANGCHUN BAO, AND CHRISTIAN RITZ. Multiple-to-Single Sound Source Localization by Applying Single-Source Bins Detection. Applied Acoustics, 138:28–38, September 2018. 17
- [116] S. RICKARD AND O. YILMAZ. On the Approximate W-disjoint Orthogonality of Speech. In 2002 IEEE International Conference on Acoustics, Speech, and Signal Processing, 1, pages I-529-I-532, May 2002. 17

- [117] J. TRAA AND P. SMARAGDIS. A Wrapped Kalman Filter for Azimuthal Speaker Tracking. *IEEE Signal Processing Letters*, 20(12):1257–1260, December 2013. 17
- [118] YE TIAN, ZHE CHEN, AND FULIANG YIN. Distributed Kalman Filter-Based Speaker Tracking in Microphone Array Networks. Applied Acoustics, 89:71–77, March 2015. 17
- [119] D. B. WARD, E. A. LEHMANN, AND R. C. WILLIAMSON. Particle Filtering Algorithms for Tracking an Acoustic Source in a Reverberant Environment. *IEEE Transactions on Speech and Audio Processing*, 11(6):826–836, November 2003. 17
- [120] WING-KIN MA, BA-NGU VO, S. S. SINGH, AND A. BADDELEY. Tracking an Unknown Time-Varying Number of Speakers Using TDOA Measurements: A Random Finite Set Approach. IEEE Transactions on Signal Processing, 54(9):3291–3304, September 2006. 17
- [121] C. EVERS, Y. DORFAN, S. GANNOT, AND P. A. NAYLOR. Source Tracking Using Moving Microphone Arrays for Robot Audition. In 2017 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 6145–6149, March 2017. 18
- [122] O. SCHWARTZ, Y. DORFAN, E. A. P. HABETS, AND S. GANNOT. Multi-Speaker DOA Estimation in Reverberation Conditions Using Expectation-Maximization. In 2016 IEEE International Workshop on Acoustic Signal Enhancement (IWAENC), pages 1–5, September 2016. 18
- [123] H. DO, H. F. SILVERMAN, AND Y. YU. A Real-Time SRP-PHAT Source Location Implementation Using Stochastic Region Contraction(SRC) on a Large-Aperture Microphone Array. In 2007 IEEE International Conference on Acoustics, Speech and Signal Processing - ICASSP '07, 1, pages I-121-I-124, April 2007. 19
- [124] MAXIMO COBOS, AMPARO MARTI, AND JOSE J. LOPEZ. A Modified SRP-PHAT Functional for Robust Real-Time Sound Source Localization With Scalable Spatial Sampling. *IEEE Signal Processing Letters*, 18(1):71–74, January 2011. 20
- [125] I. J. CLARKE. Efficient Maximum Likelihood Using Higher Rank Spectral Estimation. In Workshop on Higher-Order Spectral Analysis, 1989, pages 229–234, June 1989. 22, 23
- [126] M. A. ZATMAN, H. J. STRANGEWAYS, AND E. M. WARRINGTON. Resolution of Multimoded HF Transmissions Using the DOSE Superresolution Direction Finding Algorithm. In, Eighth International Conference on Antennas and Propagation, 1993, pages 415–417 vol.1, 1993. 22, 23
- [127] A. MORRISON, B. S. SHARIF, S. SALI, AND O. R. HINTON. An Iterative DOA Algorithm for a Space-Time DS-CDMA RAKE Receiver. In 3G Mobile Communication Technologies, 2000. First International Conference on (Conf. Publ. No. 471), pages 208–212, 2000. 22, 23
- [128] M. A. GUÉRARD AND D. GRENIER. Direction of Arrival Estimation of Acoustic Echoes Using Source Elimination Method. Canadian Journal of Electrical and Computer Engineering, 40(3):246–252, 2017. 22
- [129] ALESSIO BRUTTI, MAURIZIO OMOLOGO, AND PIERGIORGIO SVAIZER. Multiple Source Localization Based on Acoustic Map De-Emphasis. EURASIP Journal on Audio, Speech, and Music Processing, 2010(1):147495, December 2010. 22, 34, 35, 36, 40, 45, 46

- [130] ANGELO FARINA AND LAMBERTO TRONCHIN. 3D Sound Characterisation in Theatres Employing Microphone Arrays. Acta Acustica united with Acustica, 99(1):118–125, January 2013. 25
- [131] A. O'DONOVAN, R. DURAISWAMI, AND D. ZOTKIN. Imaging Concert Hall Acoustics Using Visual and Audio Cameras. In 2008 IEEE International Conference on Acoustics, Speech and Signal Processing, pages 5284–5287, March 2008. 25
- [132] CHA ZHANG, D. FLORENCIO, AND ZHENGYOU ZHANG. Why Does PHAT Work Well in Lownoise, Reverberative Environments? In 2008 IEEE International Conference on Acoustics, Speech and Signal Processing, pages 2565–2568, March 2008. 26, 37
- [133] E A LEHMANN AND A M JOHANSSON. Diffuse Reverberation Model for Efficient Image-Source Simulation of Room Impulse Responses. IEEE Transactions on Audio, Speech, and Language Processing, 18(6):1429–1439, August 2010. 29, 56, 60, 69
- [134] YOSHUA BENGIO, JÉRÔME LOURADOUR, RONAN COLLOBERT, AND JASON WESTON. Curriculum Learning. In Proceedings of the 26th Annual International Conference on Machine Learning, ICML '09, pages 41–48, Montreal, Quebec, Canada, June 2009. Association for Computing Machinery. 49, 84, 101
- [135] VASSIL PANAYOTOV, GUOGUO CHEN, DANIEL POVEY, AND SANJEEV KHUDANPUR. Librispeech: An ASR Corpus Based on Public Domain Audio Books. In 2015 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 5206–5210, April 2015. 51, 101
- [136] JOHN WISEMAN. Wiseman/Py-Webrtcvad. https://github.com/wiseman/py-webrtcvad, November 2019. 51, 101
- [137] H. W. LÖLLMANN, C. EVERS, A. SCHMIDT, H. MELLMANN, H. BARFUSS, P. A. NAYLOR, AND W. KELLERMANN. The LOCATA Challenge Data Corpus for Acoustic Source Localization and Tracking. In 2018 IEEE 10th Sensor Array and Multichannel Signal Processing Workshop (SAM), pages 410–414, July 2018. 53, 81, 90, 91, 101, 106
- [138] HEINRICH. W. LÖLLMANN, CHRISTINE EVERS, ALEXANDER SCHMIDT, HEINRICH MELLMANN, HENDRIK BARFUSS, PATRICK A. NAYLOR, AND WALTER KELLERMANN. IEEE-AASP Challenge on Acoustic Source Localization and Tracking: Documentation of Final Release. https://locata.lms.tf.fau.de/datasets/, January 2020. 53, 101, 137
- [139] JASON L. WILLIAMS. Marginal Multi-Bernoulli Filters: RFS Derivation of MHT, JIPDA, and Association-Based Member. IEEE Transactions on Aerospace and Electronic Systems, 51(3):1664–1687, July 2015. 53
- [140] KARL GRANSTRÖM, LENNART SVENSSON, YUXUAN XIA, JASON WILLIAMS, AND ÁNGEL F. GARCÍA-FEMÁNDEZ. Poisson Multi-Bernoulli Mixture Trackers: Continuity Through Random Finite Sets of Trajectories. In 2018 21st International Conference on Information Fusion (FUSION), pages 1–5, July 2018. 53
- [141] CHAO WENG, DONG YU, SHINJI WATANABE, AND BIING-HWANG FRED JUANG. Recurrent Deep Neural Networks for Robust Speech Recognition. In 2014 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 5532–5536, May 2014. 55

- [142] ANTHONY GRIFFIN, ANASTASIOS ALEXANDRIDIS, DESPOINA PAVLIDI, YIANNIS MASTORAKIS, AND ATHANASIOS MOUCHTARIS. Localizing Multiple Audio Sources in a Wireless Acoustic Sensor Network. Signal Processing, 107:54–67, February 2015. 55, 69
- [143] D. S. WILLIAMSON AND D. WANG. Time-Frequency Masking in the Complex Domain for Speech Dereverberation and Denoising. *IEEE/ACM Transactions on Audio, Speech,* and Language Processing, 25(7):1492–1501, July 2017. 55, 69
- [144] MANFRED R. SCHROEDER. Natural Sounding Artificial Reverberation. Journal of the Audio Engineering Society, 10(3):219–223, July 1962. 55
- [145] ZHONG-HUA FU AND JIAN-WEI LI. GPU-based Image Method for Room Impulse Response Calculation. Multimedia Tools and Applications, 75(9):5205-5221, May 2016. 56
- [146] EMANUËL A.P. HABETS. Room Impulse Response Generator. Technical report, International Audio Laboratories Erlangen, September 2010. 56, 69
- [147] ROBIN SCHEIBLER, ERIC BEZZAM, AND IVAN DOKMANIĆ. Pyroomacoustics: A Python Package for Audio Room Simulation and Array Processing Algorithms. In 2018 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 351–355, April 2018. 56, 69
- [148] PATRICK M. PETERSON. Simulating the Response of Multiple Microphones to a Single Acoustic Source in a Reverberant Room. The Journal of the Acoustical Society of America, 80(5):1527-1529, November 1986. 59
- [149] B. D. RADLOVIC, R. C. WILLIAMSON, AND R. A. KENNEDY. Equalization in an Acoustic Reverberant Environment: Robustness Results. *IEEE Transactions on Speech and Audio Processing*, 8(3):311–319, May 2000. 59
- [150] J ANTONIO, L GODINHO, AND A TADEU. Reverberation Times Obtained Using a Numerical Model Versus Those Given by Simplified Formulas and Measurements. ACTA ACUSTICA UNITED WITH ACUSTICA, 88:10, 2002. 59
- [151] ERIC A. LEHMANN AND ANDERS M. JOHANSSON. Prediction of Energy Decay in Room Impulse Responses Simulated with an Image-Source Model. The Journal of the Acoustical Society of America, 124(1):269–277, July 2008. 60, 69
- [152] WALLACE CLEMENT SABINE. Collected Papers on Acoustics. Cambridge : Harvard University Press, 1922. 60
- [153] JOHN NICKOLLS, IAN BUCK, MICHAEL GARLAND, AND KEVIN SKADRON. Scalable Parallel Programming with CUDA. In ACM SIGGRAPH 2008 Classes, SIGGRAPH '08, pages 16:1–16:14, New York, NY, USA, 2008. ACM. 61
- [154] TOR G. J. MYKLEBUST. Computing Accurate Horner Form Approximations to Special Functions in Finite Precision Arithmetic. arXiv:1508.03211 [cs, math], August 2015. 67
- [155] MARVIN182. Room Impulse Response Generator. https://github.com/Marvin182/rirgenerator, August 2018. 69

- [156] AMIN HASSANI, JORGE PLATA-CHAVES, MOHAMAD HASAN BAHARI, MARC MOONEN, AND ALEXANDER BERTRAND. Multi-Task Wireless Sensor Network for Joint Distributed Node-Specific Signal Enhancement, LCMV Beamforming and DOA Estimation. IEEE Journal of Selected Topics in Signal Processing, 11(3):518–533, April 2017. 69
- [157] SHMULIK MARKOVICH, SHARON GANNOT, AND ISRAEL COHEN. Multichannel Eigenspace Beamforming in a Reverberant Noisy Environment With Multiple Interfering Speech Signals. *IEEE Transactions on Audio, Speech, and Language Processing*, 17(6):1071– 1086, August 2009. 69
- [158] XIAOYI QIN, DANWEI CAI, AND MING LI. Far-Field End-to-End Text-Dependent Speaker Verification Based on Mixed Training Data with Transfer Learning and Enrollment Data Augmentation. In Interspeech 2019, pages 4045–4049. ISCA, September 2019. 69
- [159] LADISLAV MOSNER, MINHUA WU, ANIRUDH RAJU, SREE HARI KRISHNAN PARTHASARATHI, KENICHI KUMATANI, SHIVA SUNDARAM, ROLAND MAAS, AND BJORN HOFFMEISTER. Improving Noise Robustness of Automatic Speech Recognition via Parallel Data and Teacherstudent Learning. In ICASSP 2019 - 2019 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 6475–6479, Brighton, United Kingdom, May 2019. IEEE. 69
- [160] MARCO SEVERINI, DANIELE FERRETTI, EMANUELE PRINCIPI, AND STEFANO SQUARTINI. Automatic Detection of Cry Sounds in Neonatal Intensive Care Units by Using Deep Learning and Acoustic Scene Simulation. *IEEE Access*, 7:51982–51993, 2019. 69
- [161] ERIC A. LEHMANN. Fast Simulation of Acoustic Room Impulse Responses (Image-Source Method). https://www.mathworks.com/matlabcentral/fileexchange/25965fast-simulation-of-acoustic-room-impulse-responses-image-source-method, March 2012. 69
- [162] DESPOINA PAVLIDI, MATTHIEU PUIGT, ANTHONY GRIFFIN, AND ATHANASIOS MOUCHTARIS. Real-Time Multiple Sound Source Localization Using a Circular Microphone Array Based on Single-Source Confidence Measures. In 2012 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 2625–2628, March 2012. 69
- [163] ANASTASIOS ALEXANDRIDIS, ANTHONY GRIFFIN, AND ATHANASIOS MOUCHTARIS. Capturing and Reproducing Spatial Audio Based on a Circular Microphone Array. Journal of Electrical and Computer Engineering, 2013:e718574, March 2013. 69
- [164] KAIMING HE, XIANGYU ZHANG, SHAOQING REN, AND JIAN SUN. Delving Deep into Rectifiers: Surpassing Human-Level Performance on ImageNet Classification. In 2015 IEEE International Conference on Computer Vision (ICCV), pages 1026–1034, December 2015. 82
- [165] DIEDERIK P. KINGMA AND JIMMY BA. Adam: A Method for Stochastic Optimization. In 3rd International Conference for Learning Representations (ICLR), San Diego, 2015. 84, 101
- [166] ADAM PASZKE, SAM GROSS, FRANCISCO MASSA, ADAM LERER, JAMES BRADBURY, GREGORY CHANAN, TREVOR KILLEEN, ZEMING LIN, NATALIA GIMELSHEIN, LUCA ANTIGA, ALBAN DESMAI-SON, ANDREAS KOPF, EDWARD YANG, ZACHARY DEVITO, MARTIN RAISON, ALYKHAN TEJANI, SASANK CHILAMKURTHY, BENOIT STEINER, LU FANG, JUNJIE BAI, AND SOUMITH CHINTALA. Py-Torch: An Imperative Style, High-Performance Deep Learning Library. In Advances in Neural Information Processing Systems 32, pages 8026–8037, 2019. 84, 101

- [167] KYUNGHYUN CHO, BART VAN MERRIËNBOER, CAGLAR GULCEHRE, DZMITRY BAHDANAU, FETHI BOUGARES, HOLGER SCHWENK, AND YOSHUA BENGIO. Learning Phrase Representations Using RNN Encoder-Decoder for Statistical Machine Translation. In Proceedings of the 2014 Conference on Empirical Methods in Natural Language Processing (EMNLP), pages 1724-1734, Doha, Qatar, October 2014. Association for Computational Linguistics. 86
- [168] CARLOS ESTEVES, CHRISTINE ALLEN-BLANCHETTE, AMEESH MAKADIA, AND KOSTAS DANIILIDIS. Learning SO(3) Equivariant Representations with Spherical CNNs. In Proceedings of the European Conference on Computer Vision (ECCV), pages 52–68, 2018. 93, 138
- [169] OLIVER COBB, CHRISTOPHER G. R. WALLIS, AUGUSTINE N. MAVOR-PARKER, AUGUSTIN MARIG-NIER, MATTHEW A. PRICE, MAYEUL D'AVEZAC, AND JASON MCEWEN. Efficient Generalized Spherical CNNs. In International Conference on Learning Representations, September 2020. 93, 138
- [170] JASON MCEWEN, CHRISTOPHER WALLIS, AND AUGUSTINE N. MAVOR-PARKER. Scattering Networks on the Sphere for Scalable and Rotationally Equivariant Spherical CNNs. In International Conference on Learning Representations, September 2021. 93, 138
- [171] RENATA KHASANOVA AND PASCAL FROSSARD. Graph-Based Classification of Omnidirectional Images. In Proceedings of the IEEE International Conference on Computer Vision Workshops, pages 869–878, 2017. 93, 139
- [172] NATHANAËL PERRAUDIN, MICHAËL DEFFERRARD, TOMASZ KACPRZAK, AND RAPHAEL SGIER. DeepSphere: Efficient Spherical Convolutional Neural Network with HEALPix Sampling for Cosmological Applications. Astronomy and Computing, 27:130–146, April 2019. 93, 139
- [173] MICHAËL DEFFERRARD, MARTINO MILANI, FRÉDÉRICK GUSSET, AND NATHANAËL PERRAUDIN. DeepSphere: A Graph-Based Spherical CNN. In International Conference on Learning Representations, September 2019. 93, 139
- [174] JIMMY LEI BA, JAMIE RYAN KIROS, AND GEOFFREY E. HINTON. Layer Normalization. In NIPS 2016 Deep Learning Symposium, Barcelona, July 2016. 99
- [175] SEPP HOCHREITER AND JÜRGEN SCHMIDHUBER. Long Short-Term Memory. Neural Computation, 9(8):1735–1780, November 1997. 117
- [176] KYUNGHYUN CHO, BART VAN MERRIENBOER, DZMITRY BAHDANAU, AND YOSHUA BENGIO. On the Properties of Neural Machine Translation: Encoder–Decoder Approaches. In SSST@EMNLP, 2014. 117
- [177] GUO-BING ZHOU, JIANXIN WU, CHEN-LIN ZHANG, AND ZHI-HUA ZHOU. Minimal Gated Unit for Recurrent Neural Networks. International Journal of Automation and Computing, 13(3):226–234, June 2016. 119, 139
- [178] TIM MEINHARDT, ALEXANDER KIRILLOV, LAURA LEAL-TAIXE, AND CHRISTOPH FEICHTENHOFER. TrackFormer: Multi-Object Tracking with Transformers. In 2022 IEEE/CVF Conference on Computer Vision and Pattern Recognition (CVPR), pages 8834–8844, New Orleans, LA, USA, June 2022. IEEE. 120, 139, 141

- [179] NICOLAS CARION, FRANCISCO MASSA, GABRIEL SYNNAEVE, NICOLAS USUNIER, ALEXANDER KIRILLOV, AND SERGEY ZAGORUYKO. End-to-End Object Detection with Transformers. In ANDREA VEDALDI, HORST BISCHOF, THOMAS BROX, AND JAN-MICHAEL FRAHM, editors, *Computer Vision – ECCV 2020*, Lecture Notes in Computer Science, pages 213–229, Cham, 2020. Springer International Publishing. 120
- [180] XIZHOU ZHU, WEIJIE SU, LEWEI LU, BIN LI, XIAOGANG WANG, AND JIFENG DAI. Deformable DETR: Deformable Transformers for End-to-End Object Detection. In International Conference on Learning Representations, February 2022. 120
- [181] YIHONG XU, ALJOSA OSEP, YUTONG BAN, RADU HORAUD, LAURA LEAL-TAIXÉ, AND XAVIER ALAMEDA-PINEDA. How to Train Your Deep Multi-Object Tracker. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, pages 6787–6796, 2020. 121
- [182] DONG YU, MORTEN KOLBÆK, ZHENG-HUA TAN, AND JESPER JENSEN. Permutation Invariant Training of Deep Models for Speaker-Independent Multi-Talker Speech Separation. In 2017 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 241–245, March 2017. 121
- [183] HAROLD W. KUHN. The Hungarian Method for the Assignment Problem. Naval research logistics quarterly, 2(1-2):83-97, 1955. 122
- [184] MORTEN KOLBÆK, DONG YU, ZHENG-HUA TAN, AND JESPER JENSEN. Multitalker Speech Separation With Utterance-Level Permutation Invariant Training of Deep Recurrent Neural Networks. *IEEE/ACM Transactions on Audio, Speech, and Language Processing*, 25(10):1901–1913, October 2017. 123
- [185] ILYA LOSHCHILOV AND FRANK HUTTER. Decoupled Weight Decay Regularization. In International Conference on Learning Representations, 2018. 126
- [186] RAZVAN PASCANU, TOMAS MIKOLOV, AND YOSHUA BENGIO. On the Difficulty of Training Recurrent Neural Networks. In Proceedings of the 30th International Conference on Machine Learning, pages 1310–1318. PMLR, May 2013. 126
- [187] KENI BERNARDIN AND RAINER STIEFELHAGEN. Evaluating Multiple Object Tracking Performance: The CLEAR MOT Metrics. EURASIP Journal on Image and Video Processing, 2008(1):1–10, December 2008. 126
- [188] CHRISTINE EVERS, HEINRICH W. LÖLLMANN, HEINRICH MELLMANN, ALEXANDER SCHMIDT, HEN-DRIK BARFUSS, PATRICK A. NAYLOR, AND WALTER KELLERMANN. The LOCATA Challenge: Acoustic Source Localization and Tracking. IEEE/ACM Transactions on Audio, Speech, and Language Processing, 28:1620–1643, 2020. 126
- [189] LUCA CRISTOFORETTI, MIRCO RAVANELLI, MAURIZIO OMOLOGO, ALESSANDRO SOSI, ALBERTO ABAD, MARTIN HAGMUELLER, AND PETROS MARAGOS. The DIRHA Simulated Corpus. In Proceedings of the Ninth International Conference on Language Resources and Evaluation (LREC'14), pages 2629–2634, Reykjavik, Iceland, May 2014. European Language Resources Association (ELRA). 138
- [190] ELIOR HADAD, FLORIAN HEESE, PETER VARY, AND SHARON GANNOT. Multichannel Audio Database in Various Acoustic Environments. In 2014 14th International Workshop on Acoustic Signal Enhancement (IWAENC), pages 313–317, September 2014. 138

- [191] IGOR SZÖKE, MIROSLAV SKÁCEL, LADISLAV MOŠNER, JAKUB PALIESEK, AND JAN ČERNOCKÝ. Building and Evaluation of a Real Room Impulse Response Dataset. IEEE Journal of Selected Topics in Signal Processing, 13(4):863–876, August 2019. 138
- [192] EFREN FERNANDEZ-GRANDE. DTU Three-Channel Room Impulse Response Dataset for Direction of Arrival Estimation 2020, June 2021. 138
- [193] PRERAK SRIVASTAVA, ANTOINE DELEFORGE, ARCHONTIS POLITIS, AND EMMANUEL VINCENT. How to (Virtually) Train Your Sound Source Localizer. HAL Archives Ouvertes, November 2022. 138
- [194] FEMKE B. GELDERBLOM, YI LIU, JOHANNES KVAM, AND TOR ANDRE MYRVOLL. Synthetic Data For Dnn-Based Doa Estimation of Indoor Speech. In ICASSP 2021 - 2021 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 4390– 4394, June 2021. 138
- [195] KADI BOUATOUCH, OLIVIER DEILLE, JULIEN MAILLARD, JACQUES MARTIN, AND NICOLAS NOÉ. Real Time Acoustic Rendering of Complex Environments Including Diffraction and Curved Surfaces. In Audio Engineering Society Convention 120. Audio Engineering Society, May 2006. 138
- [196] RYU TAKEDA AND KAZUNORI KOMATANI. Unsupervised Adaptation of Deep Neural Networks for Sound Source Localization Using Entropy Minimization. In 2017 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), pages 2217– 2221, March 2017. 138
- [197] RYOTARO SATO, KENTA NIWA, AND KAZUNORI KOBAYASHI. Ambisonic Signal Processing DNNs Guaranteeing Rotation, Scale and Time Translation Equivariance. IEEE/ACM Transactions on Audio, Speech, and Language Processing, 29:1449–1462, 2021. 139

## Appendix A

# Network architecture of the proposed and baseline models

The following tables, some of them extracted from the supplementary material of [V], detail the network architecture of the proposed models and baseline models analyzed along the thesis:

#### A.1 Cross3D models

**Table A.1:** Architecture of Cross3D for 4x8 power maps. The output sizes correspond to a sequence of 103 maps, i.e. an input size of 3x103x4x8.

	Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
Input layer	3D causal conv PReLU	32	5x5x5	1x1x1	$\begin{array}{c} 12032\\ 32 \end{array}$	32x103x4x8 32x103x4x8
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x4x8 32x103x4x4 32x103x4x4
$\theta$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x4x4 32x103x4x2 32x103x4x2
	Transpose Reshape					103x32x4x2 103x256
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x4x8 32x103x2x8 32x103x2x8
$\varphi$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x2x8 32x103x1x8 32x103x1x8
	Transpose Reshape					103x32x1x8 103x256
	Concatenate Transpose					103x512 512x103
Output layers	1D causal conv PReLU	128	5	2	$\begin{array}{c} 327808\\ 128 \end{array}$	128x103 128x103
	1D causal conv tanh	3	5	2	1923	$3x103 \\ 3x103$

	Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
Input layer	3D causal conv PReLU	32	5x5x5	1x1x1	$\begin{array}{c} 12032\\ 32 \end{array}$	32x103x8x16 32x103x8x16
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x8x16 32x103x8x8 32x103x8x8
$\theta$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x8x8 32x103x8x4 32x103x8x4
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x8x4 32x103x8x2 32x103x8x2
	Transpose Reshape					103x32x8x2 103x512
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x8x16 32x103x4x16 32x103x4x16
$\varphi$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x4x16 32x103x2x16 32x103x2x16
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x2x16 32x103x1x16 32x103x1x16
	Transpose Reshape					103x32x1x16 103x512
	Concatenate Transpose					103x1024 1024x103
Output layers	1D causal conv PReLU	128	5	2	655488 128	128x103 128x103
	1D causal conv tanh	3	5	2	1923	3x103 3x103

**Table A.2:** Architecture of Cross3D for 8x16 power maps. The output sizes correspond to a sequence of 103 maps, i.e. an input size of 3x103x8x16.

	Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
Input layer	3D causal conv PReLU	32	5x5x5	1x1x1	$\begin{array}{c} 12032\\ 32 \end{array}$	32x103x16x32 32x103x16x32
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x16x32 32x103x16x16 32x103x16x16
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x16x16 32x103x16x8 32x103x16x8
$\theta$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x16x8 32x103x16x4 32x103x16x4
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x16x4 32x103x16x2 32x103x16x2
	Transpose Reshape					103x32x16x2 103x1024
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x16x32 32x103x8x32 32x103x8x32
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x8x32 32x103x4x32 32x103x4x32
$\varphi$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x4x32 32x103x2x32 32x103x2x32
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x2x32 32x103x1x32 32x103x1x32
	Transpose Reshape					103x32x1x32 103x1024
	Concatenate Transpose					103x2048 2048x103
Output layers	1D causal conv PReLU	128	5	2	1 310 848 128	128x103 128x103
	1D causal conv tanh	3	5	2	1923	$3x103 \\ 3x103$

**Table A.3:** Architecture of Cross3D for 16x32 power maps. The output sizes correspond to a sequence of 103 maps, i.e. an input size of 3x103x16x32.

	Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
Input layer	3D causal conv PReLU	32	5x5x5	1x1x1	$\begin{array}{c} 12032\\ 32 \end{array}$	32x103x32x64 32x103x32x64
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x32x64 32x103x32x32 32x103x32x32
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x32x32 32x103x32x16 32x103x32x16
$\theta$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x32x16 32x103x32x8 32x103x32x8
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x32x8 32x103x32x4 32x103x32x4
	Transpose Reshape					$\frac{103 x 32 x 32 x 4}{103 x 4096}$
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x32x64 32x103x16x64 32x103x16x64
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x16x64 32x103x8x64 32x103x8x64
$\varphi$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x8x64 32x103x4x64 32x103x4x64
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x4x64 32x103x2x64 32x103x2x64
	Transpose Reshape					103x32x2x64 103x4096
	Concatenate Transpose					103x8192 8192x103
Output layers	1D causal conv PReLU	128	5	2	5243008 128	128x103 128x103
	1D causal conv tanh	3	5	2	1923	3x103 3x103

Table A.4:	Architec	ture of Cro	ss3D for $32$	2x64 power	maps.	The output s	sizes correspond
to a sequence	e of 103 :	maps, i.e.	an input s	size of $3x103$	3x32x64	1.	

	Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
Input layer	3D causal conv PReLU	32	5x5x5	1x1x1	$\begin{array}{c} 12032\\ 32 \end{array}$	32x103x64x128 32x103x64x128
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x64x128 32x103x64x64 32x103x64x64
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x64x64 32x103x64x32 32x103x64x32
$\theta$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x64x32 32x103x64x16 32x103x64x16
	3D causal conv Max pooling PReLU	32	5x3x3 1x1x2	1x1x1	46 112 32	32x103x64x16 32x103x64x8 32x103x64x8
	Transpose Reshape					103x32x64x8 103x16384
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x64x128 32x103x32x128 32x103x32x128
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x32x128 32x103x16x128 32x103x16x128
$\varphi$ branch	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x16x128 32x103x8x128 32x103x8x128
	3D causal conv Max pooling PReLU	32	5x3x3 1x2x1	1x1x1	46 112 32	32x103x8x128 32x103x4x128 32x103x4x128
	Transpose Reshape					103x32x4x128 103x16384
	Concatenate Transpose					103x32768 32768x103
Output layers	1D causal conv PReLU	128	5	2	$20971648\\128$	128x103 128x103
	1D causal conv tanh	3	5	2	1923	3x103 3x103

**Table A.5:** Architecture of Cross3D for 64x128 power maps. The output sizes correspond to a sequence of 103 maps, i.e. an input size of 3x103x64x128.

#### A.2 icoCNN models

**Table A.6:** Architecture of the icoCNN model for r=1 power maps. The output sizes correspond to a sequence of 103 maps, i.e. an input size of 1x103x1x5x2x4.

Layer	Number of kernels	Kernel size	Number of parameters	Output size
icoConv	32	Hexagonal	256	32x103x6x5x2x4
ID causal conv	32	5	5152	32x103x6x5x2x4
Layer norm			64	32x103x6x5x2x4
ReLU				32x103x6x5x2x4
icoConv	32	Hexagonal	43040	32x103x6x5x2x4
1D causal conv	32	5	5152	32x103x6x5x2x4
Layer norm			64	32x103x6x5x2x4
ReLU				32x103x6x5x2x4
icoConv	32	Hexagonal	43040	32x103x6x5x2x4
1D causal conv	32	5	5152	32x103x6x5x2x4
Layer norm			64	32x103x6x5x2x4
ReLU				32x103x6x5x2x4
icoConv	32	Hexagonal	43040	32x103x6x5x2x4
1D causal conv	32	5	5152	32x103x6x5x2x4
Layer norm			64	32x103x6x5x2x4
ReLU				32x103x6x5x2x4
icoConv	32	Hexagonal	43040	32x103x6x5x2x4
1D causal conv	1	5	161	1x103x6x5x2x4
R-pooling				1x103x1x5x2x4
soft-max				103x3

Layer	Number of kernels	Kernel size	Number of parameters	Output size
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$256 \\ 5152 \\ 64$	32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040\\5152\\64$	32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8
icoPooling		Hexagonal		32x103x6x5x2x4
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040\\5152\\64$	32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040 \\ 5152 \\ 64$	32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040\\5152\\64$	32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040 \\ 5152 \\ 64$	32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4
icoConv 1D causal conv R-pooling soft-max	32 1	Hexagonal 5	$\begin{array}{c} 43040\\ 161 \end{array}$	32x103x6x5x2x4 1x103x6x5x2x4 1x103x1x5x2x4 103x3

**Table A.7:** Architecture of the icoCNN model for r=2 power maps. The output sizes correspond to a sequence of 103 maps, i.e. an input size of 1x103x1x5x4x8.

Layer	Number of kernels	Kernel size	Number of parameters	Output size
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$256 \\ 5152 \\ 64$	32x103x6x5x8x16 32x103x6x5x8x16 32x103x6x5x8x16 32x103x6x5x8x16 32x103x6x5x8x16
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040\\5152\\64$	32x103x6x5x8x16 32x103x6x5x8x16 32x103x6x5x8x16 32x103x6x5x8x16 32x103x6x5x8x16
icoPooling		Hexagonal		32x103x6x5x4x8
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040 \\ 5152 \\ 64$	32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040 \\ 5152 \\ 64$	32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8 32x103x6x5x4x8
icoPooling		Hexagonal		32x103x6x5x2x4
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040\\5152\\64$	32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040\\5152\\64$	32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$43040 \\ 5152 \\ 64$	32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4
icoConv 1D causal conv Layer norm ReLU	32 32	Hexagonal 5	$     \begin{array}{r}       43040 \\       5152 \\       64     \end{array} $	32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4 32x103x6x5x2x4
icoConv 1D causal conv R-pooling soft-max	32 1	Hexagonal 5	43 040 161	32x103x6x5x2x4 1x103x6x5x2x4 1x103x1x5x2x4 103x3

**Table A.8:** Architecture of the icoCNN model for r=3 power maps. The output sizes correspond to a sequence of 103 maps, i.e. an input size of 1x103x1x5x8x16.

#### A.3 Baseline models

**Table A.9:** Architecture of the 1D CNN for DOA estimation from 858 sequences of GCCs. The output sizes corresponds to sequence of length 103, i.e. an input size of 858x103.

Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
1D causal conv PReLU	1024	5	1	$4393984\1024$	1024x103 1024x103
1D causal conv PReLU	512	5	1	$\begin{array}{r} 2621952\\ 512 \end{array}$	512x103 512x103
1D causal conv PReLU	512	5	1	$\begin{array}{c}1311232\\512\end{array}$	512x103 512x103
1D causal conv PReLU	512	5	1	$\begin{array}{c}1311232\\512\end{array}$	512x103 512x103
1D causal conv PReLU	512	5	1	$\begin{array}{c}1311232\\512\end{array}$	512x103 512x103
1D causal conv PReLU	128	5	2	327808 512	128x103 128x103
1D causal conv tanh	3	5	2	1923	$3x103 \\ 3x103$

Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
1D causal conv PReLU	1024	5	1	$\begin{array}{c} 11264 \\ 1024 \end{array}$	$1024x103 \\ 1024x103$
1D causal conv PReLU	512	5	1	$\begin{array}{r} 2621952\\ 512 \end{array}$	512x103 512x103
1D causal conv PReLU	512	5	1	$\begin{array}{r}1311232\\512\end{array}$	512x103 512x103
1D causal conv PReLU	512	5	1	$\begin{array}{r}1311232\\512\end{array}$	512x103 512x103
1D causal conv PReLU	512	5	1	$\begin{array}{c}1311232\\512\end{array}$	512x103 512x103
1D causal conv PReLU	128	5	2	327808 512	128x103 128x103
1D causal conv tanh	3	5	2	1923	$3x103 \\ 3x103$

**Table A.10:** Architecture of the 1D CNN for DOA estimation from the coordinates of the maximums of the SRP-PHAT maps. The output sizes correspond to sequences of length 103, i.e. an input size of 2x103.

**Table A.11:** Architecture of the 2D CNN for DOA estimation from the spectrograms of the 12 microphone signals (window size K = 2048). The output sizes correspond to spectrograms with 103 time frames, i.e. an input size of  $24 \times 103 \times 2048$ .

Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
2D causal conv	128	5x5	1	76 928	1024x103
PReLU Max pooling		1x4		128	128x103x2048 128x103x512
2D causal conv PReLU	128	5x5	1	409728 128	128x103x512 128x103x512
Max pooling		1x4			128x103x128
2D causal conv PReLU	128	5x5	1	409728 128	128x103x128 128x103x128
Max pooling		1x4			128x103x32
2D causal conv PReLU	128	5x5	1	409728 128	128x103x32 128x103x32
Max pooling		1x4			128x103x8
2D causal conv PReLU	128	5x5	1	409728 128	128x103x8 128x103x8
1D causal conv PReLU	128	5	2	$163968 \\ 128$	128x103 128x103
1D causal conv tanh	3	5	2	1923	3x103 3x103

Layer	Number of kernels	Kernel size	Dilation	Number of parameters	Output size
2D conv	64	3x3	1	13888	64x833x256
$\operatorname{ReLU}$					64x833x256
Max pooling		1x8			64x833x32
2D conv	64	3x3	1	36928	64x833x32
ReLU					64x833x32
Max pooling		1x8			64x833x4
2D conv	64	3x3	1	36928	64x833x4
PReLU					64x833x4
Max pooling		1x2			64x833x2
GRU bi-directional	128			74496	833x128
GRU bi-directional	128			74496	833x128
Fully Connected	128			16512	833x128
Fully Connected	3			387	833x3
anh				128	833x3

**Table A.12:** Architecture of SELDnet for DOA estimation from spectrograms of the 12 microphone signals (window size K = 512). The output sizes correspond to spectrograms with 833 time frames, i.e. an input size of 24x833x256.

### Appendix B

## Results in every recording of the LOCATA dataset

The following tables, extracted from the supplementary materials of [VI], show the root mean squared angular error (RMSAE) obtained with the proposed models in every recording of the single source tasks of the development and evaluation partitions of the LOCATA dataset:

**Table B.1:** RMSAE [°] of the DOA estimated for the development partition of the LOCATA dataset with the proposed models using several map resolutions. The second (gray) numbers indicate the RMSAE without taking into account the frames when the sound source was silent.

	Model:	IcoCNN				Cross3D					1D CNN
	Input:	Icosah r=1	r=2	SRP-PI r=3	HAT maps r=4	Equ 4x8	iangula 8x16	ar SRP 16x32	-PHAT 32x64	$\begin{array}{c} maps \\ 64x128 \end{array}$	GCCs
Task 1	Recording 1	$9.34 \\ \scriptscriptstyle 8.85$	$\underset{\scriptstyle 3.44}{3.44}$	$\underset{\scriptscriptstyle{5.74}}{6.52}$	$\underset{\scriptstyle{5.65}}{6.71}$	$\underset{\scriptscriptstyle 18.67}{17.93}$	$\underset{\scriptstyle 14.18}{11.92}$	$\underset{\scriptscriptstyle{8.56}}{8.30}$	$\underset{\scriptscriptstyle{4.58}}{4.58}$	$\underset{\scriptscriptstyle{4.98}}{5.16}$	$\underset{\scriptscriptstyle 16.93}{16.93}$
	Recording 2	$\underset{8.36}{8.36}$	$\underset{\scriptstyle{6.77}}{6.77}$	$\begin{array}{c} 6.53 \\ \scriptstyle \scriptstyle 6.39 \end{array}$	$\substack{6.15\\ \scriptstyle{5.99}}$	$\underset{\scriptscriptstyle 20.22}{18.90}$	$\underset{\scriptscriptstyle{8.86}}{7.68}$	$\underset{7.11}{6.68}$	$\underset{\scriptstyle{4.92}}{4.92}$	$3.91$ $_{3.96}$	$\underset{\scriptscriptstyle 12.73}{12.73}$
	Recording 3	$\underset{\scriptscriptstyle 2.59}{2.59}$	$\underset{\scriptscriptstyle 3.15}{3.24}$	$\underset{\scriptscriptstyle{5.52}}{5.18}$	$\underset{\scriptstyle{5.26}}{5.19}$	$\underset{\scriptscriptstyle{11.39}}{10.35}$	$\underset{\scriptscriptstyle{9.30}}{6.34}$	$\underset{\scriptscriptstyle 3.36}{2.98}$	$\underset{\scriptscriptstyle 3.20}{3.20}$	$\underset{\scriptscriptstyle 2.29}{2.29}$	$\underset{\scriptscriptstyle{12.09}}{11.57}$
	Average	$\underset{\scriptstyle{6.60}}{6.60}$	$\underset{\scriptstyle{4.45}}{4.45}$	$\underset{\scriptscriptstyle{5.88}}{5.88}$	$\underset{\scriptscriptstyle{5.63}}{6.02}$	$\underset{\scriptscriptstyle 16.79}{15.72}$	$\underset{\scriptscriptstyle 10.78}{8.65}$	$\underset{\scriptstyle{6.34}}{5.99}$	$\underset{\scriptstyle{4.23}}{4.23}$	$\underset{\scriptscriptstyle 3.74}{3.74}$	$\underset{\scriptscriptstyle{13.92}}{13.92}$
Task 3	Recording 1	8.19	8.08 5.79	7.70	7.52	23.06	18.11	13.79	12.43	9.92	13.59
	Recording 2	8.15 6.86	$\underset{\scriptstyle{7.51}}{8.30}$	8.80 8.04	9.61 9.10	20.97 17.72	13.71 12.11	10.01 9.29	8.36 8.31	9.22 9.06	14.17 13.34
	Recording 3	$\underset{\scriptscriptstyle 7.57}{10.25}$	$\underset{\scriptscriptstyle{5.33}}{7.82}$	$\underset{4.69}{7.18}$	$\underset{\scriptscriptstyle{5.60}}{7.50}$	$\underset{\scriptscriptstyle 24.92}{21.05}$	$\underset{\scriptscriptstyle 12.62}{12.62}$	$9.83 \\ \scriptscriptstyle 8.71$	$\underset{\scriptstyle{6.56}}{7.69}$	$\underset{\scriptstyle{5.02}}{6.60}$	$\underset{\scriptscriptstyle{13.09}}{15.21}$
	Average	$\underset{7.29}{8.86}$	$\underset{\scriptstyle{6.21}}{8.07}$	$\underset{\scriptstyle{6.05}}{7.89}$	$\underset{\scriptscriptstyle{6.88}}{8.21}$	$\underset{\scriptscriptstyle{21.99}}{21.99}$	$\underset{\scriptscriptstyle 15.21}{14.85}$	$\underset{\scriptscriptstyle{10.98}}{11.21}$	$9.49$ $_{9.28}$	$\underset{\scriptscriptstyle{8.05}}{8.58}$	$\underset{\scriptscriptstyle{13.35}}{14.32}$
	Recording 1	$\underset{\scriptscriptstyle 13.31}{13.31}$	$\underset{\scriptscriptstyle 9.86}{9.86}$	$\underset{\scriptscriptstyle{7.50}}{7.43}$	$\underset{\scriptscriptstyle 8.26}{8.26}$	$\underset{\scriptscriptstyle{11.10}}{11.10}$	$\underset{\scriptstyle 9.40}{10.83}$	$\underset{\scriptscriptstyle 7.41}{7.41}$	$\underset{\scriptscriptstyle{5.70}}{5.74}$	$\underset{\scriptscriptstyle{5.49}}{5.49}$	$\underset{\scriptscriptstyle 10.96}{10.96}$
Tool: 5	Recording 2	14.04 12.07	$\underset{9.94}{10.64}$	$9.72$ $_{8.49}$	$9.39$ $^{8.15}$	$\underset{\scriptscriptstyle 20.67}{20.67}$	$\underset{15.56}{16.16}$	$\underset{15.84}{16.08}$	12.18 11.57	$\underset{\scriptscriptstyle 13.67}{13.67}$	$17.33$ $_{17.38}$
Task J	Recording 3	$\underset{\scriptscriptstyle{11.87}}{21.56}$	$\underset{\scriptscriptstyle 8.26}{20.10}$	$\underset{\scriptscriptstyle{8.50}}{18.37}$	$\underset{\scriptscriptstyle 8.62}{19.47}$	$\underset{\scriptscriptstyle 18.42}{23.57}$	$\underset{\scriptscriptstyle{14.72}}{18.25}$	$\underset{\scriptscriptstyle 10.88}{13.58}$	$\underset{\scriptscriptstyle{12.86}}{15.64}$	$\underset{\scriptscriptstyle{12.11}}{15.49}$	$\underset{\scriptscriptstyle 17.33}{20.14}$
	Average	$\underset{\scriptscriptstyle 12.42}{16.41}$	$\underset{\scriptscriptstyle{9.35}}{13.49}$	$\underset{8.16}{11.84}$	$\underset{\scriptstyle 8.34}{12.43}$	$\underset{\scriptstyle{16.73}}{18.81}$	$\underset{\scriptscriptstyle{13.23}}{15.08}$	$\underset{\scriptscriptstyle{11.38}}{12.31}$	$\underset{\scriptstyle 10.04}{11.19}$	$\underset{\scriptscriptstyle 10.42}{11.52}$	$\underset{\scriptscriptstyle 15.22}{16.13}$
Average		$\underset{\scriptscriptstyle 8.77}{10.65}$	$\underset{\scriptstyle 6.67}{8.69}$	$\underset{\scriptstyle 6.70}{8.61}$	8.88 6.95	18.74 18.50	$\underset{\scriptscriptstyle 13.07}{12.86}$	$9.83$ $_{9.57}$	8.31 7.85	$7.96$ $_{7.41}$	$\underset{\scriptstyle 14.16}{14.16}$
Median		$\underset{\scriptscriptstyle 8.36}{9.34}$	$\underset{\scriptstyle 6.77}{8.08}$	$7.43$ $_{6.39}$	$\underset{\scriptscriptstyle{5.99}}{7.52}$	$\underset{\scriptscriptstyle 18.67}{20.92}$	$\underset{\scriptscriptstyle 12.62}{12.62}$	$\underset{\scriptstyle 8.71}{9.83}$	$\underset{\scriptstyle{6.56}}{7.69}$	$\underset{5.49}{6.60}$	$\underset{\scriptscriptstyle{13.34}}{14.17}$
Standard deviation		$\underset{\scriptscriptstyle 3.10}{5.00}$	$\underset{\scriptscriptstyle 2.35}{4.66}$	$\underset{\scriptscriptstyle 1.37}{3.67}$	$\underset{\scriptstyle 1.45}{3.98}$	$\underset{\scriptstyle 4.45}{4.45}$	$\underset{\scriptscriptstyle 3.62}{3.62}$	$\underset{\scriptscriptstyle 3.67}{3.67}$	$\underset{\scriptscriptstyle 3.53}{3.53}$	$\underset{\scriptscriptstyle 3.72}{3.72}$	$\underset{\scriptscriptstyle 2.28}{2.28}$

	Model:	IcoCNN			Cross3D					1D CNN	
	Input:	Icosah r=1	r=2	SRP-P r=3	HAT maps r=4	Equ 4x8	iangula 8x16	ar SRP 16x32	-PHAT 32x64	$\begin{array}{c} maps \\ 64x128 \end{array}$	GCCs
	Recording 1	12.24	7.17	5.87	5.67	23.78	19.62	5.46	4.45	3.37	6.43
	Recording 2	2.33	3.74	5.43	5.30	8.55	6.64	2.70	2.14	1.75	6.25
	Recording 2	$10.70^{2.26}$	$\overset{3.75}{\textbf{3.78}}$	5.63 5.11	$\overset{5.32}{4.41}$	14.65	10.78	$\overset{2.76}{7.32}$	$^{2.18}_{8.45}$	5.17	$\overset{\scriptstyle 6.29}{\scriptstyle 27.04}$
	Recording 5	$^{11.75}_{4.73}$	$\overset{4.04}{3.85}$	$\overset{5.52}{4.74}$	$\overset{4.42}{\textbf{3.68}}$	$13.99 \\ 15.53$	$10.64 \\ 9.83$	6.25	$\overset{8.52}{4.45}$	$\overset{5.17}{\textbf{3.23}}$	$\overset{27.44}{4.42}$
	Recording 4	$\overset{4.92}{8.09}$	3.78 4.92	$\overset{4.70}{4.99}$	$\overset{\scriptscriptstyle 3.63}{4.80}$	$15.52 \\ 77.58$	$\overset{9.91}{53.83}$	$\overset{\scriptstyle 6.21}{22.18}$	$\overset{4.45}{6.55}$	$\frac{3.25}{4.84}$	$^{4.50}_{9.34}$
	Recording 5	8.17 3.25	4.89 3 33	5.13 3.67	4.73 3.58	62.77 17 19	43.74 8 52	$\frac{24.82}{5.47}$	7.27 3.82	5.35 <b>2 92</b>	9.27 14 11
	Recording 6	3.27 11 90	3.34	3.71	3.61	17.48	8.61	5.50	3.87	2.95	14.30
Task 1	Recording 7	11.38 11.59	12.00	8.29 8.03	7.01	18.39 18.57	15.01 15.75	7.58 7.58	3.80 3.77	4.14 4.14	19.30 19.43
	Recording 8	$\underset{\scriptscriptstyle{11.40}}{11.40}$	6.36	$\substack{6.16\\ \scriptstyle 6.04 }$	$5.35$ $_{5.37}$	$\underset{\scriptstyle{50.26}}{48.34}$	$\underset{\scriptstyle{46.88}}{\hphantom{0}}$	$\underset{\scriptstyle 10.45}{9.97}$	$6.42$ $_{6.72}$	$\underset{\scriptscriptstyle{4.74}}{4.55}$	7.19 7.49
	Recording 9	6.36	5.37	5.08	5.96	12.15	7.44	2.50	3.91	1.85	8.60 8.38
	Recording 10	4.61	4.21	6.66	7.79	41.80	38.27	42.57	25.79	16.32	19.87
	Becording 11	4.43	5.47	4.49	3.70	52.15	49.90	58.48	31.99	14.92	20.95
		$\overset{5.00}{8.35}$	7.31	$\overset{4.83}{4.38}$	$\overset{4.17}{4.36}$	$\overset{52.11}{23.10}$	51.77 11.80	58.92 5.04	$\overset{\scriptscriptstyle{33.24}}{2.98}$	3.07	$^{21.25}_{11.91}$
	Recording 12	$\overset{\scriptscriptstyle 10.21}{16.72}$	$\frac{7.27}{8.95}$	$\frac{5.32}{7.65}$	$\overset{5.04}{6.60}$	$\overset{\scriptscriptstyle 20.82}{31.34}$	$\overset{\scriptscriptstyle 12.36}{15.53}$	$\frac{5.35}{4.98}$	3.08 4.74	$\overset{3.20}{2.50}$	12.34 7.58
	Recording 13	16.66	9.52	7.79	6.71	32.23	15.72	4.94	4.93	2.52	7.86
	Average	$\underset{8.34}{8.34}$	$5.88$ $_{6.08}$	$\underset{\scriptscriptstyle{5.78}}{5.78}$	$\underset{\scriptscriptstyle{5.23}}{5.23}$	$\underset{\scriptscriptstyle 27.09}{28.62}$	$\underset{\scriptscriptstyle{22.16}}{22.16}$	$\underset{\scriptstyle 14.32}{13.89}$	$9.84 \\ \scriptscriptstyle 8.51$	$\underset{\scriptstyle{5.48}}{5.48}$	$\underset{\scriptscriptstyle{12.80}}{12.80}$
	Recording 1	9.91	7.68	7.92	10.28	19.66	15.32	12.92	9.04	7.40	12.18
	Recording 2	16.10	15.62	14.57	15.58	17.43	14.49	11.99	12.41	14.19	13.47
	Decording 2	8.92	8.29	14.39	7.71	23.34	26.86	17.65	16.77	11.88	13.71
Task 3	Recording 5	$\overset{9.27}{8.28}$	$\overset{7.50}{6.78}$	$\overset{7.52}{6.99}$	$\overset{8.13}{\textbf{7.84}}$	$^{24.39}_{14.63}$	$\overset{28.12}{14.45}$	18.50 9.90	$^{17.57}_{8.21}$	12.34 7.95	$14.23 \\ 12.10$
	Recording 4	$^{7.99}945$	6.24 6.31	${\overset{6.64}{6}}$	$^{7.48}_{643}$	$^{14.59}$ 17 49	14.43 16 68	$^{9.83}$ 11.36	8.10 9.49	7.87 8 16	12.00 8 98
	Recording 5	9.94	6.34	6.14	6.33	18.22	18.36	11.99	9.70	8.26	8.78
	Average	10.53 $8.97$	8.94 7.29	10.07	9.57	18.51 17.98	17.56 17.50	12.76 12.16	11.18 10.33	9.92	$12.09$ $_{11.62}$
Task 5	D !! 1	25.15	15.72	19.45	17.29	25.87	28.02	24.08	24.94	25.31	22.27
	Recording 1	$\overset{21.33}{12.47}$	12.80 10.02	$18.26 \\ 8.97$	14.31 9.08	$\frac{22.16}{12.86}$	$^{24.93}_{9.74}$	21.08 8.01	$^{24.58}_{7.58}$	23.87 7.77	16.83 14.40
	Recording 2	11.47 14.20	8.62 12.20	7.53	7.59	12.78	9.54	7.65	7.09	7.02 19.45	14.70 15 41
	Recording 3	9.42	8.52	7.79	7.50	12.30	14.00	8.79	6.22	6.67	10.65
	Recording 4	$\underset{10.65}{12.55}$	$\underset{7.65}{9.53}$	9.47 7.30	9.31 $_{6.45}$	18.51 18.25	11.66 11.48	9.91 9.60	$\underset{\scriptscriptstyle{8.92}}{9.42}$	$\underset{10.34}{10.34}$	12.72 $12.06$
	Recording 5	$\underset{\scriptscriptstyle{12.30}}{12.30}$	$\underset{\scriptscriptstyle 6.77}{7.19}$	$7.71 \\ \scriptscriptstyle 7.43$	$7.17$ $_{6.62}$	$\underset{\scriptscriptstyle{13.60}}{21.32}$	$\underset{\scriptscriptstyle 10.88}{23.21}$	$\underset{\scriptstyle{6.63}}{8.62}$	$\underset{\scriptstyle{6.33}}{7.33}$	$\underset{\scriptscriptstyle{5.74}}{5.97}$	$\underset{\scriptscriptstyle{12.29}}{17.57}$
	Average	$\underset{\scriptscriptstyle{13.03}}{15.48}$	11.13 <sub>8.87</sub>	$\underset{\scriptscriptstyle 9.66}{11.54}$	$\underset{\scriptstyle 8.49}{10.95}$	$\underset{\scriptscriptstyle 15.82}{19.13}$	$\underset{\scriptscriptstyle{13.38}}{17.49}$	$\underset{\scriptscriptstyle 10.75}{13.03}$	$\underset{\scriptscriptstyle 10.63}{12.20}$	$\underset{\scriptscriptstyle 10.73}{12.49}$	$\underset{\scriptscriptstyle 13.31}{16.47}$
	Average	10.20	7.69	7.85	7.43	24.90	20.32	13.45	9.84	7.86	13.30
<u>د</u>	N I.	9.50 9.91	$\overset{6.97}{7.17}$	$\overset{6.87}{6.66}$	$\overset{6.51}{6.60}$	$\overset{23.47}{18.51}$	$\overset{19.24}{15.32}$	13.03 9.90	$\overset{9.52}{7.58}$	5.97	$\overset{\scriptscriptstyle 12.65}{12.72}$
	Median	9.94 5.00	$\overset{6.79}{3.53}$	6.53 3.82	$\overset{6.45}{3.51}$	18.25 15.65	14.43 13.43	$\overset{7.65}{12.87}$	$\overset{7.09}{7.71}$	5.74 5.71	$^{12.00}_{5.67}$
Stand	ard deviation	4 17	2.40	2.60	2.26	1/ 30	13.24	13.26	8.07	5.37	5 44

**Table B.2:** RMSAE [°] of the DOA estimated for the evaluation partition of the LOCATA dataset with the proposed models using several map resolutions. The second (gray) numbers indicate the RMSAE without taking into account the frames when the sound source was silent.

## Appendix C

## Conclusiones

Following the regulations of the University of Zaragoza, this appendix includes a translated version of the conclusions of the thesis presented in section 6.1:

- La comunidad del procesado de señal ha estudiado el problema de la localización de fuentes sonoras durante décadas y las técnicas que fueron propuestas antes del afloramiento de soluciones basadas en *deep learning* no deberían ser ignoradas, ya que aportan representaciones acústicas con una gran correlación con la dirección de llegada de las fuentes sonoras y una mejor comprensión del problema que debería guiar el diseño de las nuevas soluciones basadas en *deep learning*.
- Podemos eliminar el efecto que una fuente en una posición dada genera en las funciones de correlación cruzada de los micrófonos de una agrupación mediante una combinación lineal de las funciones originales con retardos temporales.
- Cuando usamos datasets sintéticos para entrenar modelos de localización de fuentes sonoras, podemos generar trayectorias aleatorias ilimitadamente y por tanto no necesitamos limitar nuestros datasets a un número finito de trayectorias. Pese a que esto debería usarse con precaución, en los modelos entrenados con esta técnica no se han observado indicios de over-fitting al procedimiento de generación de las trayectorias pese a que este era bastante simple.
- Muchas partes del método de las imágenes para la simulación de acústica de salas pueden ser paralelizadas e implementadas eficientemente en GPUs. Esto

#### C. CONCLUSIONES

es esencial si queremos entrenar nuestros modelos con un *dataset* infinito simulando las trayectorias conforme son necesarias, dado que el método de las imágenes es computacionalmente costoso y su implementación secuencial ralentizaría demasiado los entrenamientos. La implementación propuesta reduce el tiempo de simulación en más de dos órdenes de magnitud comparada con otras implementaciones del estado del arte.

- Los mapas acústicos generados con el algoritmo SRP-PHAT son una representación robusta para ser usados como entrada de redes neuronales. Pese a que la principal fuente de información para la localización de una fuente sonora está en la posición de sus máximos, las redes neuronales son capaces de extraer información adicional de ellos.
- Es posible obtener resultados competitivos en la localización de una única fuente sonora usando únicamente convoluciones causales y evitando las redes recurrentes. Las redes recurrentes bidireccionales deberían ser evitadas si se quiere que los modelos sean útiles en aplicaciones en tiempo real.
- El uso de modelos que explotan la invariancia rotacional del problema de la estimación de la dirección de llegada nos permite reducir el tamaño de nuestros modelos. Múltiples arquitecturas son invariantes al grupo continuo de rotaciones esféricas, pero las convoluciones icosaédricas suponen una buena aproximación con sus 60 simetrías y tienen una implementación eficiente basada en redes convolucionales 2D convencionales.
- La capa soft-argmax propuesta permite transformar una salida de clasificación en una salida de regresión interpretándola como la distribución de probabilidad de la dirección de llegada y calculando su valor esperado. Esto reduce considerablemente el número de parámetros entrenables de los modelos y evita romper su equivariancia a las rotaciones.
- En las estrategias de entrenamiento invariantes a las permutaciones para *tracking* de múltiples fuentes, el contexto temporal que consideramos para elegir la permutación optima de las fuentes estimadas es crucial para reducir el número

de cambios de identidad en las trayectorias estimadas. El uso de una ventana deslizante presenta un buen compromiso entre el alto número de cambios de identidad obtenido cuando se elige una permutación diferente para cada instante temporal y los problemas de convergencia que se dan cuando se intenta entrenar modelos usando una única permutación para cada escena acústica.

- Podemos diseñar capas recurrentes invariantes a las permutaciones que operan sobre conjuntos en lugar de vectores mediante el uso de un módulo *multi-head attention* para asignar cada elemento del set de entrada a los elementos del set de estado y después actualizando cada elemento del conjunto de estado de forma independiente.
- El uso de capas recurrentes invariantes a las permutaciones, como la PI-GRU propuesta, para el tracking de múltiples fuentes es un campo de investigación prometedor, pero aún necesita un mayor estudio para que sea competitivo con las capas GRU convencionales.